

SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY

SPECIAL MEETING AGENDA JUNE 13, 2019 8:30 A.M.

Chairman Mark D. Radecki Vice Chair Cory C. Moss Board Member Abraham Cruz Board Member Catherine Marcucci Board Member Newell Ruggles

Location: City Council Chamber, 15651 East Stafford Street, City of Industry, California

Addressing the Agency:

- Agenda Items: Members of the public may address the Successor Agency on any matter listed on the Agenda. In order to conduct a timely meeting, there will be a three-minute time limit per person for any matter listed on the Agenda. Anyone wishing to speak to the Successor Agency is asked to complete a Speaker's Card which can be found at the back of the room and at each podium. The completed card should be submitted to the Secretary prior to the Agenda item being called and prior to the individual being heard by the Successor Agency.
- Public Comments (Non-Agenda Items Only): Anyone wishing to address the Successor Agency on an item <u>not</u> on the Agenda may do so during the "Public Comments" period. In order to conduct a timely meeting, there will be a three-minute time limit per person for the Public Comments portion of the Agenda. State law prohibits the Successor Agency from taking action on a specific item unless it appears on the posted Agenda. Anyone wishing to speak to the Successor Agency is asked to complete a Speaker's Card which can be found at the back of the room and at each podium. The completed card should be submitted to the Secretary prior to the Agenda item being called by the Secretary and prior to the individual being heard by the Successor Agency.

Americans with Disabilities Act:

In compliance with the ADA, if you need special assistance to participate in any meeting (including assisted listening devices), please contact the Office of the Secretary to the Successor Agency (626) 333-2211. Notification of at least 72 hours prior to the meeting will assist staff in assuring that reasonable arrangements can be made to provide accessibility to the meeting.

Agendas and other writings:

- In compliance with Government Code Section 54957.5(b), staff reports and other public records permissible for disclosure related to open session agenda items are available at City Hall, 15625 East Stafford Street, Suite 100, City of Industry, California, at the office of the Secretary of the Successor Agency during regular business hours, Monday through Thursday, 8:00 a.m. to 5:00 p.m., Fridays 8:00 a.m. to 4:00 p.m. Any person with a question concerning any agenda item may call the City Clerk's Office at (626) 333-2211
- 1. Call to Order
- 2. Flag Salute
- 3. Roll Call
- 4. Public Comments

5. **BOARD MATTERS**

5.1 Consideration of the Register of Demands for June 13, 2019

RECOMMENDED ACTION: Approve the Register of Demands and authorize the appropriate Agency Officials to pay the bills.

5.2 Consideration of the minutes of August 11, 2016 regular meeting, January 26, 2017 regular meeting, March 23, 2017 regular meeting, May 25, 2017 regular meeting, July 27, 2017 regular meeting, October 26, 2017 joint special meeting, June 7, 2018 joint special meeting and April 25, 2019 regular meeting

RECOMMENDED ACTION:

Approve as submitted.

5.3 Consideration of Resolution No. 2019-05 - A RESOLUTION OF THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY ("SA-IUDA") OF THE CITY OF INDUSTRY, CALIFORNIA, APPROVING AND ADOPTING THE FY 2019-20 PROPOSED SA-IUDA BUDGET

RECOMMENDED ACTION:

Adopt Resolution No. SA 2019-05.

5.4 Consideration of Resolution No. SA 2019-03 – A RESOLUTION OF THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN DEVELOPMENT AGENCY ("SUCCESSOR AGENCY") APPROVING A NEW LEASE AGREEMENT BY AND BETWEEN THE SUCCESSOR AGENCY AND THE INDUSTRY EAST BUSINESS CENTER, LLC, ("IEBC") FOR A PORTION OF THE PROPERTY DESCRIBED IN THE UNRECORDED LEASE AGREEMENT DATED APRIL 28, 2005 BY AND BETWEEN THE SUCCESSOR AGENCY'S PREDECESSOR IN INTEREST, THE INDUSTRY URBAN-DEVELOPMENT AGENCY, AND THE IEBC'S PREDECESSOR IN INTEREST, INDUSTRY EAST LAND, LLC, AND MORE COMMONLY KNOWN AS BUILDINGS NOS. 11 AND 12

RECOMMENDED ACTION:

Adopt Resolution No. SA 2019-03.

5.5 Consideration of Resolution No. SA 2019-04 – A RESOLUTION OF THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN DEVELOPMENT AGENCY ("SUCCESSOR AGENCY") APPROVING A NEW LEASE AGREEMENT BY AND BETWEEN THE SUCCESSOR AGENCY AND THE INDUSTRY EAST BUSINESS CENTER, LLC ("IEBC") FOR A PORTION OF THE PROPERTY DESCRIBED IN THE UNRECORDED LEASE AGREEMENT DATED APRIL 28, 2005 BY AND BETWEEN THE SUCCESSOR AGENCY'S PREDECESSOR IN INTEREST, THE INDUSTRY URBAN-DEVELOPMENT AGENCY, AND THE IEBC'S PREDECESSOR IN INTEREST, INDUSTRY EAST LAND, LLC., AND MORE COMMONLY KNOWN AS BUILDING NO. 2

RECOMMENDED ACTION:

Adopt Resolution No. SA 2019-04.

5.6 Consideration of a Betterment Agreement for the Enhancement of Landscaping between the City of Industry, Successor Agency to the Industry Urban-Development Agency and the City of Diamond Bar for the SR 57/60 Confluence Grand Avenue at Golden Springs Drive Project

RECOMMENDED ACTION:

Approve the Agreement.

6. Adjournment. Next regular Successor Agency meeting will be on Thursday, June 27, 2019 at 8:30 a.m.

SUCCESSOR AGENCY
ITEM NO. 5.1

Successor Agency To The Industry Urban-Development Agency Authorization For Payment of Bills June 13, 2019

<u>FUND</u>	DESCRIPTION	DISBURSEMENTS
221 222	IUDA ADMIN IUDA PROJECT 1 IUDA PROJECT 2	0.00 62,530.02 160,551.62
	TOTAL ALL FUNDS	223,081.64

<u>BANK</u>	NAME	DISBURSEMENTS
WFBK BOFA	WELLS FARGO BANK BANK OF AMERICA	160,551.62 62,530.02
	TOTAL ALL BANKS	223,081.64

Successor Agency To The Industry Urban Development Agency

Wells Fargo Bank June 13, 2019

WF.CHK - IUDA Admin WI				
	F Checking			·
06/13/2019		AVANT-GARDE, I	NC	\$905.00
Invoice	Date	Description	Amount	,
5449	05/01/2019	GRAND AVE/SR60 OFF-RAMP	\$905.00	
06/13/2019		CNC ENGINEERI	NG	\$114,326.25
Invoice	Date	Description	Amount	,
458690 458691	05/30/2019 05/30/2019	GRAND AVE/GOLDEN SPRINGS DR INTERSECTION IBC-EAST SIDE ROADWAYS	\$3,441.25 \$33,518.75	
458692	05/30/2019	IBC-WEST SIDE ROADWAYS	\$66,423.75	
458693	05/30/2019	IBC-FUTURE PHASES AND STUDIES	\$10,325.00	
458695	05/30/2019	IBC-TRAFFIC MITIGATION	\$370.00	
458697	05/30/2019	DIAMOND BAR CREEK	\$247.50	
06/13/2019		PBLA ENGINEER	ING, INC.	\$18,000.00
Invoice	Date	Description	Amount	
108-2-0519	05/01/2019	DRAINAGE DESIGN-IBC PROJECT	\$18,000.00	
06/13/2019		SCS ENGINEERS	3	\$21,645.67
Invoice	Date	Description	Amount	
0349633	04/30/2019	ENGINEERING SVC-IBC PROJECT	\$21,645.67	
06/13/2019		WKE, INC		\$5,674.70
Invoice	Date	Description	Amount	. ,
14001-64B	05/14/2019	57/60 FWY CONFLUENCE PROJ	\$5,674.70	
	Invoice 5449 06/13/2019 Invoice 458690 458691 458692 458693 458695 458697 06/13/2019 Invoice 108-2-0519 06/13/2019 Invoice 0349633	Invoice Date 5449 05/01/2019 06/13/2019 Invoice Date 458690 05/30/2019 458691 05/30/2019 458692 05/30/2019 458693 05/30/2019 458695 05/30/2019 458697 05/30/2019 Invoice Date 108-2-0519 05/01/2019 06/13/2019 Invoice Date 0349633 04/30/2019 06/13/2019 Invoice Date	Invoice	Invoice

Successor Agency To The Industry Urban Development Agency

Wells Fargo Bank

June 13, 2019

Check

Date

Payee Name

Check Amount

IUDAADM.WF.CHK - IUDA Admin WF Checking

Transaction Amou	Count	Status	Checks
\$160,551	5	Total	

Transaction Amount

\$62,530.02

Successor Agency To The Industry Urban Development Agency

Bank of America June 13, 2019

Check	Date			Payee Name	Check Amoun
8PJ1REV	/LOAN - 2008 PJ1 Revol Loan - Re	estricted			
1049	05/23/2019	· · · · · · · · · · · · · · · · · · ·		IUDA-ADMINISTRATIVE ACCOUNT	\$10,530.02
	Invoice	Date	Description .	Amount	, , , , , , , , ,
	5/23/19	05/23/2019	2008 SUB-LIEN TAX BOND-REG 5/23/19	\$10,530.02	
-					
PJ1.BOF	A.CHK - Project 1 BofA Check	king			
	A.CHK - Project 1 BofA Check 05/21/2019	king		IUDA-ADMINISTRATIVE ACCOUNT	\$52,000,00
		king Date	Description	IUDA-ADMINISTRATIVE ACCOUNT Amount	\$52,000.00
PJ1.BOF.	05/21/2019		Description TRANSFER FUNDS REG 5/23/19		\$52,000.00
	05/21/2019 Invoice	Date		Amount	\$52,000

Status

Total

Count

2

Checks

Successor Agency To The Industry Urban-Development Agency Authorization For Payment of Bills June 13, 2019

Reviewed By:	Date	
	_	
Approved By:	Date ·	

SUCCESSOR AGENCY
ITEM NO. 5.2

CALL TO ORDER

The Special Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:30 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Cory Moss, Vice Chairman Abraham Cruz, Board Member Newell Ruggles, Board Member

ABSENT: Roy Haber, Board Member

STAFF PRESENT: Paul J. Philips, Executive Director; Jamie M. Casso, Legal Counsel; Clem Calvillo, Agency Engineer; and Diane Schlichting, Assistant Secretary.

PUBLIC COMMENTS

There was none.

CLOSED SESSION

Assistant Secretary Schlichting announced there was a need for Closed Session as follows:

CONFERENCE WITH LEGAL COUNSEL – EXISTING LITIGATION

Pursuant to the Government Code Section 54956.9(d)(1)

Case: Successor Agency to the Industry-Urban Development Agency v. Grand

Central Recycling & Transfer Station, Inc., et al

Los Angeles Superior Count

Case No. BC550794

There were no public comments on the Closed Session item.

Mayor Radecki recessed the meeting into Closed Session at 8:31 a.m.

RECONVENE CITY COUNCIL MEETING

Mayor Radecki reconvened the meeting at 8:47 a.m. All members of the Successor Agency were present.

Legal Counsel Casso reported out of Closed Session.

With regard to Closed Session items 5.1, direction was given to Legal Counsel and took no reportable action.

ADJOURNMENT

There being no further business, the Successor Agency to the Industry Urban-Development Agency adjourned at 8:47 a.m.

Mark D. Radecki, Chairman

Diane M. Schlichting, Assistant Secretary

CALL TO ORDER

The Regular Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:33 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Abraham Cruz, Board Member Roy Haber, Board Member Newell Ruggles, Board Member

ABSENT: Cory Moss, Vice Chair

STAFF PRESENT: Paul J. Philips, Executive Director; Director of Development Services and Administration, Alex Gonzalez; Jamie M. Casso, Legal Counsel; Clem Calvillo, Agency Engineer; and Diane Schlichting, Assistant Secretary.

PUBLIC COMMENTS

There were no public comments.

BOARD MATTERS

CONSIDERATION OF A LICENSE AGREEMENT BETWEEN OHL, USA AND THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY FOR THE USE OF THE PUBLIC RIGHT-OF-WAY, TO ASSESSOR'S PARCEL NUMBERS 8719-007-928 AND 8719-007-930, LOCATED AT 248 NORTH GRAND AVENUE

Agency Engineer Calvillo provided a staff report and Deputy Agency Engineer Nelson provided additional information and both were available to answer any questions.

MOTION BY BOARD MEMBER HABER, AND SECOND BY BOARD MEMBER RUGGLES TO APPROVE THE AGREEMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT

BOARD MEMBERS:

MOSS

ABSTAIN E

BOARD MEMBERS:

NONE

Assistant Secretary Schlichting announced there was a need for Closed Session as follows:

CLOSED SESSION

6.1 CONFERENCE WITH LEGAL CONSEL - EXISTING LITIGATION

Pursuant to Government Code Section 54956.9(d)(1)

Case: Successor Agency to the Industry Urban-Development Agency v.

Grand Central Recycling & Transfer Station, Inc., et al.

Los Angeles Superior Court

Case No. BC 550794

There were no public comments on the Closed Session item.

Chairman Radecki recessed the meeting into Closed Session at 8:39 a.m.

RECONVENE CITY COUNCIL MEETING

Chairman Radecki reconvened the meeting at 8:46 a.m. All members of the Successor Agency were present.

Legal Counsel Casso reported out of Closed Session.

With regard to Closed Session items 6.1, direction was given to Legal Counsel, and took no reportable action.

ADJOURNMENT

There being Development	•		Agency	to	the	Industry	Urban-
		Mark I	D. Radec	ki, (Chair	man	

Diane M. Schlichting, Assistant Secretary

CALL TO ORDER

The Regular Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:32 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Cory Moss, Vice Chair

Abraham Cruz, Board Member Roy Haber, Board Member Newell Ruggles, Board Member

STAFF PRESENT: Paul J. Philips, Executive Director; Jamie M. Casso, Legal Counsel; Clem Calvillo, Agency Engineer; and Diane Schlichting, Assistant Secretary.

PUBLIC COMMENTS

There were no public comments.

BOARD MATTERS

CONSIDERATION OF AMENDMENT NO. 1 TO THE AGREEMENT FOR CONSULTING SERVICES WITH THOMSEN ENGINEERING, INC., 14-THOMSEN 13-01 A MP 99-31 #26, FOR THE DIAMOND BAR CREEK IMPROVEMENT PROJECT, EXTENDING THE TERM OF THE AGREEMENT FROM APRIL 24, 2017 TO APRIL 24, 2020

Agency Engineer Calvillo provided a staff report and Deputy Agency Engineer Nelson provided additional information and responded to questions.

MOTION BY VICE CHAIR CORY MOSS, AND SECOND BY BOARD MEMBER CRUZ TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, VC/MOSS, AND

C/RADECKI

NOES:

BOARD MEMBERS:

NONE NONE

ABSENT ABSTAIN

BOARD MEMBERS: BOARD MEMBERS:

NONE

CONSIDERATION OF AMENDMENT NO. 1 TO THE AGREEMENT FOR CONSULTING SERVICES WITH WKE, INC., 14-WKE 13-01 A MP 99-31 #26, FOR THE DIAMOND BAR CREEK IMPROVEMENT PROJECT, EXTENDING THE TERM OF THE AGREEMENT FROM APRIL 24, 2017 TO APRIL 24, 2020

MOTION BY BOARD MEMBER HABER, AND SECOND BY BOARD MEMBER RUGGLES TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, VC/MOSS, AND

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT

BOARD MEMBERS:

NONE

ABSTAIN

BOARD MEMBERS:

NONE

CONSIDERATION OF AMENDMENT NO. 2 TO THE AGREEMENT FOR CONSULTING SERVICES WITH LEIGHTON CONSULTING, INC., 14-LEIGHTON 13-01 B MP 99-31 #26, FOR THE DIAMOND BAR CREEK IMPROVEMENT PROJECT, EXTENDING THE TERM OF THE AGREEMENT FROM APRIL 24, 2017 TO APRIL 24, 2020

MOTION BY BOARD MEMBER RUGGLES, AND SECOND BY BOARD MEMBER CRUZ TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, VC/MOSS, AND

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT

BOARD MEMBERS:

NONE

ABSTAIN

BOARD MEMBERS:

NONE

CONSIDERATION OF AMENDMENT NO. 2 TO THE AGREEMENT FOR CONSULTING SERVICES WITH SAGE ENVIRONMENTAL GROUP, 14-SAGE 13-01 B MP 99-31 #26, FOR THE DIAMOND BAR CREEK IMPROVEMENT PROJECT, EXTENDING THE TERM OF THE AGREEMENT FROM APRIL 24, 2017 TO APRIL 24, 2020

MOTION BY BOARD MEMBER CRUZ, AND SECOND BY BOARD MEMBER HABER TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, VC/MOSS, AND

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT

BOARD MEMBERS:

NONE

ABSTAIN BOARD MEMBERS:

NONE

CLOSED SESSION

Assistant Secretary Schlichting announced there was a need for Closed Session as follows:

6.1 CONFERENCE WITH LEGAL COUNSEL – EXISTING LITIGATION

Pursuant to Government Code Section 54956.9(d)(1)

Case: Successor Agency to the Industry Urban-Development Agency v.

Grand Central Recycling & Transfer Station, Inc., et al.

Los Angeles Superior Court

Case no. BC550794

There were no public comments on the Closed Session item.

Chairman Radecki recessed the meeting into Closed Session at 8:42 a.m.

RECONVENE CITY COUNCIL MEETING

Chairman Radecki reconvened the meeting at 9:04 a.m. All members of the Successor Agency were present.

Legal Counsel Casso reported out of Closed Session.

With regard to Closed Session items 6.1, direction was given to Legal Counsel, and there was no reportable action.

ADJOURNMENT

There being no further business, the Successor Agency to the Industry Urban-Development Agency adjourned at 9:05 a.m.

Mark D. Radecki, Chairman

Diane M. Schlichting, Assistant Secretary

CALL TO ORDER

The Regular Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:40 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Cory Moss, Vice Chair

Abraham Cruz, Board Member Roy Haber, Board Member Newell Ruggles, Board Member

STAFF PRESENT: Paul J. Philips, Executive Director; Jamie M. Casso, Legal Counsel; Clem Calvillo, Agency Engineer; and Diane Schlichting, Assistant Secretary.

PUBLIC COMMENTS

There were no public comments.

BOARD MATTERS

CONSIDERATION OF AMENDMENT NO. 2 TO THE AGREEMENT FOR CONSULTING SERVICES WITH CASEY O'CALLAGHAN GOLF COURSE DESIGN, INC., 5-CALLAGHAN 10-01 B MP 99-31 #22, FOR THE FREEWAY CONFLUENCE PROJECT EXTENDING THE TERM OF THE AGREEMENT FROM MAY 25, 2017 TO MAY 25, 2018

Agency Engineer Calvillo provided a staff report and Deputy Agency Engineer Nelson provided additional information and both were available to answer any questions.

MOTION BY VICE CHAIR MOSS, AND SECOND BY BOARD MEMBER HABER TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, HABER, RUGGLES, VC/MOSS,

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT

BOARD MEMBERS:

NONE

ABSTAIN BOARD MEMBERS:

NONE

CLOSED SESSION

Assistant Secretary Schlichting announced there was a need for Closed Session as follows:

6.1 CONFERENCE WITH LEGAL COUNSEL – EXISTING LITIGATION

Bureyont to Covernment Code Section 54956 9(d)(1)

Pursuant to Government Code Section 54956.9(d)(1)

Case: Skyscraper Brewing Company, Inc., v. Successor Agency to the

Industry-Urban Development Agency, City of Industry, et al.

Los Angeles Superior Court East District

Case no. KC068505

There were no public comments on the Closed Session item.

Chairman Radecki recessed the meeting into Closed Session at 8:47 a.m.

RECONVENE CITY COUNCIL MEETING

Chairman Radecki reconvened the meeting at 10:18 a.m. All members of the Successor Agency were present.

Legal Counsel Casso reported out of Closed Session.

With regard to Closed Session items 6.1, direction was given to Legal Counsel, and took no reportable action.

ADJOURNMENT

There being no further business, the Successor Agency to the Industry Urban-Development Agency adjourned at 11:51 a.m.

	·
	M
	Mark D. Radecki, Chairman
	Mark B. Radooki, Orlainman
	- · · · · · · · · · · · · · · · · · · ·
D: M O I !! I !! A !! ! (O !	
Diane M. Schlichting, Assistant Secretary	
Diane in. Comoning, Addictant Occident	

CALL TO ORDER

The Regular Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:32 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Abraham Cruz, Board Member Catherine Marcucci, Board Member Newell Ruggles, Board Member

ABSENT: Cory Moss, Vice Chair

STAFF PRESENT: Paul J. Philips, Executive Director; Matt Gorman, General Counsel (representing Casso & Sparks LLP); Clem Calvillo, Agency Engineer; and Diane Schlichting, Assistant Secretary.

PUBLIC COMMENTS

There were no public comments.

CONSIDERATION OF AMENDMENT NO. 3 TO THE AGREEMENT FOR CONSULTING SERVICES WITH PBLA ENGINEERING, INC., FOR THE INDUSTRY BUSINESS CENTER PROJECT (14-PBLA 13-01 C MP 99-31 #16), IN THE AMOUNT OF \$500,000.00, FOR A TOTAL AGREEMENT AMOUNT NOT-TO-EXCEED \$3,406,835.00, ON A TIME-AND-MATERIAL BASIS

Agency Engineer Calvillo provided a staff report and Deputy Agency Engineer Nelson provided additional information and both were available to answer any questions.

MOTION BY BOARD MEMBER RUGGLES, AND SECOND BY CHIARMAN RADECKI TO APPROVE THE AMENDMENT. MOTION CARRIED 4-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, MARCUCCI, RUGGLES, C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT ABSTAIN BOARD MEMBERS: BOARD MEMBERS:

VC/MOSS

NONE

ADJOURNMENT

There being no further business, the Successor Agency to the Industry Urban-Development Agency adjourned at 8:38 a.m.

Mark D. Radecki, Chairman

Diane M. Schlichting, Assistant Secretary

CITY COUNCIL AND SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY JOINT SPECIAL MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA OCTOBER 26, 2017 PAGE 1

CALL TO ORDER

The Special Meeting of the City Council and the Successor Agency to the Industry Urban-Development Agency of the City of Industry, California, was called to order by Mayor Mark D. Radecki at 9:00 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Mayor/Chairman Mark D. Radecki.

ROLL CALL

PRESENT: Mark D. Radecki, Mayor/Chairman

Cory C. Moss, Mayor Pro Tem/Vice Chair

Abraham N. Cruz. Council Member/Board Member Catherine Marcucci, Council Member/Board Member Newell W. Ruggles, Council Member/Board Member

STAFF PRESENT: Paul J. Philips, City Manager; James M. Casso, City Attorney; Alex Gonzalez, Director of Development Services and Administration; Troy Helling, Senior Planner; and Diane M. Schlichting, Chief Deputy City Clerk.

PUBLIC COMMENTS

There were no public comments.

CLOSED SESSION

Chief Deputy City Clerk Schlichting announced there was a need for Closed Session as follows:

CONFERENCE WITH LEGAL COUNSEL – EXISTING LITIGATION 5.1 Pursuant to Government Code Section 54956.9(d)(1): Case: City of Diamond Bar v. Oversight Board of the Successor Agency to the Industry Urban-Development Agency; Successor Agency to the Industry Urban-Development Agency; et al.

CITY COUNCIL AND SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY JOINT SPECIAL MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA OCTOBER 26, 2017 PAGE 2

Superior Court of California, County of Sacramento Case No. 34-2017-80002718-CU-WM-GDS

- 5.2 CONFERENCE WITH LEGAL COUNSEL EXISTING LITIGATION
 Pursuant to Government Code Section 54956.9(d)(1)
 Case: City of Chino Hills v. Oversight Board of the Successor Agency to the Industry Urban-Development agency; et al.
 Superior Court of California, County of Sacramento
 Case No. 34-2017-80002719-CU-WM-GDS
- 5.3 CONFERENCE WITH LEGAL COUNSEL ANTICIPATED LITIGATION Initiation of litigation pursuant to Government Code Section 54956.9(d)(4): One Potential Case

There were no public comments on the Closed Session item.

Mayor Radecki recessed the meeting into Closed Session at 9:01 a.m.

RECONVENE CITY COUNCIL MEETING

Mayor Radecki reconvened the meeting at 11:29 a.m. All members of the City Council were present.

City Attorney Casso reported out of Closed Session.

With regard to Closed Session items 5.1, 5.2, and 5.3, direction was given to Legal Counsel, no final action was taken. Nothing further to report

ADJOURNMENT

There being no further business, the City Council adjourned at

MARK D. RADECKI MAYOR

CITY COUNCIL AND SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY JOINT SPECIAL MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA OCTOBER 26, 2017 PAGE 3

DIANE M. SCHLICHTING CHIEF DEPUTY CITY CLERK

JUNE 7, 2018 PAGE 1

CALL TO ORDER

The Joint Special Meeting of the City Council and the Successor Agency to the Industry Urban-Development Agency of the City of Industry, California, was called to order by Mayor/Chairman Radecki at 5:00 p.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Mayor/Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Mayor/Chairman

Cory Moss, Mayor Pro Tem/Vice Chair

Abraham N. Cruz, Council Member/Board Member Catherine Marcucci, Council Member/Board Member Newell W. Ruggles, Council Member/Board Member

STAFF PRESENT: Troy Helling, Acting City Manager; Jamie M. Casso, City Attorney; Bianca Sparks, Assistant City Attorney; Joshua Nelson, Contract City Engineer; and Diane M. Schlichting, City Clerk.

PUBLIC COMMENTS

There were none.

CLOSED SESSION

City Clerk Schlichting announced there was a need for Closed Session as follows:

5.1 Conference with real property negotiators pursuant to government code section 54956.8:

a. Property:

17647 Gale Avenue, City of Industry, CA

also known as Assessor Parcel Numbers

8264-012-923, 8264-013-913, 8264-013-9144

Agency Negotiators:

Troy Helling, Acting City Manager

James M. Casso, City Attorney

JUNE 7, 2018 PAGE 2

Negotiating Parties:

P.T. Enterprises, LLC

Under Negotiation:

Price and terms of payment

b. Property: 555 El Encanto Road, City of Industry, CA,

also known as Assessor Parcel Number 8208-

027-901, 8208-027-901

Agency Negotiators: Troy Helling, Acting City Manager

James M. Casso, City Attorney

Negotiating Parties:

Successor Agency to the Industry Urban-

Development Agency

Under Negotiation:

Price and terms of payment

There were no public comments on the Closed Session item.

Mayor/Chairman Radecki recessed the meeting into Closed Session at 5:02 p.m.

RECONVENE CITY COUNCIL MEETING

Mayor/Chairman Radecki reconvened the meeting at 6:27 p.m. All members of the City Council were present.

City Attorney Casso reported out of Closed Session.

With regard to Closed Session items 5.1, direction was given to staff, no further action was taken.

ACTION ITEMS

CONSIDERATION OF RESOLUTION NOS. CC 2018-17 AND SA 2018-02 -6.1 APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 15432 NELSON AVENUE, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

JUNE 7, 2018 PAGE 3

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-17 AND SA 2018-02. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.2 CONSIDERATION OF RESOLUTION NOS. CC 2018-18 AND SA 2018-03 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR PROPERTY LOCATED AT THE SOUTH SIDE OF WORKMAN MILL ROAD AND CROSSROADS PARKWAY NO, COUNTY OF LOS ANGELES (APN: 8120-027-270) AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-18 AND SA 2018-03. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.3 CONSIDERATION OF RESOLUTION NOS. CC 2018-19 AND SA 2018-04 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR SOUTH OF STAFFORD STREET, NORTH OF UPRR, AND WEST OF HACIENDA BOULEVARD, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

JUNE 7, 2018 PAGE 4

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-19 AND SA 2018-04. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.4 CONSIDERATION OF RESOLUTION NOS. CC 2018-20 AND SA 2018-05 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 15600 STAFFORD STREET AND 100 NORTH HACIENDA BOULEVARD, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-20 AND SA 2018-05. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.5 CONSIDERATION OF RESOLUTION NOS. CC 2018-21 AND SA 2018-06 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 15252 STAFFORD STREET, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

JUNE 7, 2018 PAGE 5

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-21 AND SA 2018-06. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.6 CONSIDERATION OF RESOLUTION NOS. CC 2018-22 AND SA 2018-07 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 911 BIXBY DRIVE, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-22 AND SA 2018-07. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES: COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES: COUNCIL/BOARD MEMBERS: NONE ABSENT: COUNCIL/BOARD MEMBERS: NONE ABSTAIN: COUNCIL/BOARD MEMBERS: NONE

6.7 CONSIDERATION OF RESOLUTION NOS. CC 2018-23 AND SA 2018-08 – APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR THE SOUTHWESTERLY AND NORTHWESTERLY CORNERS OF HACIENDA BOULEVARD AND STAFFORD STREET, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

JUNE 7, 2018 PAGE 6

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-23 AND SA 2018-08. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES:

COUNCIL/BOARD MEMBERS:

NONE NONE

ABSENT: ABSTAIN:

COUNCIL/BOARD MEMBERS: COUNCIL/BOARD MEMBERS: NONE

CONSIDERATION OF RESOLUTION NOS. CC 2018-25 AND SA 2018-10 -6.8 APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 15555 STAFFORD STREET, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-25 AND SA 2018-10. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES:

COUNCIL/BOARD MEMBERS:

NONE

ABSENT:

COUNCIL/BOARD MEMBERS:

NONE

ABSTAIN:

COUNCIL/BOARD MEMBERS:

NONE

CONSIDERATION OF RESOLUTION NOS. CC 2018-26 AND SA 2018-11 -6.9 APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 15660 STAFFORD STREET, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

JUNE 7, 2018 PAGE 7

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-26 AND SA 2018-11. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES:

COUNCIL/BOARD MEMBERS:

NONE NONE

ABSENT: ABSTAIN: COUNCIL/BOARD MEMBERS: COUNCIL/BOARD MEMBERS: NONE

6.10 CONSIDERATION OF RESOLUTION NOS. CC 2018-28 AND SA 2018-12 -APPROVING THE PURCHASE AND SALE AGREEMENT BETWEEN THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY AND THE CITY OF INDUSTRY FOR 425 PARRIOTT PLACE, CITY OF INDUSTRY AND MAKING THE REQUISITE CEQA FINDINGS

Contract City Engineer Josh Nelson provided slides and an overview of the property to the Council and was available to answer any questions.

MOTION BY MAYOR PRO TEM/VICE CHAIR MOSS, AND SECOND BY COUNCIL MEMBER/BOARD MEMBER RUGGLES, TO ADOPT RESOLUTION NOS. CC 2018-28 AND SA 2018-12. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

COUNCIL/BOARD MEMBERS: CRUZ, MARCUCCI, RUGGLES, MOSS,

RADECKI

NOES:

COUNCIL/BOARD MEMBERS:

NONE

ABSENT:

COUNCIL/BOARD MEMBERS:

NONE

ABSTAIN:

COUNCIL/BOARD MEMBERS:

NONE

ADJOURNMENT

There being no further business, the Joint Special Meeting adjourned at 6:46 p.m.

JUNE 7, 2018 PAGE 8

> MARK RADECKI MAYOR/CHAIRMAN

DIANE M. SCHLICHTING CITY CLERK/SECRETARY

CALL TO ORDER

The Regular Meeting of the Successor Agency to the Industry Urban-Development Agency was called to order by Chairman Radecki at 8:30 a.m. in the City of Industry Council Chamber, 15651 East Stafford Street, California.

FLAG SALUTE

The flag salute was led by Chairman Radecki.

ROLL CALL

PRESENT: Mark Radecki, Chairman

Cory Moss, Vice Chair

Abraham Cruz, Board Member Catherine Marcucci, Board Member Newell Ruggles, Board Member

STAFF PRESENT: Troy Helling, Executive Director; Bing Hyun, Assistant City Manager; Jamie M. Casso, Legal Counsel; Joshua Nelson, Contract City Engineer; and Julie Robles, Assistant Secretary.

PUBLIC COMMENTS

There were no public comments.

CONSIDERATION OF THE REGISTER OF DEMANDS FOR APRIL 25, 2019

MOTION BY VICE CHAIR MOSS, AND SECOND BY CHAIRMAN RADECKI TO APPROVE THE REGISTER OF DEMANDS AND AUTHORIZE THE APPROPRIATE CITY OFFICIALS TO PAY THE BILLS. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, MARCUCCI, RUGGLES, VC/MOSS,

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT:

BOARD MEMBERS:

NONE

ABSTAIN:

BOARD MEMBERS:

NONE

SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY REGULAR MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA APRIL 25, 2019 PAGE 2

5.2 CONSIDERATION OF THE REGISTER OF DEMANDS FOR MAY 23, 2019

MOTION BY BOARD MEMBER CRUZ, AND SECOND BY VICE CHAIR MOSS TO APPROVE THE REGISTER OF DEMANDS AND AUTHORIZE THE APPROPRIATE CITY OFFICIALS TO PAY THE BILLS. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, MARCUCCI, RUGGLES, VC/MOSS,

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT:

BOARD MEMBERS:

NONE

ABSTAIN: BOARD MEMBERS:

NONE

5.3 CONSIDERATION OF THE MINUTES OF JUNE 21, 2018 SPECIAL JOINT MEETING AND THE APRIL 25, 2019 REGULAR MEETING

MOTION BY BOARD MEMBER RUGGLES, AND SECOND BY VICE CHAIR MOSS TO APPROVE AS SUBMITTED. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

AYES:

BOARD MEMBERS:

CRUZ, MARCUCCI, RUGGLES, VC/MOSS,

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT:

BOARD MEMBERS:

NONE

ABSTAIN:

BOARD MEMBERS:

NONE

5.4 CONSIDERATION OF AMENDMENT NO. 2 TO THE LICENSE AGREEMENT WITH OHL, USA, INC. TO EXTEND THE TERM FOR THE USE OF ASSESSOR'S PARCEL NOS. 8719-007-928 AND 8719-007-930, LOCATED AT 248 NORTH GRAND AVENUE

Contract City Engineer Josh Nelson provided a staff report to the Council and was available to answer any questions.

MOTION BY VICE CHAIR MOSS, AND SECOND BY BOARD MEMBER CRUZ TO APPROVE THE AMENDMENT. MOTION CARRIED 5-0, BY THE FOLLOWING VOTE:

SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY REGULAR MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA APRIL 25, 2019 PAGE 3

AYES:

BOARD MEMBERS:

CRUZ, MARCUCCI, RUGGLES, VC/MOSS,

C/RADECKI

NOES:

BOARD MEMBERS:

NONE

ABSENT:

NONE

BOARD MEMBERS: BOARD MEMBERS: ABSTAIN: NONE

CONSIDERATION OF RESOLUTION NO. SA 2019-03 - A RESOLUTION OF 5.5 THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN DEVELOPMENT AGENCY ("SUCCESSOR AGENCY") APPROVING A NEW LEASE AGREEMENT BY AND BETWEEN THE SUCCESSOR AGENCY AND THE INDUSTRY EAST BUSINESS CENTER, LLC, ("IEBC") FOR A PORTION OF THE PROPERTY DESCRIBED IN THE UNRECORDED LEASE AGREEMENT DATED APRIL 28, 2005 BY AND BETWEEN THE SUCCESSOR AGENCY'S PREDECESSOR IN INTEREST, THE INDUSTRY URBAN-DEVELOPMENT AGENCY, AND THE IEBC'S PREDECESSOR IN INTEREST, INDUSTRY EAST LAND, LLC, AND MORE COMMONLY KNOWN AS BUILDINGS NOS. 11 AND 12

THIS ITEM WAS PULLED FROM THE AGENDA. PER LEGAL COUNSEL CASSO, INTEND TO BRING THIS ITEM BACK AT THE JUNE 13, 2019 MEETING.

CONSIDERATION OF RESOLUTION NO. SA 2019-04 - A RESOLUTION OF THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN DEVELOPMENT AGENCY ("SUCCESSOR AGENCY") APPROVING A NEW LEASE AGREEMENT BY AND BETWEEN THE SUCCESSOR AGENCY AND THE INDUSTRY EAST BUSINESS CENTER, LLC ("IEBC") FOR A PORTION OF THE PROPERTY DESCRIBED IN THE UNRECORDED LEASE AGREEMENT DATED APRIL 28, 2005 BY AND BETWEEN THE SUCCESSOR AGENCY'S PREDECESSOR IN INTEREST, THE INDUSTRY URBAN-DEVELOPMENT AGENCY, AND THE IEBC'S PREDECESSOR IN INTEREST, INDUSTRY EAST LAND, LLC., AND MORE COMMONLY KNOWN AS BUILDING NO. 2

THIS ITEM WAS PULLED FROM THE AGENDA. PER LEGAL COUNSEL CASSO, INTEND TO BRING THIS ITEM BACK AT THE JUNE 13, 2019 MEETING.

ADJOURNMENT

SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY REGULAR MEETING MINUTES CITY OF INDUSTRY, CALIFORNIA APRIL 25, 2019 PAGE 4

There being no further business, the Development Agency adjourned at 8.37 a		Agency	to the	Industry	Urban-
	Mark I	D. Radec	ki, Chair	man	
Julie Robles, Assistant Secretary					

SUCCESSOR AGENCY
ITEM NO. 5.3

MEMORANDUM

TO:

Board of Directors

FROM:

Troy Helling, City Manager

STAFF:

Yamini Pathak, Director of Finance

Dean Yamagata, Frazer, LLP - Financial Consultant

DATE:

June 13, 2019

SUBJECT:

Consideration of Resolution No. 2019-05 - A Resolution of the

Successor Agency To The Industry Urban-Development Agency ("SA-IUDA") of the City of Industry, California, Approving and Adopting the

FY 2019-20 Proposed SA-IUDA Budget.

Budget Recap

Below is a summary of the revenues and expenditures for the Successor Agency to the Industry Urban-Development Agency.

SUCCESSOR AGENCY TO THE IUDA
BUDGET COMPARISON TO PRIOR YEAR
PROPOSED BUDGET 2019-2020

	F	PROPOSED REVENUES		PROPOSED BUDGETED EXPENDITURES						
	2019-2020	2018-2019	% - CHANGE	2019-2020	2018-2019	% - CHANGE				
220 SA ADMINISTRATION	100	600	-83%	1,721,000	2,186,840	-21%				
221 PROJECT AREA # 1	7,762,000	1,200,000	547%	12	-	0%				
222 PROJECT AREA # 2	7,595,000	6,520,000	16%	68,769,238	103,457,400	-34%				
223 PROJECT AREA # 3	48,000	15,000	220%			0%				
230 PUBLIC WORKS		7,500	-100%	100		0%				
231 LAND	1,003,000	25,000	3912%	-	10	0%				
240 RPTTF FUND	49,881,452		0%	-	-	0%				
241 DEBT SERVICES / PROJECT AREA #1	42,037,400	41,548,430	1%	42,769,094	42,819,490	0%				
242 DEBT SERVICES / PROJECT AREA #2	11,121,600	10,261,500	8%	58,714,318	26,591,815	121%				
243 DEBT SERVICES / PROJECT AREA #3	6,559,600	6,355,240	3%	5,834,987	5,831,740	0%				
Total SA TO IUDA	126,008,152	65,933,270	91%	177,808,637	180,887,285	-2%				

The proposed FY 2019-20 Successor Agency Budget will total \$177,808,637.00 in expenditures and will be supported by \$126,008,152.00 in revenues consisting of RPTTF revenues, tax override revenues, available bond proceeds, lease income from SA owned properties, and property sales. Any expenditures above the revenues received will be supported by future property sales.

FISCAL IMPACT

The FY 2019-20 Proposed SA-IUDA will total \$177,808,637.00 in expenditures and \$126,008,152.00 in revenue.

RECOMMENDATION

Staff recommends the Successor Agency to the Industry Urban-Development Agency board to approve and adopt Resolution No. SA 2019-05, adopting the Proposed Operating Budget for the Successor Agency to the Industry Urban-Development Agency for Fiscal Year 2019-2020.

Attachments:

- 1. Resolution No. SA 2019-05 Resolution Approving the FY 20 Proposed SA Budget
- 2. Exhibit A FY 2019-20 Proposed SA Budget

RESOLUTION NO. SA 2019-05

A RESOLUTION OF THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY OF THE CITY OF INDUSTRY, CALIFORNIA, APPROVING AND ADOPTING THE FY 2019-20 PROPOSED SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY BUDGET

WHEREAS, On June 11, 2019, the Successor Agency to the Industry Urban-Development Agency of the City of Industry held a Special Budget Workshop meeting and received a presentation on the FY 2019-20 ("FY 20") Proposed Operating Budget for the City of Industry and all its affiliated entities, including the Successor Agency to the Industry Urban-Development Agency ("SA") and

WHEREAS, the SA-IUDA's FY 2019-20 Proposed Budget is hereby incorporated into the City's FY 2019-20 Proposed Operating Budget.

NOW, THEREFORE, THE SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY OF THE CITY OF INDUSTRY, CALIFORNIA, DOES HEREBY RESOLVE AS FOLLOWS:

- **Section 1.** The above recitals are true and correct and are incorporated herein by reference.
- **Section 2.** By approving Resolution No. CC 2019-05, the Proposed Successor Agency Budget of **\$177,808,637** in expenditures and will be supported revenues consisting of RPTTF tax increment revenues, tax override revenues, available bond proceeds, lease income from SA owned properties, and property sales is hereby approved and adopted for Fiscal Year 2019-2020. Any expenditures above the revenues received will be supported by property sales.
- <u>Section 3.</u> Contingent upon City Council approval, the SA-IUDA's Budget for the fiscal year 2019-2020 is hereby approved and incorporated into the City's FY 2019-20 Proposed Operating Budget.
- <u>Section 4.</u> The provisions of this Resolution are severable and if any provision, clause, sentence, word or part thereof is held illegal, invalid, unconstitutional, or inapplicable to any person or circumstances, such illegality, invalidity, unconstitutionality, or inapplicability shall not affect or impair any of the remaining provisions, clauses, sentences, sections, words or parts thereof of the Resolution or their applicability to other persons or circumstances.
- **Section 5.** The Secretary shall certify to the passage and adoption of this resolution and enter it into the book of original resolutions.

ABSENT:	BOARD MEMBERS:	
ABSTAIN:	BOARD MEMBERS:	
NOES:	BOARD MEMBERS:	
AYES:	BOARD MEMBERS:	

PASSED, APPROVED AND ADOPTED by the Successor Agency to the Industry

Urban-Development Agency special meeting held on June 13, 2019, by the following

vote:

SUCCESSOR AGENCY TO THE IUDA BUDGET COMPARISON TO PRIOR YEAR PROPOSED BUDGET 2019-2020

Exhibit A

		PROPOSED REVENUES		PROPOSED BUDGETED EXPENDITURES						
	2019-2020	2018-2019	% - CHANGE	2019-2020	2018-2019	% - CHANGE				
220 SA ADMINISTRATION	100	600	-83%	1,721,000	2,186,840	-21%				
221 PROJECT AREA # 1	7,762,000	1,200,000	547%	-	-	0%				
222 PROJECT AREA # 2	7,595,000	6,520,000	16%	68,769,238	103,457,400	-34%				
223 PROJECT AREA # 3	48,000	15,000	220%	-	-	0%				
230 PUBLIC WORKS	-	7,500	-100%	-	-	0%				
231 LAND	1,003,000	25,000	3912%	-	-	0%				
240 RPTTF FUND	49,881,452	-	0%	-	-	0%				
241 DEBT SERVICES / PROJECT AREA #1	42,037,400	41,548,430	1%	42,769,094	42,819,490	0%				
242 DEBT SERVICES / PROJECT AREA #2	11,121,600	10,261,500	8%	58,714,318	26,591,815	121%				
243 DEBT SERVICES / PROJECT AREA #3	6,559,600	6,355,240	3%	5,834,987	5,831,740	0%				
Total SA TO IUDA	126,008,152	65,933,270	91%	177,808,637	180,887,285	-2%				

SUCCESSOR AGENCY PROPOSED ANNUAL BUDGET REVENUE SUMMARY

REVENUE SUMMARY BY FUND SUCCESSOR AGENCY			AMENDED BUDGET 2017-18	ACTUAL 2017-18	ADOPTED BUDGET 2018-19	AMENDED BUDGET 2018-19	ACTUAL 3/31/2019	PROPOSED BUDGET 2019-20		
ADMINISTRATION - 220		\$ 100	\$ 100	\$ 37	\$ 600	\$ 600	\$ -	\$ 100		
PROJECT AREA # 1 - 221		1,196,325	1,196,325	2,389,892	1,200,000	1,200,000	5,405,229	7,762,000		
PROJECT AREA #2 - 222		7,163,250	7,163,250	8,956,813	6,520,000	6,520,000	7,773,420	7,595,000		
PROJECT AREA # 3 - 223		15,000	15,000	35,915	15,000	15,000	34,558	48,000		
PUBLIC WORKS - 230		7,500	7,500	3,754	7,500	7,500				
LAND - 231		5,100	5,100	828,708	25,000	25,000	47,431,113	1,003,000		
RPTTF FUND - 240		-	-	•	-	· •	-	49,881,452		
DEBT SERVICES / PROJECT AREA #1 - 241		40,733,760	40,733,760	41,669,682	41,548,430	41,548,430	23,047,214	42,037,400		
DEBT SERVICES / PROJECT AREA #2 - 242		10,060,300	10,060,300	11,056,667	10,261,500	10,261,500	5,251,216	11,121,600		
DEBT SERVICES / PROJECT AREA #3 - 243		6,230,625	6,230,625	6,513,994	6,355,240	6,355,240	3,363,972	6,559,600		
	TOTAL SUCCESSOR AGENCY	\$ 65,411,960	\$ 65,411,960	\$ 71,455,462	\$ 65,933,270	\$ 65,933,270	\$ 92,306,722	\$ 126,008,152		

Successor Agency PROPOSED ANNUAL BUDGET SA EXPENDITURES

OBJECT#	ACCOUNT DESCRIPTION	ADOPTED BUDGET 2017-18		BUDGET BUDGET		ACTUAL 2017-18		ADOPTED BUDGET 2018-19		AMENDED BUDGET 2018-19		ACTUAL 3/31/2019			PROPOSED BUDGET 2019-20
FUND 220															
5001	SALARIES	\$	593,730	\$	593,730	\$	-	\$	-	\$	-	\$	-	\$	_
5120.01	PROFESSIONAL SERVICES		279,270		279,270		76,839		-		-		6,058		-
5120.02	LEGAL SERVICES		696,000		696,000		281,056		-		-		-		-
5432	SA IUDA ADMINISTRATION EXPENSES		-		-		1,018,670		1,622,740		1,622,740		1,880,040		1,721,000
8510	PROPERTY MAINTENANCE		65,100		65,100		6,040		-		-		-		-
8510.01	PROPERTY MAINTENANCE - CIVIC FINANCIAL CENTER		509,600		509,600		345,357		564,100		564,100		286,402		-
8510.03	PROPERTY MAINTENANCE - TRES' HERMANOS		127,400		127,400		54,178		-		-		170,734		-
5018	OFFICE SUPPLIES & POSTAGE		-		-		254		-		-		-		-
5025	MISCELLANEOUS		4,000		4,000		118		-		-		-		-
5620	VEHICLE EXPENSES		-		-		979		-		-		-		-
5021	DUES AND SUBSCRIPTIONS		-		-		296		-		-		-		-
5012	GENERAL INSURANCE AND BONDING		49,000		49,000		30,301		-		-		-		-
5220	APPRAISAL FEES		9,000		9,000		-		-		-		-		-
5740	PROPERTY TAXES AND ASSESSMENTS		159,000		159,000		164,593		-		-		100,906		-
	SA ADMINISTRATION TOTAL	\$	2,492,100	\$	2,492,100	\$	1,978,681	\$	2,186,840	\$	2,186,840	\$	2,444,140	\$	1,721,000
FUND 222															
5100	ADMINISTRATIVE COSTS	\$	412,520	\$	412,520	\$	2,000	\$	2,000	\$	2,000	\$	_	Ś	_
5120.01	PROFESSIONAL SERVICES	•	682,000	•	682,000	*	84,756	~	100,000	~	100,000	Ţ	48,823	Y	12,957,072
5130	PLANNING, SURVEY AND DESIGN		11,155,390		11,155,390		8,315,599		15,428,400		15,428,400		2,048,946		4,245,675
5900	GENERAL ENGINEERING		399,000		399,000		462,497		600,000		600,000		256,529		4,243,073
5190	SITE CLEARANCE COSTS		470,000		470,000		-		-		-		-		_
5200	PROJECT IMPROVEMENT COSTS		82,280,055		82,280,055		55,704,772		87,327,000		87,327,000		13,787,134		51,542,191
5640	ADVERTISING AND PRINTING		34,300		34,300		2,420		2.,22.,000		-		15,707,154		31,342,131
5025	MISCELLANEOUS		25,000		25,000		-,		-		_		5		24,300
					•								•		24,300
	PROJECT AREA #2 TOTAL	\$	95,458,265	\$	95,458,265	\$	64,572,044	\$	103,457,400	\$	103,457,400	\$	16,141,437	\$	68,769,238
FUND 241															
9470.01	2015 A PROJ # 1 TAX ALLOC REV REFUNDING BOND TAXABLE-PRINCIPAL	Ś	36,180,000	\$	36,180,000	\$	32,663,098	\$	36,945,000	\$	36.045.000	٠	25 700 102	٠	27 025 000
9470.02	2015 A PROJ # 1 TAX ALLOC REV REFUNDING BOND TAXABLE-INTEREST	*	6,667,535	٦	6,667,535	Ļ	6,667,535	ڔ	5,870,490	ş	36,945,000 5,870,490	Þ	35,799,192	\$	37,925,000
9750	FISCAL AGENT FEES		2,000		2,000		-		3,870,490 4,000		5,870,490 4,000		5,870,490 -		4,840,094 4,000
					,				.,		.,550				4,000
	DEBT SERVICE PROJECT AREA #1 TOTAL	\$	42,849,535	\$	42,849,535	\$	39,330,633	\$	42,819,490	\$	42,819,490	\$	41,669,682	\$	42,769,094

Successor Agency PROPOSED ANNUAL BUDGET SA EXPENDITURES

OBJECT #	ACCOUNT DESCRIPTION	ADOPTED BUDGET 2017-18		AMENDED BUDGET 2017-18		ACTUAL 2017-18		ADOPTED BUDGET 2018-19		AMENDED BUDGET 2018-19		ACTUAL 3/31/2019		PROPOSED BUDGET 2019-20
9470.02 9470.03 9470.04 9471.02 9471.03 9471.04 9750	2015 A PROJ # 2 TAX ALLOC REV REFUNDING BOND TAX EXEMP-PRINCIPAL 2015 B PROJ # 2 TAX ALLOC REV REFUNDING BOND TAXABLE -PRINCIPAL 2015 A PROJ # 2 SUB TAX ALLOC REV REFUNDING BOND -PRINCIPAL 2015 A PROJ # 2 TAX ALLOC REV REFUNDING BOND TAX EXEMP-INTEREST 2015 B PROJ # 2 TAX ALLOC REV REFUNDING BOND TAXABLE -INTEREST 2015 A PROJ # 2 SUB TAX ALLOC REV REFUNDING BOND -INTEREST FISCAL AGENT FEES	\$	700,000 16,040,000 3,255,000 332,250 10,336,065 597,715 2,000	\$	700,000 16,040,000 3,255,000 332,250 10,336,065 597,715 2,000	\$	54,452 - - 332,250 10,336,064 298,856 4,000	\$	735,000 16,420,000 - 297,250 9,135,565 - 4,000	\$	735,000 16,420,000 - 297,250 9,135,565 - 4,000	\$	735,000 1,575,111 - 297,250 9,135,563 - -	\$ 770,000 50,905,000 - 260,500 6,774,818 - 4,000
	DEBT SERVICE PROJECT AREA #2 TOTAL	\$	31,263,030	\$	31,263,030	\$	11,025,622	\$	26,591,815	\$	26,591,815	\$	11,742,924	\$ 58,714,318
FUND 243														
9470.05 9470.06 9741.05 9471.06 9750	2015 A PROJ # 3 TAX ALLOC REV REFUNDING BOND TAX EXEMP-PRINCIPAL 2015 B PROJ # 3 TAX ALLOC REV REFUNDING BOND TAXABLE -PRINCIPAL 2015 A PROJ # 3 TAX ALLOC REV REFUNDING BOND TAX EXEMP-INTEREST 2015 B PROJ # 3 TAX ALLOC REV REFUNDING BOND TAXABLE -INTEREST FISCAL AGENT FEES	\$	710,000 3,110,000 336,250 1,524,185 2,000	\$	710,000 3,110,000 336,250 1,524,185 2,000	\$	710,000 3,110,000 336,250 1,524,187 2,000	\$	740,000 3,325,000 300,750 1,461,990 4,000	\$	740,000 3,325,000 300,750 1,461,990 4,000	\$	740,000 3,325,000 300,750 1,461,987	\$ 780,000 3,425,000 263,750 1,362,237 4,000
	DEBT SERVICE PROJECT AREA #3 TOTAL	\$	5,682,435	\$	5,682,435	\$	5,682,437	\$	5,831,740	\$	5,831,740	\$	5,827,737	\$ 5,834,987
	TOTAL SUCCESSOR AGENCY	\$	177,745,365	\$	177,745,365	\$	122,589,417	\$	180,887,285	\$	180,887,285	\$	77,825,920	\$ 177,808,637

SUCCESSOR AGENCY ITEM NO. 5.4

Staff Report & Resolution will be provided

Prior to the Meeting

LEASE AGREEMENT

THIS LEASE AGREEMENT (hereinafter referred to as "Agreement") entered into this 23rd day of May, 2019 by and between the SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY, a body corporate and politic (hereinafter referred to as "Lessor") and INDUSTRY EAST BUSINESS CENTER, LLC, a Delaware limited liability company authorized to do business in the State of California (hereinafter referred to as "Company"):

WITNESSETH:

WHEREAS, Lessor is the owner of certain real property located in the State of California described on Exhibit "A" attached hereto (hereinafter referred to as "Premises"); and

WHEREAS, by an unrecorded Lease Agreement dated April 28, 2005 (the "Lease"), Lessor's predecessor in interest, the Industry Urban-Development Agency leased the Premises and certain other real property located in the State of California (the "Larger Property") to Company's predecessor in interest, Industry East Land, LLC. Industry East Land, LLC assigned its interest in the Lease to Company by an unrecorded Assignment of Lease dated September 21, 2005. It was the intent of the parties that the Larger Property would be subdivided, financed and developed, in phases, over a number of years; and

WHEREAS, in order to facilitate the subdivision, financing and development of the Larger Property in phases, Article 4.8 of the April 28, 2005 Lease granted Company the right to have any portion of the Larger Property removed from the legal description of the April 28, 2005 Lease and made the subject of a separate lease between Lessor and Company for the remainder of the term of the April 28, 2005 Lease and on the same terms and conditions as the April 28, 2005 Lease; and

WHEREAS, Lessor and Company are removing the Premises described on Exhibit "A" from the legal description of the April 28, 2005 Lease and entering into this separate lease of the Premises pursuant to Article 4.8 of the April 28, 2005 Lease; and

WHEREAS the term of the Lease has not commenced pursuant to Article 1.2 of the Lease prior to this Agreement. The term of the leasehold interest under this Agreement shall commence upon the date specified in Section 1.2.1 herein.

WHEREAS, Company proposes to lease the Premises from Lessor and to develop thereon and operate and manage certain buildings and other improvements as permitted by this Agreement; and

WHEREAS, Company is engaged in the business of planning, constructing, maintaining, leasing financing and operating such projects; and

WHEREAS, on and subject to the terms and conditions contained in this Agreement, Lessor is willing to lease the Premises to Company for such purposes;

NOW, THEREFORE, for and in consideration of the agreements, covenants and conditions herein, Lessor and Company agree as follows:

ARTICLE I

1.1 **DEFINITIONS**

In addition to the terms defined elsewhere in this Agreement, the following terms shall have the following meanings:

- 1.1.1 The term "Approved Budget," whenever used herein, means any written budget prepared by Company and approved by the Lessor's Designated Representative pursuant to the procedure set forth in Section 1.6 entitled BUDGET APPROVAL.
- 1.1.2 The term "Assignee," whenever used herein, means the purchaser or any heir, successor, or assign of a lender subsequent to a sale or assignment as defined in Section 2.18 entitled FINANCING.
- 1.1.3 The term "Capital Improvement Expenditures," whenever used herein, means any expenses associated with the Facilities (other than the costs and expenses of their initial construction) which are treated as capital expenditures under generally accepted accounting principles, consistently applied.
- 1.1.4 The term "Lessor's Designated Representative (hereinafter referred to as 'LDR')," whenever used herein, means the Executive Director of Lessor, or a successor designee acting on behalf of the Lessor.
- 1.1.5 The term "Commence Construction," whenever used herein, means commencing construction of the Facilities on the Premises by Company causing its construction contractor to obtain occupancy and control of the area and to begin actual site development and construction thereon.
- 1.1.6 The term "Facilities," whenever used herein, means all the buildings, utilities, rail improvements, site development work, parking lots, roadways, landscaping and other improvements constructed by Company on the Premises in accordance with the terms and conditions of this Agreement.
- 1.1.7 The term "Company," whenever used herein, means Industry East Business Center, LLC, a limited liability company authorized to do business in the State of California, which is entering into this Agreement as the developer and operator of the Facilities on the Premises as described herein.
- 1.1.8 The term "Lessor," whenever used herein, means the Successor Agency to the Industry Urban-Development Agency, a body corporate and politic, and where this Agreement speaks of "Approval by Lessor," or "Approved by Lessor," such phrases mean approving action by the Successor Agency to the Industry Urban-Development Agency.
- 1.1.9 The term "CC&Rs," whenever used herein, means any Covenants, Conditions and Restrictions developed by the Company and approved by the LDR for Sublessees and Tenants which will include, but not be limited to, specific guidelines for uses of the Premises.

- 1.1.10 The term "Debt Service," whenever used herein, means the Company's payment of principal, interest and any other sum due and owing (monthly or otherwise) pursuant to the terms and conditions of any Leasehold Mortgage for construction, interim and/or permanent financing of the Facilities, but excluding any penalties, late charges, default interest and other amounts (except principal and interest) payable solely as a result of a breach or default under such Leasehold Mortgage, unless such breach or default is caused in part by the acts or omissions of Lessor, including the failure of Lessor to make Capital Contributions as called for in this Agreement.
- 1.1.11 The term, "Commencement Date," whenever used herein, means the date that the term of this Lease commences under the terms of Section 1.2.1 below.
- 1.1.12 The term "Environmental Laws," whenever used herein, means any one or all of the laws and/or regulations of the Environmental Protection Agency or any other federal, state or local agencies regulating Hazardous Material, including, but not limited to the following as the same are amended from time to time:

COMPREHENSIVE ENVIRONMENTAL RESPONSE, COMPENSATION AND LIABILITY ACT (42 U.S.C. Section 9601 et seq.)

RESOURCE CONSERVATION AND RECOVERY ACT (42 U.S.C. Section 6901 et seq.)

TOXIC SUBSTANCES CONTROL ACT, as amended (15 U.S.C. Section 2601 et seq.)

SAFE DRINKING WATER ACT (42 U.S.C. Section 300h et seq.)

CLEAN WATER ACT (33 U.S.C. Section 1251 et seq.)

CLEAN AIR ACT (42 U.S.C. Section 7401 et seq.)

and the regulations promulgated thereunder and any other laws, regulations and ordinances (whether enacted by the Federal, State or local government) now in effect or hereinafter enacted that deal with the regulation or protection of the environment (including, but not limited to, the ambient air procedures and records detailing chlorofluorocarbons [CFC]), ambient air, ground water, surface water and land use, including sub-strata land.

1.1.13 The term "Hazardous Material," whenever used herein, means the definitions of hazardous substance, hazardous material, toxic substance, regulated substance or solid waste as defined within the following:

COMPREHENSIVE ENVIRONMENTAL RESPONSE, COMPENSATION AND LIABILITY ACT (42 U.S.C. Section 9601 et seq.)

RESOURCE CONSERVATION AND RECOVERY ACT (42 U.S.C. Section 6901 et seq.)

HAZARDOUS MATERIALS TRANSPORTATION ACT (49 U.S.C. Section 1801 et seq.) and all present or future regulations promulgated thereto.

DEPARTMENT OF TRANSPORTATION TABLE (49 C.F.R. Section 172.101) and amendments thereto

ENVIRONMENTAL PROTECTION AGENCY (40 C.F.R. Part 302 and amendments thereto)

All substances, materials and wastes that are, or that become, regulated under, or that are classified as hazardous or toxic under any environmental law, whether such laws are Federal, State or local.

- 1.1.14 The term "Leasehold Mortgagee", wherever used herein, means any lender of Company holding the beneficial interest in any Leasehold Mortgage and which Leasehold Mortgagee is a bank, savings and loan, insurance company, pension trust, real estate investment trust, mortgage company, or other institutional lender unaffiliated with the Company.
- 1.1.15 The term "Leasehold Mortgage", wherever used herein means any mortgage or deed of trust which is in favor of a Leasehold Mortgagee encumbering the Company's interest in this Agreement, the leasehold estate created hereunder and/or the Facilities from time to time located on the Premises and which has been incurred for, and the proceeds of which are used for, the purpose of financing the construction of the Facilities and the payment of other Project Costs or refinancing an existing Leasehold Mortgage. The proceeds of any refinancing, to the extent not used to pay off the Leasehold Mortgage being refinanced, shall be used for the purpose of: (i) repairing, constructing or reconstructing the Facilities; (ii) held in reserve for such purposes; (iii) the payment of other Project Costs; (iv) used for the repayment of Equity Contributions and interest thereon or (v) used for such other purposes as Lessor and Company may, in writing, agree. No Leasehold Mortgage shall encumber any interest in any real property other than Company's leasehold interest under this Agreement and the Facilities.
- 1.1.16 The term "Management Fee," whenever used herein, means the payment, as a Project Cost, to Company or any property manager it selects for the administration and management of the Premises and the Facilities. Such fee shall be three percent (3%) of Total Revenue received by Company from Tenants and Sublessees each month during the term of this Agreement. The Management Fee shall be paid to Company or its property manager on a monthly or other basis selected by Company, as funds are available from Total Revenue to pay the Management Fee, before any payments of Rent or repayment of Equity Contributions.
- 1.1.17 The term "Maintenance and Operations expense," whenever used herein, means the expense for maintenance, operation, administration and repair of the Premises and the Facilities, including, but not limited to, repair, maintenance or replacement of the plumbing, heating, ventilating and air conditioning systems, electrical systems, lighting facilities, fire protection systems, utility installations, fixtures, walls, foundations, roof, ceilings, floors, structural systems, doors, glass, skylights, landscaping and irrigating systems, driveways, parking lots, fences, retaining walls, signs, sidewalks, and the cost of all janitorial service, trash disposal, water, gas, electricity, and other utilities, together with any taxes thereon.

- 1.1.18 The term "Net Revenue," whenever used herein, means the amount of cash available after deducting from Total Revenue in the following order of priority: (a) Debt Service; (b) all Project Costs incurred during such calendar year of the term of this Agreement; (c) reasonable reserves for future Maintenance and Operations expense, future Capital Improvement expense or any reserve required by any Leasehold Mortgagee and (d) the repayment of Equity Contributions and interest thereon.
- 1.1.19 The term "Participating Parties" or "Parties," whenever used herein, means Company as lessee and the Successor Agency to the Industry Urban-Development Agency as Lessor (hereinafter jointly referred to as "Parties"), their successors and assigns.
- 1.1.20 The term "Premises," whenever used herein, means that area described on Exhibit "A". Final legal descriptions of the Premises will be attached to the Memorandum of Lease described in Section 1.2.4.
- 1.1.21 Except as otherwise provided in this Agreement, the term "Project Cost," whenever used herein, means all necessary costs incurred by Company (excluding Rent) prior to or after the date hereof in connection with the subdividing, designing, constructing, owning, leasing, financing or managing the Premises or the Facilities and other improvements constructed on the Premises by Company, or its successors and assigns. Except as otherwise provided in this Agreement, Project Cost shall include (without duplication), but not be limited to, the following: (a) all architectural and engineering expense, (b) all plan check fees, building permit fees, school fees, drainage fees and all other governmental, railroad and utility fees, licenses and permit cost, (c) all costs of environmental impact reports, traffic studies, biological studies and all other reports and studies required by governmental agencies, railroads or utility companies, (d) all costs to subdivide and re-subdivide the Premises, (e) all costs incurred by Company in constructing the Facilities or other permitted improvements on the Premises, (f) all Capital Improvement Expense, (g) all Maintenance and Operations expense, (h) all Management Fees, (i) except as provided in Section 2.12.2.10, all costs incurred for uninsured losses, earthquake, flood or other casualty, or repairs or replacements to the Premises and the Facilities or other improvements and the unreimbursed portion of any insured losses, (j) except as provided in Article III, all costs of complying with Environmental Laws, (k) all insurance premiums for insurance required hereunder and property taxes, (1) all leasing commissions and all other marketing and advertising expense, (m) all legal and accounting fees, (n) all loan fees, points, appraisal fees and other costs associated with the obtaining of Leasehold Mortgages, (o) all interest on Leasehold Mortgages, (p) all costs incurred by Company to create a community facilities district or districts or other assessment or other public financing district or districts to finance the construction of street improvements, utilities or other public improvements upon the Premises and (q) all other expenses related to the use, maintenance, leasing, financing and operation of the Premises and the Facilities. Project Costs shall only include costs, expenses and other amounts to the extent the same are reasonable and competitive in amount. Without limiting the preceding sentence, any costs, expenses or other amounts specified in an Approved Budget or otherwise approved in writing by LDR shall be deemed a reasonable and competitive amount. Further, Project Cost shall not include any cost, expense or other amount which is included within any other cost, expense or amount which is permitted by this Agreement to be deducted from Total Revenue in calculating Net Revenue, the intention of the Parties being that no cost, expense or other amount which is so permitted to be deducted from Total Revenue be deducted therefrom more than once.

- 1.1.22 The term "Sublease," whenever used herein, means the documents signed by a Sublessee or Tenant of Company for the leasing of a portion of the Premises or space in the Facilities.
- 1.1.23 The term "Sublessee" or "Tenant," whenever used herein, means any individual, corporation, limited liability company, trust, business, firm or other entity that leases or otherwise occupies or uses all or any portion of the Premises or the Facilities under a lease, rental agreement or other arrangement with Company.
- 1.1.24 The term "Release," whenever used herein, means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, disposing or dumping of any Hazardous Material in violation of Environmental Laws.
- 1.1.25 The term "Total Revenue," whenever used herein, means the total amount of all rents, charges, fees and/or other income derived or received, directly or indirectly, by Company from the use, operation and/or leasing of all or part of the Premises or the Facilities, including, without limitation, all rents, charges, fees and/or other income and amounts received from Sublessees or Tenants of all or any part of the Premises or the Facilities during each full or partial calendar year of the term of this Agreement and the net proceeds received by Company as a result of a Total Taking, Partial Taking or Temporary Taking from an eminent domain proceeding referred to in Section 2.19. Insurance proceeds are excluded from Total Revenue as they are covered by other provisions of this Agreement.
- 1.1.26 The term "Equity Contribution" whenever used herein, means any funds provided by Company, Lessor or any Leasehold Mortgagee in accordance with the provisions of Section 1.7.1.1 or Section 1.7.1.2 hereof.
- 1.1.26 The term "Rent" whenever used herein, means fifty percent (50%) of Net Revenue during each calendar year of the term of this Agreement.

1.2 TERM AND CONDITIONS PRECEDENT

- 1.2.1 The term of this Agreement shall commence upon the expiration of 30 days following approval by the First District Consolidated Oversight Board and the California Department of Finance, if, and/or, as required by law.
- 1.2.2 The term of this Agreement will expire upon the earlier of: (a) sixty five (65) years from the recording of a Notice of Completion on the first building constructed by Company on the Premises or (b) sixty eight (68) years from the Commencement Date.
- 1.2.3 Lessor and Company agree to execute a memorandum of this Agreement evidencing the existence of this Agreement, the ownership of the Facilities by Company, the rights of Company in the Premises, the fact that this Agreement contains a right of first refusal to purchase the Premises and setting forth the Commencement Date and Termination Date of this Agreement. When the first Notice of Completion is recorded, Company and Lessor shall execute and record an amended memorandum of this Agreement setting forth the exact Termination Date.

Company will be entitled to receive, as a Project Cost, an ALTA leasehold policy of title 1.2.4 insurance issued by a title company selected by Company, with liability in an amount reasonably determined by Company and insuring Company's interests hereunder. Such leasehold policy will be subject only to exceptions permitted by Company. By September 1, 2019, Lessor shall deliver a preliminary title reports covering the Premises, as well as copies of the underlying document listed as exceptions to title is such report, to Company. If Company objects to any exceptions, it will give written notice of such objections (the "Notice of Objections") to Lessor within such one hundred twenty (120) days after receipt of the preliminary report and the underlying documents. Company's failure to give a Notice of Objections within such One Hundred Twenty (120) day period shall be conclusively deemed to be Company's acceptance of all exceptions. Within thirty (30) days following receipt of a Notice of Objections, Lessor, at its option, may cure, or agree to cure prior to the commencement of the term, some or all of the exceptions that are the subject of the Notice of Objections. If at the end of such thirty (30) day period, Lessor has not cured, or agreed to cure prior to the commencement of the term, any or all of such exceptions, Company may elect by written notice given within fifteen (15) days thereafter to terminate this Lease (the "Termination Notice"). Company's failure to give the Termination Notice within such fifteen (15) day period shall be conclusively deemed to be Company's election to accept such leasehold policy subject to all then uncured exceptions. Upon any termination of this Agreement pursuant to the provisions of this Section, the parties shall be released from all further liabilities and obligations under this Agreement, except for such as have accrued prior to the date of such termination. Notwithstanding the foregoing, Lessor agrees to remove, prior to the commencement of the term of this Lease, all loans secured by deeds of trust or mortgages encumbering the Premises as well as any leases or rental agreements encumbering all or any portion of the Premises. Lessor agrees not to create or permit the creation of any additional exceptions to title, after the date of the preliminary title report, without the written consent of Company.

As Facilities are constructed on the Premises, Company shall be entitled to obtain, as a Project Cost, additional ALTA leasehold title insurance with liability in an amount reasonably determined by Company and insuring Company's interests and the interest of any Leasehold Mortgagee hereunder.

1.2.5 Any amendments to this Agreement will not be effective as against any Leasehold Mortgagee unless Company has obtained the written consent of such Leasehold Mortgagee.

1.3 PREMISES

1.3.1 Lessor does hereby demise and let unto Company and Company does hereby take from Lessor that certain real property (hereinafter referred to as "Premises") as follows:

The land described on Exhibit "A" which is attached hereto and made a part hereof, together with all improvements located thereon (which improvements are owned by Company) together with all appurtenances, rights, privileges and easements appertaining thereto.

1.3.2 Lessor shall cause the work described on Exhibit "B" to be performed in accordance with the provisions of Exhibit "B" which is attached hereto and made a part hereof. Lessor shall commence such work within thirty (30) days after the date of this

Agreement and shall complete such work as soon as reasonably possible, subject to extensions due to force majeure events as provided in Section 4.1. Lessor shall use commercially reasonable efforts to coordinate the timing of such work with Company's development of the Premises. All such work shall be at Lessor's sole cost and expense and shall not be considered an Equity Contribution by Lessor.

1.4 USE OF PREMISES

- 1.4.1 Upon performance of the agreements, provisions and conditions contained in this Agreement, Company will have the use of the Premises for the redevelopment of the Premises and construction and operation of Facilities and for any other business activities related thereto and for retail, warehouse, industrial, office and any other use permitted by applicable zoning regulations and for no other purposes, unless approved in writing by the LDR. So long as no default has occurred and is continuing hereunder, Lessor covenants peaceful and quiet enjoyment of the Premises by Company. Lessor covenants and warrants that the Premises are not now affected by any covenants, restrictions, easements or agreements which prohibit the construction, operation and maintenance upon the Premises of commercial structures with appurtenant parking, nor do the same prevent access to and from the Premises and the roadways adjoining the Premises.
- 1.4.2 Company acknowledges that the Rent payable to Lessor under this Agreement is directly dependent upon the generation by Company of Net Revenue, and that the failure by Company to use commercially reasonable efforts to maximize Net Revenue will have a direct and adverse effect upon Lessor's economic return under this Agreement. Therefore, as a material inducement to Lessor to enter into this Agreement, Company agrees that it will at all times construct, manage, lease and operate all Facilities in a manner and through the use of commercially reasonable efforts so as to maximize Net Revenue throughout the term of this Agreement.

1.5 DEVELOPMENT AND SUBLEASING

- 1.5.1 Company will develop the Premises and construct the Facilities substantially in accordance with plans and specifications prepared by Company in accordance with applicable governmental statutes, ordinances, rules and regulations, including those of the City of Industry and the Successor Agency to the Industry Urban-Development Agency.
- 1.5.2 Company may enter into subleases with Sublessees or Tenants on terms and conditions that are satisfactory to Company, provided such subleases are on sublease forms that have been approved by the LDR. LDR must approve or disapprove any sublease forms within 15 days of submittal by Company. Failure of the LDR to give written notice of disapproval, stating the specific objections to any such form, within the time specified shall be deemed approval.
 - 1.5.2.1 Company may negotiate changes to such approved forms which are typically given to Sublessees or Tenants in similar transactions without the consent of Lessor. In the event there are substantive changes or exceptions to the standard forms of sublease arrangements which are not typically given to Sublessees or Tenants in similar transactions, the Company must obtain the written approval of LDR, which approval will not be unreasonably withheld. LDR must approve or disapprove any such requested change within 2 business days of submittal by Company.

Failure of the LDR to give written notice of disapproval, stating the specific reason for disapproving the requested change within the time specified shall be deemed approval.

- 1.5.2.2 All Subleases must be for purposes as permitted in Section 1.4, entitled USE OF PREMISES, of this Agreement and shall expressly provide that they are subject and subordinate to all provisions of this Agreement.
- 1.5.2.3 Except pursuant to Subleases as provided in this Section 1.5, Company shall not suffer or permit the use or occupancy of all or any portion of the Premises by any person or entity.
- 1.5.2.4 Rental interruption insurance or rental abatement insurance (as the case may be) shall be maintained (if available) by Company or Sublessee's or Tenants' providing for the payment of at least six (6) months' rent, taxes, insurance and maintenance expenses payable by Sublessees or Tenants under its Sublease.
- 1.5.3 If requested by Lessor, the Company will provide Lessor with a copy of any rules, regulations or other standards of operation developed by Company and distributed to Sublessees and Tenants.
- 1.5.4 In the event that the sublease of a portion of the Premises constitutes a change of ownership for property tax purposes, it is the desire of the Parties that such change in ownership and reappraisal of the subleased property be limited to the particular portion of the Premises which are the subject of such sublease and not be deemed a change of ownership for assessment purposes for any remaining portion of the Premises.

1.6 BUDGET APPROVAL

- 1.6.1 Prior to the construction of any new building on the Premises, Company shall submit to Lessor copies of a written construction budget for such new building. Each such budget will show all projected Project Costs related to the design, construction, financing and leasing of such new building.
 - 1.6.1.1 Within fifteen (15) days of receipt of the proposed budget, the LDR will review and approve or disapprove the proposed budget submitted by the Company. If the LDR does not respond in writing within fifteen (15) days, the budget shall be deemed to be the Approved Budget.
 - 1.6.1.1.1 If disapproved, the LDR will inform Company in writing of its disapproval describing the disapproved provisions of the proposed budget or requesting further clarification of the budget elements. Company will respond within fourteen (14) days with verification of the budget elements or with a modified written budget that is reasonably satisfactory to the LDR. The Participating Parties agree to negotiate in good faith to resolve any conflicting issues that may arise. If the LDR fails to timely respond, the proposed budget will be deemed approved and will become an Approved Budget.

- 1.6.1.1.1 If, however, the Participating Parties cannot agree upon the provisions of the proposed budget and/or the elements contained in the proposed budget, or, if following approval of the Approved Budget, disputes arise concerning the Approved Budget which the Parties are unable to resolve through good faith negotiation, a neutral third party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises will be selected by the LDR to arbitrate the disputed terms.
- 1.6.1.1.2 If, however, the Company does not accept the neutral third party selected by the LDR, Company will be allowed to select a second neutral party with similar experience. The two selected parties will then select a third neutral party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises who will arbitrate the disputed terms.
- 1.6.1.1.3 The LDR and Company agree to be bound by the decisions reached by the selected arbitrator(s). The Participating Parties will cause the arbitrator(s) to make a determination within fourteen (14) days following submittal.
- 1.6.1.1.4 The Participating Parties agree that each will bear its own costs and expenses incurred for attorney's fees and all other preparation and presentation costs for the arbitration process. If only one arbitrator is selected, the Participating Parties will equally share the cost of such arbitrator. If three arbitrators are used, each Party will bear the cost of the arbitrator selected by it and the Participating Parties will equally share the cost of any third arbitrator.
- 1.6.1.2 The agreed upon budget will be deemed the Approved Budget for the proposed building.
- 1.6.2 Company will be entitled to expend funds in accordance with the Approved Budget for the construction of the proposed building. The cost of constructing any such building shall be a Project Cost. In the event Company is over-budget on a particular line item, Company may reallocate excess funds from one line item to another line item to cover any short falls in any line item. Any expenses not covered by the Approved Budget which increase the total cost of constructing such building by more than ten percent (10%) are subject to the reasonable written approval of LDR. Any Equity Contribution required to cover any such additional Project Cost shall be the responsibility of Company.

1.7 RENTALS AND EQUITY CONTRIBUTIONS

Rentals due under this Agreement and Equity Contributions for the construction and operation of the Facilities will be as follows:

1.7.1 Equity Contributions.

- 1.7.1.1 The Parties anticipate that from time to time during the term of this Agreement, the Company will obtain financing for the development of the Premises and the Facilities and other Project Costs in accordance with the terms and conditions of Section 2.18 entitled FINANCING. Any funds required for the completion of the initial development of the Premises and construction of the Facilities, in excess of the financing obtained by Company, shall be provided by Company as an Equity Contribution.
- 1.7.1.2 Any additional capital required for the operation or maintenance of the Premises and the Facilities or other Project Costs following completion of construction of the Facilities on any separate legal parcel of the Premises and any capital required to repay any Leasehold Mortgages thereon, shall be contributed fifty percent (50%) by Company and fifty percent (50%) by Lessor within thirty (30) days of the receipt of demand by Company for such capital, which demand shall be accompanied by information and evidence, in reasonable detail, substantiating the amount and reasons for such demand. If any party fails to make such Equity Contribution, the other party may make such Equity Contribution for the delinquent party.
- 1.7.1.3 In the event of default by Company and the subsequent foreclosure and sale of the leasehold interest to another party, or in the event of a deed in lieu of foreclosure, the total unpaid balance of the Leasehold Mortgage at the date of the foreclosure sale or recording of the deed in lieu of foreclosure, including all costs of foreclosure, shall be considered an Equity Contribution of the Leasehold Mortgagee or purchaser at foreclosure, as the case may be, as of the date of such foreclosure sale or the date of recording the deed in lieu of foreclosure.
- 1.7.1.4 Any portion of Total Revenue remaining after payment by Company of Debt Service and all costs (except Rent) associated with the ownership, operation, financing, maintenance and leasing of the Premises and the Facilities, during each calendar year of the term of this Agreement will be applied to the Parties Equity Contributions until such time as all Equity Contributions are repaid in full together with interest thereon. All Equity Contributions made in accordance with the provisions of Section 1.7.1.1 and all Equity Contributions made in accordance with Section 1.7.1.3 shall be repaid in full prior to the repayment of any Equity Contributions made in accordance with Section 1.7.1.2. All Equity Contributions shall bear interest from the date of contribution until repaid at the lower of the annual rate of interest publicly announced from time to time by Bank of America, N.A. as its "prime" rate of interest plus three (3%) percent or the maximum rate allowed by law. The interest rate payable hereunder shall

fluctuate with any change in the "prime" rate, and such fluctuations in the interest rate shall be effective on the effective date of each and every change in the "prime" rate as, from time to time, announced by Bank of America, N.A. as its "prime" rate of interest. Should Bank of America, N.A. no longer exist or fail to announce a "prime" rate, the "prime" rate, for purposes of this Lease, shall be the "prime" rate published in The Wall Street Journal, or if The Wall street Journal fails to publish a "prime" rate, the rate that most closely approximates the "prime" rate published in The Wall Street Journal or other similar financial publication.

1.7.1.5 The Participating Parties will acknowledge the date any equity contribution is paid in full by written notice from the Company and acknowledgment by the LDR.

1.7.2 Rent.

- 1.7.2.1 Fifty percent (50%) of all Net Revenue, if any, during each calendar year or partial calendar year during the term of this Agreement shall be paid to Lessor as Rent for the Premises.
- 1.7.3 Rent shall be paid to Lessor in monthly installments on or before the twenty-fifth (25th) day following the end of each calendar month. Such payments shall be accompanied by a written statement setting forth in detail the sources and amount of Total Revenue received for the preceding month and allowable deductions for the Net Revenue calculation. A check for the Lessor's fifty percent (50%) share of Net Revenue will be submitted with such statement. Except as otherwise provided in this Agreement, all Rent shall be paid without deduction, offset or abatement.
- 1.7.4 Company will make all payments to the Lessor at 15625 East Stafford Street, Suite 200, City of Industry, California 91744 or to such other place as Lessor may direct Company in writing.
- 1.7.5 In addition to, and without limiting, any other right or remedy which Lessor may have under this Agreement or at law or in equity, in the event any required payment is not made by Company to Lessor as required and remains unpaid for a period of thirty (30) days or more, the Lessor will be entitled to, and Company will pay to the Lessor, interest at the lower of ten percent (10%) per annum or the maximum amount allowed by law on all amounts unpaid and which remain unpaid thirty (30) days past the due date.
- 1.7.6 On or prior to April 30, annually during the term of this Agreement and within one hundred twenty (120) days after the expiration of the term of this Agreement, Company will provide Lessor with a statement showing in detail the sources and amount of Total Revenue received for the entire preceding calendar year. Such statement shall be prepared by the Company's chief financial officer in accordance with sound accounting principles and practices and shall contain a written opinion of such officer as to whether the Total Revenue calculations and distribution of Net Revenue and Rent has been made in accordance with the provisions of this Agreement. Should such statements show that the amount of Rent paid during the period of review was less than that which was due, such statement shall be accompanied by Company's payment of the additional amount to Lessor. Should such statement show that Company paid Lessor more than was due, after review and verification by LDR, a credit will be issued to be applied against future

monthly installments of Rent, except that if such should be the case at the end of the last month of this Agreement, Lessor will refund the overpayment to Company.

1.8. RECORDS AND AUDIT

- 1.8.1 Throughout the term of this Agreement, Company shall keep and maintain, in accordance with sound accounting principles and practices, accurate and complete books, records and accounts of Total Revenue and all items constituting deductions for purposes of calculating Net Revenue. Within ten days of request by Lessor, Company agrees to provide for inspection and copying by the Lessor or its designated representatives, at a location in the metropolitan area of Los Angeles County, California, such books, records, and accounts for the then current and three (3) immediately preceding calendar years. Such inspection shall take place Monday through Friday, 9:00 AM to 5:00 PM, holidays excluded.
- 1.8.2 Lessor will, at any time, have the right to cause an audit of the Company's operation of the Premises and of Total Revenue and all items constituting deductions for purposes of calculating Net Revenue, for the then current or three (3) immediately preceding calendar years. Such audit shall be made by a Certified Public Accountant of Lessor's selection. If such audit discloses that the calculation of Net Revenue previously provided to Lessor by Company is understated (either intentionally or unintentionally) by a greater margin than one percent (1%) of Company's Total Revenue for the period of review, then Company will immediately pay to Lessor the cost of such audit; otherwise the cost of the audit will be paid by Lessor. In all events, Company shall pay to Lessor the additional payments shown to be payable to Lessor by Company.

1.9 IMPROVEMENTS, MAINTENANCE AND REPAIR BY LESSOR

1.9.1 Except as provided in Exhibit "B" and Section 3.1.2 dealing with contamination of the Premises, Lessor has no direct responsibility or obligation for any maintenance, repair or replacement of the Premises or improvements.

1.10 CONSTRUCTION OF IMPROVEMENTS BY COMPANY

- 1.10.1 Company will, construct and install the Facilities including, grading, fencing, paving, lighting, driveways, roads, railroad facilities, utilities, parking lots, drainage, buildings and other improvements.
- 1.10.2 In the event Company has not commenced the construction of Facilities or submitted a proposal for the construction of Facilities on a parcel of the Premises within 10 years after the Commencement Date, Lessor shall have the right to terminate this Agreement as to any such undeveloped parcel of the Premises. Lessor shall give Company and any Leasehold Mortgagee encumbering any such undeveloped parcel of the Premises ninety days prior written notice and the right to submit a proposal for the development of any such parcel or commence construction thereon before exercising its right to terminate. In the event a proposal for the development of such undeveloped parcel is submitted within said ninety days, Lessor shall not have said right to terminate. Lessor agrees not to exercise its right to terminate as to any such undeveloped parcel until any Leasehold

- Mortgagee encumbering such parcel has been given its rights to cure or foreclose as set forth in Section 2.18.
- 1.10.3 All improvements or alterations by Company will be in accordance with all applicable governmental rules and regulations. Upon completion of any initial or additional improvements during the term hereof, Company will provide as-built drawings of same to the Lessor along with a certification of construction costs for all permanent improvements.
- 1.10.4 At any time during the term or any extension of this Agreement, Company may, as a Project Cost, add to or alter initially constructed Facilities, subject to all conditions set forth in 1.10.3 above. Any such addition or alteration will be performed in a workmanlike manner in accordance with all applicable governmental regulations and requirements and will not weaken or impair the structural strength or reduce the value of the Premises or improvements thereon.

1.11 MAINTENANCE AND REPAIR

- 1.11.1 Throughout the term of this Agreement, Company shall maintain, or cause to be maintained, as a Maintenance and Operations expense, the Premises and the Facilities in good condition and repair (including the making of any necessary replacements, both structural and nonstructural), ordinary wear and tear excepted, and in accordance with all applicable laws, rules, regulations and orders of all (i) federal, state, county, municipal, and other governmental agencies and bodies having jurisdiction, (ii) all insurance companies insuring all or any part of the Premises or the Facilities or both. Company's obligation shall include, but not be limited to, all structural and nonstructural portions of the Premises and the Facilities, all heating, ventilating and air conditioning systems and escalators, parking lots, driveways, driving lanes, sidewalks, walkways, common areas and landscaped areas.
- 1.11.2 If Company shall fail to perform any of its obligations under this Section 1.11, Lessor may, but shall not be obligated to, give Company written notice specifying such failure in reasonable detail. If Company does not remedy such failure, or does not commence to remedy such failure and thereafter diligently pursue such remedy to completion, within fifteen (15) days following receipt of such notice, Lessor may, but shall not be obligated to, remedy such failure. If Lessor remedies any such failure, all reasonable costs and expenses incurred by Lessor in connection therewith shall be paid to Lessor by Company within ten (10) days following written demand, which shall be accompanied by copies of receipts for all such expenses incurred. Following such payment, the amount paid may be included as a Maintenance and Operations expense.

1.12 APPROVALS TO BE REASONABLY GIVEN

It is understood and agreed that all provisions of this Agreement that require approval by the Lessor or the LDR will receive timely response and such approvals will not be unreasonably withheld. Unless another time limit is otherwise specified in this Agreement, any approval or disapproval must be given within twenty (20) days. If the Lessor or LDR does not respond in writing within twenty (20) days, the item for which Company requests approval shall be deemed approved. If disapproved, the LDR will inform Company in writing of its disapproval and state the reason for such disapproval. The Participating Parties agree to negotiate in good faith to resolve any conflicting issues that may arise. If, however, the Participating Parties cannot agree

upon the item that was disapproved, a neutral third party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises will be selected by the LDR to arbitrate the disputed issue. If, however, the Company does not accept the neutral third party selected by the LDR, Company will be allowed to select a second neutral party with similar experience. The two selected parties will then select a third neutral party with at least 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises who will arbitrate the disputed issue. The LDR and Company agree to be bound by the decisions reached by the selected arbitrator. The Participating Parties will cause the arbitrator to make a determination within fourteen (14) days following submittal. The Participating Parties agree that each party will bear its own costs and expenses incurred for attorney's fees, preparation and presentation costs for the arbitration process. The Participating Parties will share the cost of any third arbitrator.

Notwithstanding the foregoing, in the event of an emergency, the Company may take immediate, commercially reasonable action to prevent loss, injury or damage to persons or property or to preserve the Premises and the Facilities without the prior approval of Lessor and any costs reasonably incurred in connection therewith shall be a Project Cost.

ARTICLE II

2.1 ASSIGNMENT

- 2.1.1 Company will not assign this Agreement or its rights or duties hereunder or any estate created hereunder, in whole or in part, except with the prior written approval of Lessor, which approval will not be withheld unreasonably or delayed; provided, that it shall be reasonable for Lessor to withhold its approval if the assignee presented is not a proper and fit person or entity with financial resources and demonstrated business experience sufficient in Lessor's reasonable business judgment to be capable of performing the obligations hereunder which are the subject of the assignment. Further, any such assignment will be specifically subject to all provisions of this Agreement. Any assignment without the prior written approval of Lessor is Void.
 - 2.1.1.1 Except as provided in Section 2.1.1.3, any voluntary or involuntary transfer of fifty percent (50%) or more of Company's or any permitted successor's or assign's voting common stock, or the transfer of fifty percent (50%) or more of the partnership or membership interest, or the acquisition or transfer of fifty percent (50%) or more of Company's ownership, or the transfer of substantially all of the assets of the Company or any such successor or assign will be deemed an assignment requiring the prior written approval of Lessor.
 - As a condition precedent to any assignment, Company shall provide Lessor reasonable prior written notice of the proposed assignment with such appropriate documentation and other information as Lessor shall reasonably request in order for Lessor to make its decision to grant or withhold approval. If the prior written approval of Lessor is given to any such assignment, the proposed assignee shall, in recordable form, expressly assume all of the terms, covenants and conditions of this Agreement which are the subject of the assignment. Any assignment will not release Company from its obligations under this Agreement arising from events occurring prior to the date of assignment,

but shall release Company from all obligations arising after the effective date of the assignment.

- 2.1.1.3 Any transfers by partners or members of Company or shareholders of partners or members of Company to each other or for estate purposes or upon death will not be considered an assignment hereunder.
- 2.1.2 The foregoing provisions relating to assignment shall not be applicable to a Leasehold Mortgage hypothecating or encumbering Company's interest in this Agreement and the leasehold estate created thereby and the Facilities located on the Premises and further shall not be applicable to any sale by judicial foreclosure or pursuant to a power of sale by any Leasehold Mortgagee or to any transfer in lieu of such sale, or to any sales or transfers by any Leasehold Mortgagee (or any affiliate, parent or subsidiary of any Leasehold Mortgagee that acquires title to the Company's interest in this Agreement and the leasehold estate created thereby and the Facilities as a result of a sale by judicial foreclosure or pursuant to a power of sale or any transfer in lieu of such sale), subsequent to a sale by judicial foreclosure or pursuant to a power of sale or subsequent to any transfer in lieu of such sale. Any further assignments shall be subject to the provisions of Section 2.1, 2.1.1.1, 2.1.1.2 and 2.1.1.3.

2.2 SUBLEASING

Company will not sublease, sublet or rent to, or permit any persons, firms or corporations to occupy any part of the Premises or the Facilities without having complied with the following:

2.2.1 Company may enter into any Subleases which comply with Section 1.5 entitled DEVELOPMENT AND SUBLEASING. A copy of any such executed Sublease will be provided to the LDR. Any arrangements for the leasing of space which are not in conformance with Section 1.5 entitled DEVELOPMENT AND SUBLEASING shall not be entered into without the prior written approval of the LDR.

2.3 ATTORNMENT

- 2.3.1 All Subleases entered into by the Company will be subject to all terms and conditions of this Agreement. If Company defaults under this Agreement, all Sublessees will recognize Lessor as the successor to the Company under their respective Subleases, and will render performance thereunder to Lessor as if the Sublease were executed directly between Lessor and the Sublessees; provided, however, Lessor agrees that so long as a Sublessee is not in default under its Sublease, Lessor will provide quiet enjoyment to such Sublessee and Lessor agrees to be bound by all of the terms and conditions of each such Sublease.
- 2.3.2 All Subleases entered into by the Company will contain the following provision:

If, by reason of a default on the part of Company in the performance of the terms and provisions of the underlying Agreement, the underlying Agreement and the leasehold estate of Company thereunder is terminated by summary proceedings or otherwise in accordance with the terms of the underlying Agreement, Sublessee will attorn to Lessor and recognize Lessor as lessor under this Sublease; provided, however, Lessor agrees that

so long as Sublessee is not in default, Lessor agrees to provide quiet enjoyment to Sublessee and to be bound by all the terms and conditions of this Sublease.

2.3.3 In the event this Agreement is terminated for any reason, all Sublessees will be liable to Lessor for their payment of rents and fees and will be subject to all the provisions and terms contained in their Sublesses.

2.4 SUCCESSORS AND ASSIGNS

Subject to the provisions of this Article II, this Agreement shall inure to the benefit of and be binding upon the legal representatives, successors and assigns of the Parties.

2.5 CONTROL OF PERSONNEL

Company will, in and about the Premises, exercise reasonable control over the conduct, demeanor and appearance of its employees, agents and representatives and the conduct of its contractors and suppliers. The provisions of this Section 2.5 shall in no way limit the liability of the Company with respect to the negligence, acts or omissions of its employees, agents, representatives, contractors and suppliers, nor shall such provisions limit any obligations of indemnity on the part of the Company under this Agreement.

2.6 SIGNS

Company and its Sublessees and Tenants shall have the right to install, erect and operate signs upon the Premises so long as such signs are installed, erected, operated and maintained in compliance with all applicable laws and ordinances.

2.7 ENTRY AND INSPECTION OF PREMISES

- 2.7.1 Lessor, its authorized officers, employees, agents, contractors, subcontractors or other representatives will have the right to enter upon all portions of the Premises and the Facilities constituting public areas when the same are open to the public, and shall have the right to enter upon all other portions of the Premises and the Facilities during normal business hours, for the purpose of inspecting the Premises and the Facilities, determining whether Company is in compliance with the terms and provisions of this Agreement, and for fulfilling Lessor's obligations and exercising Lessor's rights hereunder; provided however, that such entry will be in such manner as to not unreasonably interfere with the operations or reasonable security requirements of Company or its Sublessees. Except in the event of an emergency, Lessor shall provide Company at least two (2) business days prior written notice of its intention to enter upon the non-public areas of the Premises or the Facilities for such purposes, and, if Company so elects, Lessor shall be accompanied by a representative of Company during such entry.
- 2.7.2 No such entry upon the Premises by or on behalf of Lessor will cause or constitute a termination of this Agreement nor be deemed to constitute an interference with the possession thereof nor constitute a revocation of or interference with any of Company's rights in respect thereof for exclusive use of the Premises.
- 2.7.3 The rights of entry for inspection and other purposes as contemplated by the parties to this Agreement, pursuant to this Section, are for the sole benefit of the parties. No benefit to any third party is contemplated or intended.

2.8 INTENTION OF PARTIES

This Agreement is intended solely for the benefit of Lessor and Company and is not intended to benefit, either directly or indirectly, any third party or member(s) of the public at large. Any work done or inspection of the Premises by Lessor is solely for the benefit of Lessor and Company.

2.9 LIENS

Subject to the rights to contest set forth in Section 2.10 entitled Taxes and Assessments, Company will cause to be removed any and all tax liens and liens arising out of or because of any construction or installation performed by or on behalf of Company or any of its contractors or subcontractors upon the Premises or arising out of or because of the performance of any work or labor to it or them at the Premises or the furnishing of any materials to it or them for use at the Premises. Should any such lien be made or filed, Company will bond against or discharge the same within thirty (30) days after written request by LDR. The cost of bonding against or discharging any liens relating to construction or installation of the Facilities shall be a Project Cost.

2.10 TAXES AND ASSESSMENTS

Throughout the term of this Agreement, Company shall pay before delinquency, as a Project Cost, all real and personal property taxes and assessments, including, without limitation, all city, county, school district and other taxes and general and special assessments, levied upon or assessed against the Premises, any Facilities now or hereafter located thereon, the leasehold estate, any personal property of Company located on or in the Premises or any Facilities, or arising in respect of the occupancy, use or possession of the Premises or the Facilities located thereon and which are assessed or become due during the term of this Agreement. All such taxes for the first and last years of the term hereof shall be equitably prorated between the parties. Company shall have the right, in Company's or Lessor's name, to contest the validity of any tax or assessment by appropriate proceedings timely instituted and diligently prosecuted. Lessor, if requested by Company, shall cooperate with Company in any such proceedings; provided, that all costs and expenses (including attorneys' fees and costs) incurred by Lessor in connection therewith shall be paid by Company as a Project Cost. Company shall not be responsible for any of Lessor's franchise, inheritance, income, succession, transfer or other tax levied on the Lessor's rights in and to the Premises or Lessor's right to receive income from the Premises or under this Agreement. Company will keep current municipal, state or local licenses or permits required for the conduct of its business.

2.11 INDEMNITY

During the term of this Agreement, Company agrees to indemnify and hold Lessor forever harmless from and against all claim, liability, loss, demand, judgments or other expense (including, but not limited to, defense costs, expenses and reasonable attorney fees) imposed upon Lessor by reason of injuries or death of persons (including wrongful death) and damages to property caused during and because of Company's use or occupancy of the Premises or any actions or non-actions of Company, its officers, employees, agents, or other representatives, including movement of vehicles, provided, however, that (without limiting Lessor's right to coverage under any applicable insurance policies) such indemnity will not apply as to any negligent or intentional act or omission of Lessor, its employees, agents or representatives.

During the term of this Agreement, Lessor agrees to indemnify and hold Company harmless from and against all claim, liability, loss, demand, judgments or other expense (including, but not limited to, defense costs, expenses and reasonable attorneys fees) imposed upon Company by reason of injuries or death of persons (including wrongful death) and damages to property caused by Lessors use or occupancy of the Premises or any actions or non-actions of Lessor, its officers, employees, agents, contractors or other representatives, including movement of vehicles, provided, however, that (without limiting Company's right to coverage under applicable insurance policies) such indemnity will not apply to any negligent or intentional act or omission of Company, its employees, agents or representatives.

2.12 INSURANCE AND BONDS

2.12.1 Bonds

- 2.12.1.1 No bonds will be required for any Facilities constructed by Commerce Construction Co. Otherwise, Company, as a Project Cost, will require its general contractor on any construction contract in excess of two million dollars to furnish bonds covering the faithful performance of the construction of the improvements or installation of equipment, payment of all obligations arising thereunder to take effect upon completion of the project. Bonds may be secured through the contractor's usual sources provided the surety is authorized and licensed to do business in the State of California. Company will be allowed to name any Leasehold Mortgagee as an additional obligee under any such bond.
- 2.12.1.2 Prior to execution of a construction contract with a contractor required to furnish bonds, and not later than ten (10) calendar days after notification of award, Company will require such contractor to furnish contract bonds to LDR as follows:
 - a. Labor and material payment bond in the amount of one hundred percent (100%) of the contract price.
 - b. Payment and performance bond in the amount of one hundred percent (100%) of the contract price.

LDR may, if considered appropriate in its sole judgment, waive the requirements of this Section 2.12.1.2 upon written request by Company.

- 2.12.1.3 The bonds referred to in Section 2.12.1.1 and 2.12.1.2 above will be written on payment and performance bond and labor and material payment bond forms generally used for similar projects in Southern California and approved by LDR.
- 2.12.1.4 Company will require its contractor to require the attorney-in-fact who executes the required bonds on behalf of the surety to affix thereto a certified and current copy of his power of attorney.

2.12.2 Insurance

2.12.2.1 Before any construction contractor (including Commerce Construction Co.) Commences Construction of any Facilities or any equipment installation on or about the Premises, the construction contractor shall procure and maintain insurance for such construction and installation protecting both Company and Lessor, as well as the construction contractor. Such insurance will provide coverage and limits as are commercially reasonable in the industry. Such insurance will include, but is not limited to:

General Liability on an "occurrence" basis only
Automobile Liability
Builder's Risk equal to the maximum contract cost of the project.

- 2.12.2.2 Company's (or its contractor's) insurance will be primary as respects
 Lessor and Company, their officers, employees and authorized agents.
 Any other coverage carried by or available to Lessor, its officers,
 employees and authorized agents will be excess over the insurance
 required of the Company or its contractor and shall not contribute with it.
- 2.12.2.3 Company and its contractors will maintain worker's compensation insurance in the amounts and form as required by California law.

 Certificates evidencing the valid, effective insurance policies will be provided to LDR.
- 2.12.2.4 Company will keep insured with responsible insurance underwriters the Facilities and any improvements from time to time constructed by it upon and within the Premises to the extent of not less than the full replacement value of such improvements from time to time (without deduction for depreciation) using an industry standard all risk form of protection. Company will use commercially reasonable efforts to cause its Sublessees or Tenants to keep insured with responsible insurance underwriters any improvements constructed by such Sublessees or Tenants to the same extent.
- Company will obtain and keep in full force and effect a policy(s) of 2.12.2.5 general liability insurance on an "occurrence" basis only and not "claims made." The coverage must be provided either on an ISO Commercial General Liability form, or an Accord Comprehensive General Liability form, or equivalent, approved by the LDR and Company. Any exceptions to coverages must be fully disclosed on the required Certificate. If other than these forms are submitted as evidence of compliance, complete copies of such policy forms will be submitted to LDR within ten (10) days after notice to Company. Policies must include, but need not be limited to, coverages for bodily injury, property damage, personal injury, Broad Form property damage, premises and operations, severability of interest, products and completed operations, contractual and independent contractors with no exclusions of coverage for liability resulting from the hazards of explosion, collapse, and underground property damage.

Company will maintain limits of no less than five million dollars (\$5,000,000) combined single limit per occurrence for bodily injury (including death), personal injury and property damage.

- 2.12.2.6 Company will furnish Automobile Liability coverage for claims for damage because of bodily injury or death of any person, or property damage arising out of the ownership, maintenance or use of any motor vehicles whether owned, hired or non-owned. Company will maintain limits of no less than five million dollars (\$5,000,000) combined single limit "per accident" for bodily injury and property damage.
- 2.12.2.7 All required insurance coverage as stated in Section 2.12.2 will be a Project Cost and evidenced by a current Certificate(s) of Insurance. Such Certificates will include, but will not be limited to, the following:
 - 2.12.2.7.1 All Certificates for each insurance policy are to be signed by a person authorized by that insurer and licensed by the State of California.
 - Each insurance company's rating as shown in the latest Best's Key Rating Guide will be fully disclosed and entered on the required Certificates of Insurance. If the insurance company providing the coverage has a Best rating of less than A+, the adequacy of the insurance supplied by Company (or its contractor), including the rating and financial health of each insurance company providing coverage, is subject to the approval by the LDR. Such approval will not be unreasonably withheld.
 - 2.12.2.7.3 Company (or its contractor) will furnish renewal Certificates for the required insurance during the period of coverage required by this Agreement. Company (or its contractor) will furnish renewal Certificates for the same minimum coverages as required in this Agreement.
 - 2.12.2.7.4 Lessor, its officers and employees shall be covered as additional insureds with respect to liability arising out of the activities by or on behalf of the named insured in connection with this Agreement. All property insurance policies will contain a waiver of subrogation clause in favor of Lessor.
 - 2.12.2.7.5 Each insurance policy supplied by the Company (or its contractor) must be endorsed to provide that the amount and type of coverage afforded to the Lessor by the terms of this Agreement will not be suspended, voided, canceled or reduced in coverage or in limits except after thirty (30) days' prior written notice by mail to Company, Lessor and any Leasehold Mortgagee.
 - 2.12.2.7.6 Any deductible, as it relates to coverage provided under this Agreement, will be fully disclosed on the

Certificates of Insurance. Any deductible provided will be reasonable and customary for the type of risk.

- 2.12.2.7.7 If aggregate limits are imposed on the insurance coverage, then the amounts of such limits must be not less than five million dollars (\$5,000,000) per occurrence or per accident. All aggregates must be fully disclosed and the amount entered on the required certificate of insurance. Company's insurer must notify the LDR of any erosion of the aggregate limits. The "per occurrence" limits of insurance required herein must be maintained in full, irrespective of any erosion of aggregate. A modification of the aggregation limitation may be permitted if it is deemed necessary and approved by the LDR and Company.
- 2.12.2.8 If the Company fails to maintain any of the insurance coverages required herein, then the Lessor, after 10 days prior notice to Company, may purchase replacement insurance or pay the premiums that are due on existing policies in order that the required coverages may be maintained. The Company is responsible for any expenses paid by the Lessor to maintain such insurance and Lessor may collect the same from the Company.
- 2.12.2.9 The insurance requirements specified herein do not relieve the Company (or its contractor) of its responsibility or limit the amount of its liability to the Lessor or other persons and the Company is encouraged to purchase such additional insurance as it deems necessary.
- 2.12.2.10 Company (or its contractor) is responsible for and must remedy all damage or loss to any property, including property of Lessor, caused in whole or in part by the Company or its contractor, any subcontractor or anyone employed, directed or supervised by the Company; provided, however neither Company nor any Leasehold Mortgagee shall be liable for any uninsured or underinsured loss relating to the Premises or the Facilities, unless such loss is a result of Company's failure to obtain the insurance it is required to maintain under this Agreement, in which event Company shall be liable for such loss and such loss shall not be a Project Cost. Company is responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with this Agreement.
- 2.12.2.11 Company and Lessor agree and acknowledge that any Leasehold Mortgagee shall be named as an additional insured under any insurance policy relating to the Premises, the Facilities or otherwise required under this Agreement. Company agrees to provide any Leasehold Mortgagee with appropriate certificates of insurance relating to all policies.
- 2.12.2.12 Company, at its option, may satisfy its obligations hereunder to insure within the coverage of any so-called blanket policy or policies of insurance which it now or hereafter may carry, by appropriate

amendment, rider, endorsement or otherwise; provided, however, that the interests of Lessor shall thereupon be as fully protected by such blanket policy or policies as they would be if this option to so insure by blanket policy were not permitted.

2.12.2.13

Waiver of Subrogation. Without affecting any other rights or remedies, Company and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is not limited by the amount of insurance carried or required, or by any deductibles applicable thereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Company, as the case may be, so long as the insurance is not invalidated thereby.

2.13 DAMAGE AND DESTRUCTION

In the event that any Facilities located on the Premises are damaged or destroyed, and the net proceeds of insurance are sufficient to cover the Company's estimated cost of rebuilding (or if Company failed to maintain insurance it is required to maintain under this Agreement and the proceeds of such insurance, had it been maintained, would have been sufficient to cover the Company's estimated cost of rebuilding), Company shall repair and rebuild the said Facilities, and this Agreement shall continue in full force and effect, and Company shall commence and continue such repair or rebuilding with reasonable diligence and shall complete such repair and rebuilding within a reasonable time after the same is commenced; provided, however, that any delay in the completion of the said repairs resulting from fire or other casualty, strikes, shortages of material or labor, governmental laws, rules and regulations, the elements or matters beyond the reasonable control of Company, shall extend such reasonable time within which Company may complete said repairs or rebuilding by the period of such delay. The net proceeds of any insurance, to the extent that such proceeds are received by Lessor or any Leasehold Mortgagee, less any expenses of recovery thereof (including attorneys' fees), shall be made available to Company to be applied to the cost and expense of repair or rebuilding the damage or destruction insured against, subject to reasonable conditions and payable on the usual architect's certificates. but Lessor or any Leasehold Mortgagee holding said insurance proceeds may withhold an amount reasonably necessary to insure completion of such repairs or rebuilding (not to exceed ten percent (10%) of such proceeds) until completion and the expiration of the period within which mechanics' or materialmen's liens may be filed and until receipt of satisfactory evidence that no liens exist.

In the event the Company maintains the insurance it is required to maintain under this Agreement and the proceeds of insurance are not sufficient to cover the Company's estimated cost of rebuilding, or in the event the damage or destruction occurs during the last 10 years of the lease term, Company will have the option to terminate this Agreement, subject to the rights of any Leasehold Mortgagee, which option will be exercisable by written notice to Lessor within forth five (45) days after the occurrence of such event. Company will only have such option to terminate this Agreement as to the specific legal parcel or parcels upon which the Facilities are damaged or destroyed. Any such termination by Company shall require the prior written consent of any Leasehold Mortgagee. In the event the Company elects to terminate this Agreement based upon such damage, destruction, or substantial loss, the Company will be liable for and will pay for all cleanup or demolition of the Premises necessary to make the Premises ready for repair,

replacement, restoration or rebuilding: (a) if the Company or its employees or agents caused such damage, destruction or substantial loss to occur, (b) to the extent of any insurance proceeds payable with respect to such damage, destruction or substantial loss, or (c) to the extent of any insurance proceeds which would have been payable with respect to such damage, destruction or substantial loss had Company maintained the insurance it is required to maintain under this Agreement. In the event Company does not exercise such option then Company will promptly repair, replace, restore or rebuild said improvements.

2.14 DEFAULT BY COMPANY

2.14.1 Default by Company

Subject to the provisions of Section 2.16 and Section 2.18, Company will be considered in default as Lessee under this Agreement in the event of any one or more of the following occurrences:

2.14.1.1 Company fails to pay the rent or any other money payments required by this Agreement when the same are due and the continuance of such failure for a period of thirty (30) days after written notice thereof from the LDR to Company or Company fails to maintain any insurance required to be maintained by it under this Agreement and such failure continues for a period of fifteen (15) days after written notice thereof from the LDR to Company.

2.14.1.2Company fails to perform any other covenant contained in this Agreement and such failure continues for a period of sixty (60) days after written notice thereof from the LDR to Company; provided, however, that if a nonmonetary default cannot be cured with reasonable diligence within such sixty (60) day period, then Lessor shall not have the right to terminate this Agreement or pursue any other remedy against Company, as long as Company commences the curing of such default within said sixty (60) day period and thereafter proceeds with the curing of such default to completion with reasonable diligence, with allowance for delays due to the action or failure to act of governmental authorities, strikes, acts of God or other matters beyond the reasonable control of Company.

2.14.1.3 Company voluntarily abandons the Premises for a period of sixty (60) days following written notice from the LDR to Company.

2.14.2 Cure

Subject to the mortgagee protection provisions set forth in this Agreement, Company will be considered in default of this Agreement if Company fails to fulfill any of the terms, covenants, or conditions set forth in this Agreement following the expiration of the cure period and written notice set forth in Sections 2.14.1 and 2.14.3.

2.14.3 Termination For Default By Company

Subject to the lender protection provisions of Section 2.18, if default is made by Company as described in Section 2.14.1 hereinabove, and such default is not cured as

provided in Sections 2.14.1 and 2.14.2, Lessor shall provide Company with an additional written notice thereof and if such failure to cure continues for an additional period of sixth (60) days. Lessor shall be entitled to pursue any and all rights and remedies which it may have under this Agreement or at law or in equity, including, without limitation, the right to terminate this Agreement. All of such rights and remedies shall be cumulative and not alternative.

- 2.14.3.1 If Lessor elects to terminate this Agreement, it will in no way prejudice the right of action for rental arrearages owed by Company.
- 2.14.3.2 Redelivery and disposal of improvements will be as described in Section 2.17, entitled SURRENDER AT END OF TERM.

2.15 DEFAULT BY LESSOR

2.15.1 Default By Lessor

Lessor will be considered in default of this Agreement if Lessor fails to fulfill any of the terms, covenants or conditions set forth in this Agreement if such failure shall continue for a period of more than sixty (60) days after delivery by Company of a written notice of such default.

2.15.2 Cure

If the default cannot be cured with reasonable diligence within such Sixty (60) day period, then Company shall not have the right to terminate this Agreement or pursue any other remedy against Lessor, as long as Lessor commences the curing of such default within said sixty (60) day period and thereafter proceeds with the curing of such default to completion with reasonable diligence, with allowance for delays due to the action or failure to act of governmental authorities, strikes, acts of God or other matters beyond the reasonable control of Lessor.

2.15.3 Remedies

Subject to the mortgagee protection provisions set forth in this Agreement, if default is made by Lessor as described in Section 2.15.1 hereinabove, and is not cured as provided in Section 2.15.2 hereinabove, Company shall provide Lessor with an additional written notice thereof and if such failure to cure continues for an additional period of sixty (60) days, Company shall be entitled to pursue any and all rights and remedies which it may have under this Agreement or at law or in equity, including, without limitation, the right to terminate this Agreement. All of such rights and remedies shall be cumulative and not alternative. Redelivery and disposal of improvements will be as described in Section 2.17, entitled SURRENDER AT END OF TERM.

2.16 WAIVERS AND ACCEPTANCE OF FEES

2.16.1 No waiver of default by either party hereto of any of the terms, covenants or conditions hereof to be performed, kept or observed will be construed to be or act as a waiver of any subsequent default of any of the terms, covenants, conditions herein contained to be performed, kept and observed.

- 2.16.2 No acceptance of fees or other money payments in whole or in part for any period or periods during or after default of any of the terms, conditions or covenants to be performed, kept or observed by the Company will be deemed a waiver on the part of the Lessor of any right or remedy which it may have under this Agreement or at law or in equity.
- 2.16.3 No acceptance of fees or other money payments in whole or in part for any period or periods during or after default of any of the terms, conditions or covenants to be performed, kept or observed by the Lessor will be deemed a waiver on the part of the Company of any right or remedy which it may have under this Agreement or at law or in equity.

2.17 SURRENDER AT END OF TERM

- 2.17.1 Subject to the mortgagee protection provisions set forth in Section 2.18 and subject to the rights of Sublessee's, Company covenants that at the termination of this Agreement, howsoever caused, it will quit and surrender the Company's right, title and interest in the Premises in good repair and condition, excepting reasonable wear and tear, acts of God, the public enemy or the action of the elements.
- 2.17.2 All Facilities located on the Premises at the expiration or sooner termination of the term of this Agreement, howsoever caused, shall, without payment or compensation of any kind to Company, then become Lessor's property, in fee simple, free and clear of all claims, liens and encumbrances to or against them by Company, any Leasehold Mortgagee, any Sublessee or Tenant, or any other third party claiming by or through Company. Subject to the mortgagee protection provisions set forth in Section 2.18 and subject to the rights of Sublessee's, upon termination of this Agreement howsoever caused, Company shall have the right to remove from the Premises, within sixty (60) days following the Termination Date, all equipment, trade fixtures and personal property belonging to Company or Sublessee's, provided Company or its Sublessees repair any damage caused by such removal.

For purposes of this Subsection 2.17.2, the words "equipment, trade fixtures and personal property" will include, but not be limited to, (i) signs (electrical or otherwise), (ii) all equipment used in connection with the conduct of Company's or its Sublessees business whether or not such equipment is attached to the Premises or the Facilities, (iii) any other mechanical device, and (iv) all other miscellaneous equipment, furnishings and fixtures installed on or placed on or about the Premises and used in connection with Company's or its Sublessees business thereon; provided, however, that such words shall not include elevators, escalators, plumbing systems, electrical systems, life safety systems, boilers, heating, ventilating and air conditioning systems, floor and wall coverings, ceiling lights, built-in shelving and cabinets, doors, windows, outside walls and fencing, and landscaping. All equipment, trade fixtures and personal property that is not removed within sixty (60) days following the Termination Date, shall become the property of Lessor.

2.17.3 Without limiting the provisions of this Section 2.17, Company agrees that upon the expiration or sooner termination of this Agreement, howsoever caused, it will execute, acknowledge and deliver to Lessor within thirty (30) days after demand from Lessor, all such documents and instruments as shall be reasonably necessary to evidence and

confirm the transfer of ownership of the Facilities from Company to Lessor, "as-is" in their existing condition without any further consideration required from Lessor.

2.18 FINANCING

- 2.18.1 Notwithstanding anything to the contrary contained in this Agreement, Company will have the right at such time or times as Company desires and without Lessor's consent, to hypothecate Company's interest in all or part of this Agreement, the Premises and the Facilities with one or more Leasehold Mortgages. Such Leasehold Mortgages may contain such terms and conditions as are acceptable to Company and Company will have the right, without the consent of Lessor, at any time during the term hereof to execute and deliver to any or all of its Leasehold Mortgagees any documents which will operate as collateral security for any loan or loans made, even if such document or documents result in a form or type of conveyance or assignment of the leasehold interest demised hereunder. It is hereby agreed that Company or any such Leasehold Mortgagees will have the right to immediately record such document or document(s) with an appropriate public official or officials. Company agrees that copies of all such documents of conveyance or assignment as contained in this Section 2.18 will be provided to the LDR forthwith. Lessor agrees to cooperate in executing any documents reasonably requested of Lessor by Company or the Leasehold Mortgagee in connection with any such Leasehold Mortgage; PROVIDED, HOWEVER, UNDER NO CIRCUMSTANCES SHALL LESSOR BE OBLIGATED TO SUBORDINATE ITS FEE INTEREST IN THE PREMISES TO ANY LEASEHOLD MORTGAGE, AND, NOTWITHSTANDING ANY TERM OR PROVISION OF ANY SUCH LEASEHOLD MORTGAGE OR THIS AGREEMENT TO THE CONTRARY, UNDER NO CIRCUMSTANCES SHALL ANY SUCH LEASEHOLD MORTGAGE CONSTITUTE AN INDEBTEDNESS OR OBLIGATION OF LESSOR NOR SHALL LESSOR BE LIABLE IN ANY WAY FOR THE PAYMENT OF ANY PORTION OF THE INDEBTEDNESS EVIDENCED BY SUCH LEASEHOLD MORTGAGE OR FOR THE PAYMENT OR PERFORMANCE OF ANY OTHER OBLIGATION THEREUNDER OR SECURED THEREBY.
- 2.18.2 Lessor will deliver to any such Leasehold Mortgagee written notice of any default of Company under the terms of this Agreement and said notice will specify the nature of the default. Before terminating this Agreement, Lessor will allow such Leasehold Mortgagee to cure or commence to cure any default of Company in accordance with the provisions of this Agreement. The time period to cure any default of Company will commence when said notice is delivered to Leasehold Mortgagee and Leasehold Mortgagee shall have the same lengths of time to cure the specified default as are permitted Company in Section 2.14. In the event Company fails to timely cure a default after receipt of written notice and expiration of any applicable cure period, Lessor agrees to provide any Leasehold Mortgagee with a second written notice and provide such Leasehold Mortgagee with an additional thirty (30) day cure period. Lessor will not have the right to exercise any remedies under this Agreement so long as a Leasehold Mortgagee is diligently prosecuting to complete a cure of any default. If such default is of a nature which is incapable of being cured by Leasehold Mortgagee, Lessor agrees not to exercise its remedies arising from such default if (i) Leasehold Mortgagee notifies Lessor in writing within such thirty (30) day cure period that Leasehold Mortgagee intends to foreclose its mortgage and Leasehold Mortgagee commences and diligently pursues such foreclosure; and (ii) Leasehold Mortgagee makes all payments due by Company under this Agreement through the date of foreclosure.

- 2.18.3 Any default by the Company in the payment of money as required under the terms of this Agreement may be cured by the Leasehold Mortgagee in accordance with the terms of Sections 2.14 and 2.18.2 of this Agreement, and Lessor will accept any such payment or cure from such Leasehold Mortgagee during the term of the Leasehold Mortgage.
 - 2.18.3.1 Should the Company default under the terms of this Agreement and should the default be such that it cannot be cured by the payment of money, Lessor will accept payments of rent from such Leasehold Mortgagee and this Agreement will not terminate, but will remain in full force and effect, pending Leasehold Mortgagee's cure of such default within the time periods described herein or resort to foreclosure or sale proceedings under its deed of trust or other security instruments.
- 2.18.4 If any default has been cured by a Leasehold Mortgagee or Assignee, Lessor agrees that upon completion of any foreclosure proceedings or sale under the deed of trust or other security instrument securing the loan, or upon delivery of a deed in lieu of foreclosure, the Leasehold Mortgagee or purchaser at such sale or any heir, successor, or assign of Leasehold Mortgagee (Assignee) subsequent to such sale will be recognized by Lessor as the lessee under the terms of this Agreement for all purposes for the remaining term hereof. The leasehold interest of the Leasehold Mortgagee or such Assignee will not be adversely affected or terminated by reason of any nonmonetary default occurring prior to the completion of such proceedings or sale, provided such default has been promptly remedied (to the extent it is capable of being remedied by Leasehold Mortgagee), or if such default requires possession to cure, provided such Leasehold Mortgagee, Assignee or purchaser promptly commences to cure upon taking possession of the Premises.
- 2.18.5 Such Leasehold Mortgagee will not become personally liable under the terms and obligations of this Agreement unless and until it assumes the obligations and is recognized by Lessor as lessee under this Agreement and will be liable only so long as such Leasehold Mortgagee or Assignee maintains ownership of the leasehold interest or estate and recourse to such Leasehold Mortgagee or Assignee shall be limited solely to Leasehold Mortgagee's or Assignee's interest in the Premises.
- 2.18.6 At any time during the term of this Lease, within ten (10) days after a written request by the Company or any Leasehold Mortgagee, Lessor, through Lessor's Designated Representative, will execute, acknowledge and deliver to the Company or such person or entity as the Company designates, a certificate stating:
 - that this Agreement is the only Agreement between Lessor and Company concerning the Premises and is unmodified and in full force and effect in accordance with its terms (or if there have been modifications, that this Agreement is in force and effect as modified, and identifying the modification agreements, or if this Agreement is not in full force and effect, that it is not);
 - (b) the commencement and expiration dates of this Agreement and the date to which rental has been paid to Lessor under this Agreement;
 - (c) whether or not, to the knowledge of the Lessor, there is an existing default by Company in the payment of Rent or any other sum of money under this Agreement, and whether or not there is any other existing default by either party

- under this Agreement with respect to which a notice of default has been served, and if there is such a default specifying its nature and extent;
- (d) whether or not, to the knowledge of Lessor, there are any set-offs, defenses or counter-claims against enforcement of the obligations to be performed by Lessor or Company under this Agreement; and,
- (e) such other reasonable information relating to this Agreement that a Leasehold Mortgagee or assignee may request.
- 2.18.7 All notices of default required to be delivered to a Leasehold Mortgagee under this Agreement will be sent to any Leasehold Mortgagee by overnight courier service or certified mail, return receipt requested. No such notice shall be valid or effective as against the Leasehold Mortgagee until and unless actually received by the Leasehold Mortgagee as evidenced by the courier service's delivery records or the return receipt.
- 2.18.8 In the event this Agreement is terminated for any reason prior to the end of the term (it being the intent that the Agreement will remain in full force and effect if Leasehold Mortgagee performs), Lessor shall enter into a new Agreement with Leasehold Mortgagee holding a first priority interest covering the Premises, provided that the Leasehold Mortgagee (i) requests such new Agreement by written notice to Lessor within sixty (60) days after termination, and (ii) cures all prior defaults of Company that are capable of being cured by Leasehold Mortgagee. The new Agreement shall be for the remainder of the term, effective at the date of such termination, at the same rent and on the same covenants, agreements, conditions, provisions, restrictions and limitations contained in this Agreement. The new Agreement shall have the same title priority as this Agreement and shall be subject only to the exceptions to title having priority over this Agreement or such additional exceptions to which such Leasehold Mortgagee has consented in writing. In the event the Lessor and such Leasehold Mortgagee enter into any such new Agreement, title to all the Facilities located upon the Premises as of the date of such new Agreement shall automatically vest in such Leasehold Mortgagee. If requested by such Leasehold Mortgagee, Lessor agrees to execute and deliver to Leasehold Mortgagee within 10 days after request therefore, a quitclaim deed in recordable form conveying title to such Facilities to such Leasehold Mortgagee. Lessor also agrees to assign to such Leasehold Mortgagee all Subleases with Tenants whose Subleases attorn to Lessor upon the termination of this Agreement.
- 2.18.9 Lessor shall cooperate in amending this Agreement from time to time to add any provision which may reasonably be requested by any Leasehold Mortgagee or proposed lender for the purpose of implementing the mortgagee protection provisions contained in this Agreement and allowing such Leasehold Mortgagee reasonable means to protect or preserve the lien of its loan and the value of its security. Lessor agrees to execute and deliver (and to acknowledge if necessary for recording purposes) any agreement necessary to effect any such amendment; provided, however, that any such amendment shall not in any way affect or change the term of this Agreement or the Rent or other amounts payable to Lessor under this Agreement, subordinate the fee interest of Lessor in the real property underlying the Premises, nor otherwise in any material respect adversely affect any rights of Lessor under this Agreement.
- 2.18.10 The bankruptcy or insolvency of Company will not operate or permit the Lessor to terminate this Agreement as long as all Rent or other monetary payments required to be

- paid by Company continue to be paid and other required obligations are performed in accordance with the terms of this Agreement.
- 2.18.11 To the extent any of the terms of this Agreement are inconsistent with the terms of this Section regarding mortgagee protection provisions, the mortgagee protection provisions will control.
- 2.18.12 Every Leasehold Mortgage shall contain a provision that copies of all notices of default by Company thereunder must be sent to Lessor. In the event of any default by Company under any Leasehold Mortgage, the Lessor reserves the right to make any payments due to the Leasehold Mortgagee before the Leasehold Mortgagee resorts to any foreclosure or sale proceedings under its deed of trust or other security instrument.
 - 2.18.12.1 Following any foreclosure or deed in lieu of foreclosure the leasehold estate created by this Agreement and the Facilities may be transferred or assigned as provided in Section 2.1.2.
- 2.18.13 Any mortgage placed by the Lessor on the fee title to the Premises shall be subordinate to this Agreement (and any replacement to this Agreement), any separate lease entered into in accordance with Section 4.8 and all Subleases.
- 2.18.14 Prior to any termination of this Agreement by Lessor, any Leasehold Mortgagee shall be allowed sufficient time to complete any foreclosure action, including time for delays due to official restraint (including by law, process or injunction issued by a court), so long as such Leasehold Mortgagee is making payments required by this Agreement which can be reasonably determined prior to acquiring the Company's interest under this Agreement. Leasehold Mortgagee shall have the right to terminate foreclosure proceedings at any time if Company has cured all defaults under any loan from Leasehold Mortgagee. However, no such termination of foreclosure proceedings shall prevent Lessor from pursuing, or continuing to pursue, its rights and remedies under this Agreement against Company for any uncured defaults under this Agreement.
- 2.18.15 So long as any Leasehold Mortgage is in effect, there shall be no merger of the leasehold estate created by this Agreement into the fee simple estate in the Premises without the prior written consent of the Leasehold Mortgagee.
- 2.18.16 Any Leasehold Mortgagee shall have the right to participate in any settlement or adjustment of losses under insurance policies maintained by Company under this Agreement. Such Leasehold Mortgagee shall be named as a loss payee or additional insured, as applicable, in accordance with any loan documents executed by Company, under the insurance policies required under this Agreement. Company shall use commercially reasonable efforts to obtain provisions in Leasehold Mortgages providing that any proceeds of insurance shall first be used for the purposes provided for in this Agreement before any portion thereof is applied to repay any indebtedness under such Leasehold Mortgage.

2.19 EMINENT DOMAIN

2.19.1 In the event of any acquisition of or damage to all or any part of the Premises or any interest therein by eminent domain, whether by condemnation proceeding or transfer in avoidance of an exercise of the power of eminent domain or otherwise during the term

of this Agreement, the rights and obligations of the parties with respect to such appropriation shall be as provided in this Section. As used herein:

- a. "Taking" shall mean any taking or damage, including severance damage, by eminent domain or by inverse condemnation or for any public or quasi-public use under any statute. The transfer of title may be either a transfer resulting from the recording of a final order in condemnation or a voluntary transfer or conveyance in lieu of an exercise of eminent domain or while condemnation proceedings are pending. The taking shall be deemed to take place on the earlier of (i) the date actual physical possession is taken by the condemnor or (ii) the date on which the title passes to the condemnor.
- b. "Total Taking" shall mean the taking of fee title to all of the Premises or so much of the Premises that the portions of the Premises not so taken are, in the reasonable judgment of Company, not reasonably suited for the uses of the Premises that have been previously made by Company. A Temporary Taking, defined below, may be treated as a Total Taking if it meets the test set forth above.
- c. "Partial Taking" shall mean any taking of fee title that is not a Total Taking.
- d. "Temporary Taking" shall mean the taking for temporary use of all or any portion of the Premises for a period ending on or before the expiration of the term.
- e. "Award" shall mean all compensation paid for the taking whether pursuant to a judgment or by agreement or otherwise.
- f. "Notice of Intended Taking" shall mean any notice or notification on which a reasonably prudent man would rely and which he would interpret as expressing an existing intention of taking as distinguished from a mere preliminary inquiry or proposal and shall include without limitation the service of a condemnation summons and complaint on either Lessor or Company or the receipt by either Lessor or Company from a condemning agency or entity of a written notice of intent to take containing a description or map of the taking reasonably defining the extent thereof.
- 2.19.2 Upon receipt of any of the following by either Party hereto, such Party shall promptly deliver a copy thereof, endorsed with the date received, to the other and any Leasehold Mortgagee which has given its name and address to such Party for such purpose.
 - a. Notice of Intended Taking;
 - b. Service of any legal process relating to the condemnation of all or any part of the Premises;
 - c. Notice in connection with any proceedings or negotiations with respect to any such condemnation; and
 - d. Notice of intent or willingness to make or negotiate a private purchase, sale or other transfer in lieu of condemnation.

- 2.19.3 Lessor, Company and any Leasehold Mortgagee that would be affected by such taking, shall each have the right to represent its respective interest in each proceeding or negotiation with respect to any taking or intended taking and to make full proof of its claims. No agreements, settlement, sale or transfer to or with the condemning authority shall be made without the prior written consent of Lessor, Company and Leasehold Mortgagee, which consent shall not be unreasonably withheld. Each of the parties hereto agrees to execute and deliver to the other any instruments that may be required to effectuate or facilitate any of the provisions of this Section where such execution or delivery will not adversely affect the right of such party to receive just compensation for any loss sustained in such negotiation or proceeding.
- 2.19.4 In the event of a Total Taking, this Agreement and all interests, rights, liabilities and obligations of the Parties hereunder shall terminate as of the date of Taking, except for liabilities and obligations arising out of events occurring prior to the date of termination.

In the event of a Taking of fee title to less than all of the Premises which Company reasonably believes to be a Total Taking, Company may, by written notice to Lessor approved by all Leasehold Mortgagees, within one hundred twenty (120) days after Company receives Notice of Intended Taking thereof, elect to treat such taking as a Total Taking. If Company fails to make such timely election, such Taking shall be deemed to be a Partial Taking for all intents and purposes.

Any taking determined to be a Total Taking in accordance with the foregoing paragraph shall be treated as a Total Taking if (i) Company delivers possession of the Premises to Lessor within one hundred twenty (120) days after Company shall have delivered written notice to Lessor electing to treat such taking as a Total Taking and (ii) Company has complied with all of the provisions hereof relating to the apportionment of the awards.

In a Total Taking, all sums, including damages and interest, awarded for the fee or the leasehold or both shall be distributed in the following order of priority:

- a. First, to Leasehold Mortgagees to the extent of the then balance due on all Leasehold Mortgages;
- b. Second, to Lessor an amount equal to the then present value of the reversionary interest of Lessor in the real property underlying the Premises at the Expiration of this Agreement;
- c. Third, subject to Section 1.1.25, the balance of the Award to Company.

All sums awarded for the leasehold or the fee shall (i) be delivered to Lessor and Company (or to Leasehold Mortgagee), respectively, if such award has been apportioned between Lessor and Company by such condemning authority, or (ii) deposited promptly with Leasehold Mortgagee (or in the event that there is no lender of record, with an escrow agent selected by Company in the reasonable exercise of its discretion), if only a single award is made, and distributed and disbursed as set forth above. Sums being held by the escrow agent pending disbursement shall be deposited in a federally insured interest bearing account and, upon distribution, each party having a right to any of the sums being disbursed shall be entitled to receive the interest attributable to its share of said sums.

2.19.5 In the event of a Partial Taking, this Agreement shall remain in full force and effect as to the portion of the Premises remaining.

On a Partial Taking, all sums, including damages and interest, awarded for the fee or the leasehold or both shall be distributed in the following order of priority:

- a. First, to Company an amount equal to the cost of making all repairs and restorations to any Facilities on the Premises affected by such taking to the extent necessary to restore the same to a complete architectural and economically viable or functioning units (to the extent permitted, however, taking into consideration the amount of land remaining after such taking or purchase);
- b. Second, to any Leasehold Mortgagee to the extend that the security if its Leasehold Mortgage has been impaired as a result of the Partial Taking or as required by its loan documents, whichever is greater;
- c. Third, to Lessor an amount equal to the then present value of the reversionary interest of Lessor in the real property underlying the Premises at the Expiration of this Agreement in that portion of the real property underlying the Premises that is taken in such Partial taking;
- d. Fourth, subject to Section 1.1.25, the balance of the Award to Company.

All sums awarded for the leasehold or the fee shall (i) be delivered to Lessor and Company (or to Leasehold Mortgagee), respectively, if such award has been apportioned between Lessor and Company by such condemning authority, or (ii) deposited promptly with Leasehold Mortgagee (or in the event that there is no lender of record, with an escrow agent selected by Company in the reasonable exercise of its discretion), if only a single award is made, and distributed and disbursed as set forth above. Sums being held by the escrow agent pending disbursement shall be deposited in a federally insured interest bearing account and, upon disbursement, each party having a right to any of the sums being disbursed shall be entitled to receive the interest attributable to its share of said sums.

- 2.19.6 In the event of a Temporary Taking, this Agreement shall remain in full force and effect, neither the rents reserved hereunder nor the term hereof shall be reduced or affected in any way, and, subject to Section 1.1.25, Company shall be entitled to any award for the use or estate taken, subject to the requirements of any Leasehold Mortgagee.
- 2.19.7 Unless the respective values are determined by the court in the eminent domain proceeding, the values of the interests for which Lessor and Company are entitled to compensation in the event of a Total or Partial Taking shall be determined by the mutual written agreement of Lessor and Company. If Lessor and Company are unable to agree on the value of said interests within thirty (30) days after the deposit of the sums awarded with the escrow agent, then within thirty (30) days after the expiration of that period, each such party shall submit its good faith estimate of the value of said interests as of the date of the taking. If the higher of said estimates is not more than 105% of the lower of such estimates, the value shall be the average of the submitted estimates. If otherwise, then within ten (10) days the question shall be submitted to arbitration pursuant to the rules of the American Arbitration Association.

ARTICLE III

3.1 ENVIRONMENTAL POLICY

3.1.1 Violation Of Environmental Laws

During the term of this Agreement, Company will not cause or permit any Hazardous Material to be used, generated, manufactured, produced, stored, brought upon, or released on, under or about the Premises, or transported to and from the Premises, by Company, its Sublessees and Tenants, or their respective agents, employees, contractors, invitees or a third party in violation of the Environmental Laws.

- 3.1.1.1 LDR will have access to the Premises to inspect same to insure that Company is using the Premises in accordance with Environmental Laws.
- Company, at the LDR's reasonable request, as a Project Cost, will conduct such testing and analysis as necessary to ascertain whether Company is using the Premises in compliance with environmental requirements. Any such tests will be conducted by qualified independent experts chosen by Company and subject to LDR's reasonable written approval. Copies of such reports from any such testing will be provided to LDR.
- 3.1.1.3 Company will promptly provide copies of all notices, reports, claims, demands or actions concerning any environmental concern or release or threatened release of Hazardous Materials to Lessor.

3.1.2 Contamination Of Premises

- 3.1.2.1 If the illegal use of any Hazardous Material on, under or about the Premises Company, its agents, employees or contractors results in any contamination of the Premises, in violation of an Environmental Law, Company will promptly take all actions as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Company will take all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual harm and to bring the Premises into compliance with all environmental requirements. If any such contamination is a result of the gross negligence of Company or its employees or if Company or its employees intentionally contaminate the Premises, all costs incurred to bring the Premises into compliance with environmental laws shall be at Company's cost and expense and not as a Project Cost. If any such contamination is a result of the actions of agents, contractors, third parties or the actions of Company that were not intended to contaminate the Premises, all costs incurred to bring the Premises into compliance with environmental laws shall be a Project Cost.
- 3.1.2.2 Lessor will be solely responsible for any environmental condition existing on or about the Premises prior to the commencement of the term of this Agreement or any environmental conditions caused by Lessor, its agents, employees, contractors or invitees during the term. Lessor will promptly take all actions, at

its cost and expense, as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Lessor will take all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual harm and to bring the Premises into compliance with all environmental requirements.

3.1.2.3 If the illegal use of any Hazardous Material on, under or about the Premises Sublessees and Tenants, their respective agents, employees, contractors, servants or invitees results in any contamination of the Premises, in violation of an Environmental Law, Company will promptly take all actions, as a Project Cost, as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Company will take all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual with all environmental harm and to bring the Premises into compliance requirements. All amounts recovered (whether by way of settlement of a claim, judgment, indemnity or otherwise) by Company from any such Sublessee, Tenant, agent, employee, contractor, servant or invitee in connection with the foregoing matters shall be included in Total Revenue for the full or partial calendar year of the term of this Agreement in which recovered (Company's reasonable costs of recovery shall be deducted as a Project Cost), or if recovered following the expiration or termination of the term of this Agreement, fifty percent (50%) of such recovery (after deducting from such recovery the Company's reasonable costs of recovery not theretofore included in Project Costs) shall be promptly paid over to Lessor.

3.1.3 Compliance With All Governmental Authorities

During the term of this Agreement, Company will promptly make all submissions required of Company and comply with all requirements of the appropriate governmental authority under all Environmental Laws. During the term of this Agreement, Lessor will promptly make all submissions required of Lessor to, and comply with, all requirements of the appropriate governmental authority under all Environmental Laws.

3.1.3.1 Should any governmental agency having authority over such matters determine that a site characterization, site assessment, and/or cleanup plan be prepared or that a cleanup should be undertaken because of any spills or discharges of hazardous materials at the Premises by Company, its agents, employees or contractors which occur during the term of this Agreement, then Company shall prepare and submit the required plans and financial assurances, and carry out the approved plans. If any such spills or discharges are a result of the gross negligence of Company or its employees or if Company or its employees intentionally spilled or discharged hazardous materials at the Premises, all costs incurred to prepare and submit the required plans and carry out the approved plans shall be at Company's cost and expense and not as a Project Cost. If any such spills or discharges are a result of the actions of agents, contractors, third parties or the actions of Company that were not intended to spill or discharge hazardous materials at the Premises, all costs incurred to prepare and submit the required plans and carry out the approved plans shall be a Project Cost.

- 3.1.3.2 Company hereby agrees to indemnify Lessor from all costs incurred in connection with any investigation of site conditions or any cleanup, remediation, removal or restoration work required by any Federal, State or local governmental agency or political subdivision because of an environmental condition on, under or about the Premises that is caused by Company, its Sublessees, their agents, employees, contractors or invitees. If any such environmental condition is a result of the gross negligence of Company or its employees or if Company or its employees intentionally caused such environmental condition, all costs incurred to bring the Premises into compliance with environmental laws shall be at Company's cost and expense and not as a Project Cost. If any such environmental condition is a result of the actions of Sublessees, agents, contractors, third parties or the actions of Company that were not intended to contaminate the Premises, all costs incurred to bring the Premises into compliance with environmental laws shall be a Project Cost. Lessor hereby agrees to indemnify Company from all costs incurred in connection with any investigation of site conditions or any cleanup, remediation, removal or restoration work required by any Federal, State or local governmental agency or political subdivision because of an environmental condition on, under or about the Premises that existed prior to the Commencement Date of this Agreement or was caused by Lessor, its agents, employees, contractors or invitees during the term.
- 3.1.3.3 Should any governmental agency having authority over such matters determine that a site characterization, site assessment, and/or cleanup plan be prepared or that a cleanup should be undertaken because of any spills or discharges of hazardous materials at the Premises by Sublessees and Tenants, and their respective agents, employees, contractors or invitees which occur during the term of this Agreement, then Company shall (as a Project Cost) prepare and submit the required plans and financial assurances, and carry out the approved plans. Company will promptly provide all information requested by Lessor to determine the applicability of the Environmental Laws to the Premises, or to respond to any governmental investigation or to respond to any claim of liability by third parties which is related to such environmental contamination. All amounts recovered (whether by way of settlement of a claim, judgment, indemnity or otherwise) by Company from any such Sublessee, Tenant, agent, employee, contractor, servant or invitee in connection with the foregoing matters shall be included in Total Revenue for the full or partial calendar year of the term of this Agreement in which recovered (Company's reasonable costs of recovery shall be deducted as a Project Cost), or if recovered following the expiration or termination of the term of this Agreement, fifty percent (50%) of such recovery (after deducting from such recovery the Company's reasonable costs of recovery not theretofore included in Project Costs) shall be promptly paid over to Lessor.

ARTICLE IV

4.1 FORCE MAJEURE

Neither Lessor, Company or any Leasehold Mortgagee will be deemed to be in breach of this Agreement by reason of failure to perform any of its obligations hereunder if, while and to the extent that such failure is due to strikes, boycotts, labor disputes, embargoes, shortages of materials, acts of God, acts of the public enemy, acts of governmental authority, unusual weather

conditions, floods, riots, rebellion or sabotage. However, the provisions of this Section will not apply to, nor excuse any failure by Company to pay Rents, fees or any other money payments required under the provisions, covenants or agreements contained in this Agreement.

4.2 QUIET ENJOYMENT

Lessor agrees that, on payment of the Rents and fees and performance of the covenants, conditions and agreements on the part of Company to be performed hereunder, Company will have the right to peaceably occupy and enjoy the Premises.

4.3 NOTICES

All notices, requests, consents and approvals under this Agreement will be given only by personal delivery, certified mail return receipt requested, overnight courier service or facsimile. The same shall be deemed to have been given: (i) if personally delivered, upon receipt, (ii) if by certified mail, upon the date indicated on the return receipt, (iii) if by overnight courier service, upon the date delivered as shown by the records of the courier, and (iv) if by facsimile, at the time of electronic confirmation of successful transmission.

Notices intended for the Lessor will be addressed to:

Successor Agency to the Industry Urban-Development Agency 15625 East Stafford Street, Suite 200 City of Industry, California 91744 Attn: Executive Director

Facsimile: (626) 961-6795

or to such other address as may be designated by the Lessor by written notice to Company.

Notices intended for the Company will be addressed to:

Industry East Land, LLC c/o Majestic Realty Co. 13191 Crossroads Parkway North City of Industry, California 91746 Attn: Edward P. Roski, Jr. Facsimile: (562) 692-1553

or to such other address as may be designated by the Company by written notice to Lessor.

4.4 HEADINGS, TITLES OR CAPTIONS

Article, section or paragraph headings, titles or captions are inserted only as a matter of convenience and for reference, and in no way define, limit or describe the scope or extent of any provision of this Agreement.

4.5 INVALID PROVISIONS

It is expressly understood and agreed by and between the parties hereto that in the event any covenant, condition or provision herein contained is held to be invalid by any court of competent jurisdiction, the invalidity of such covenant, condition or provision will in no way affect any

other covenant, condition or provision herein contained; provided, however, that the invalidity of any such covenant, condition or provision does not materially prejudice either Lessor or Company in their respective rights and obligations contained in the valid covenants, conditions or provisions of this Agreement.

4.6 STATE OF CALIFORNIA LAW AND VENUE

This Agreement will be interpreted under and governed by the internal laws of the State of California, without regard to principles of conflicts of law. Venue shall be County of Los Angeles.

4.7 ENTIRE AGREEMENT

This document represents the entire agreement between the parties hereto and it cannot be effectively modified or canceled by mutual agreement or in any manner, nor may any term or provisions of this Agreement be waived, except by an instrument in writing approved by Lessor and executed by duly authorized officials, officers or agents of the Parties.

4.8 SUBDIVISION OF THE PREMISES AND RIGHT TO SEPARATE LEASES

- The parties hereto acknowledge and agree that the Premises will be developed by Company with 4.8.1 Facilities, in phases, over a number of years. During the term of this Agreement, Company may, from time to time, as a Project Cost, to subdivide the Premises to create such separate legal parcels as Company reasonably deems desirable for the development of the Premises and to cause subdivision maps covering the Premises, or any portion thereof, to be prepared and processed through the appropriate governmental agencies, executed by Company and Lessor, and filed for record in accordance with all laws and ordinances applicable thereto. Subject to the foregoing, at any time, and from time to time during the term of this Agreement, Lessor agrees to, within 15 days after the written request of Company, execute and deliver such instruments as may be appropriate, necessary, or required for: (i) the grant or dedication of any easement, right of way or other property right to any public entity or service corporation for ingress, egress, road or utility purposes and reasonably required for the subdivision or development of the Premises, or (ii) the obtaining of governmental approvals, consents, zoning changes, conditional uses, variances, subdivision maps or the like, or for the purpose of providing adequate utility services to the Premises, or permitting Company to construct the Facilities or other improvements on the Premises or make any alteration or addition to the Facilities.
- 4.8.2 The parties further acknowledge and agree that as a result of such phased development, Company will be obtaining construction, interim and permanent financing for the development of various portions of the Premises and the Facilities at different times during the Term of this Agreement. In order to facilitate such phased development and to facilitate separate financing for various phases thereof, the parties hereby agree that the Company shall have the right, at any time, and from time to time, during the term of this Agreement, or any extension thereof, to have any portion of the Premises removed from the legal description of the Premises and to have the Lessor and Company enter into a new separate lease covering the portion of the Premises so removed from this Agreement.
 - 4.8.2.1 All such new leases shall be for the remainder of the Term of this Agreement and shall be on the same terms and conditions as this Agreement, including, without limitation, the right of first refusal as set forth herein. If requested by Company, Lessor agrees to grant any easements reasonably required for access, road or utility purposes between the

- Premises remaining subject to this Agreement and the portion of the Premises which becomes subject to any such new lease.
- 4.8.2.2 At any time Company desires a separate lease, Company shall prepare an amendment to this Agreement deleting the portion of the Premises to be covered by the separate lease and shall also prepare such separate lease and submit both documents to Lessor for execution. Lessor agrees to execute such amendment and separate lease within twenty (20) days after Company submits such documents. Concurrently with the execution of each such amendment and separate lease, the Parties shall execute a memorandum in recordable form evidencing the existence of such separate lease, the portion of the Premises subject to such separate lease, the effective date and termination date of such separate lease and the fact that there is a right of first refusal to purchase the Premises. Concurrently with the execution of such amendment and separate lease, the parties shall also execute a memorandum in recordable form evidencing the existence of such amendment to this Agreement. Any such separate lease shall have the same title priority as this Agreement and shall be subject only to the exceptions to title having priority over this Agreement or such additional exceptions to which Company has consented in writing. In the event the parties enter into any such separate lease, title to all Facilities located upon the Premises subject to such separate lease shall automatically vest in Company as of the effective date of such separate lease. If requested by Company, Lessor agrees to execute and deliver to Company within 10 days after request therefore, a quitclaim deed in recordable form conveying title to such Facilities on the premises subject to the separate lease to Company. Lessor also agrees to assign to Company all Subleases with Tenants whose Subleases attorn to Lessor upon the termination of this Agreement and creation of such separate lease.
- 4.8.2.3 The creation of a separate lease of a particular parcel that was included in the initial conveyance, pursuant to this Article 4.8 is intended to merely continue the initial grant. The amendment of this Lease to delete separately leased premises and the creation of any separate lease is not intended to: (1) expand the rights or interests created by the initial conveyance; (2) constitute a "creation" of a taxable possessory interest; (3) constitute a "renewal" or "extension" of a taxable possessory interest; and/or (4) grant any additional property or authorize uses not already permitted by this Lease.

4.9 RIGHT OF FIRST REFUSAL TO PURCHASE

At all times during the term of this Agreement and any extensions or renewals thereof, should Lessor receive a bona fide offer from any third party to purchase all or any portion of the Premises, which offer Lessor desires to accept, Lessor shall, before accepting such offer, provide Company with a full and complete copy of such offer and shall first offer in writing to sell the portion of the Premises which is the subject of such offer to Company on the same terms and conditions as set forth in said offer. Upon receipt of any such notice and copy of such offer from Lessor, Company shall have thirty (30) days thereafter within which to accept the same. Should Company fail to accept any such offer within said thirty (30) day period, Lessor shall be free to sell said portion of the Premises to the original offeror upon the same terms and conditions offered to Company without further notice to Company. Should Lessor after having made such offer to Company as above described, fail to sell said portion of the Premises upon the same terms and conditions offered to Company, Lessor shall give Company notice and the first right to purchase in the manner set forth above of any further or different offers received by Lessor for the purchase of said portion of the Premises and shall first offer to sell the same to Company upon the same terms and conditions before accepting any such further or different offer. The right of first refusal

set forth herein is a continuing right of first refusal to purchase and shall apply to all subsequent bona fide offers from third parties after a sale by Lessor, or its successors, to a party other than Company.

4.10 UTILITIES

To the extent not paid by Sublessees, Company shall, as a Project Cost, pay for all water, gas, heat, light, power, telephone service, trash disposal and other utilities and services supplied to the Premises and the Facilities located thereon, together with any taxes thereon.

4.11 OWNERSHIP OF IMPROVEMENTS

All improvements constructed on the Premises by Company (including, without limitation, the Facilities) at any time and from time to time, will be owned by Company during the term of this Agreement. If requested by Company, any Leasehold Mortgagee or title insurance company, Lessor agrees to execute and deliver to Company within ten 10 days after request therefor, a quitclaim deed in recordable form conveying or confirming title to all such improvements in Company. All such buildings and improvements constructed upon the Premises by Company are and shall remain real property and may not be severed from this Agreement or the leasehold estate created hereby.

4.12 LEASING COMMISSIONS

In the event Company retains the services of Majestic Realty Co. or any affiliate of Company to Sublease the Premises and the Facilities, the leasing commissions paid to Majestic Realty Co. or any such affiliate, as a Project Cost, shall not exceed the scheduled commissions for similar services charged by the major brokerage companies operating in the County of Los Angeles, California.

4.13 COVENANTS FOR NON-DISCRIMINATION

The Company covenants by and for itself and any successors in interest that there shall be no discrimination against or segregation of any person or group of persons on account of race, color, creed, religion, sex, marital status, age, handicap, national origin or ancestry in the use and occupancy of the Premises, nor shall the Company itself or any person claiming under or through it establish or permit any such practice or practices of discrimination or segregation with reference to the use or occupancy of the Premises.

- 4.13.1 The Company shall refrain from restricting the rental or lease of the Premises on the basis of race, color, creed, religion, sex, marital status, handicap, national origin or ancestry of any person. All such leases or contracts shall contain or be subject to substantially the following nondiscrimination or nonsegregation clauses:
 - 1. In leases: "The lessee herein covenants by and for himself or herself, his or her heirs, executors, administrators and assigns, and all persons claiming under or through him or her, and this lease is made and accepted upon and subject to the following conditions:

"There shall be no discrimination against or segregation of any person or group of persons on account of race, color, creed, religion, sex, marital status, handicap, ancestry or national origin in the leasing, subleasing, transferring, use,

occupancy, tenure or enjoyment of the premises herein leased nor shall the lessee himself or herself, or any person claiming under or through him or her, establish or permit any such practice or practices of discrimination or segregation with reference to the selection, location, number, use or occupancy of tenants, lessees, sublessees, subtenants or vendees in the premises herein leased."

- 2. In contracts: "There shall be no discrimination against or segregation of, any person, or group of persons on account of race, color, creed, religion, sex, marital status, handicap, ancestry or national origin, in the sale, lease, sublease, transfer, use, occupancy, tenure or enjoyment of the premises, nor shall the transferee himself or herself or any person claiming under or through him or her, establish or permit any such practice or practices of discrimination or segregation with reference to the selection, location, number, use or occupancy of tenants, lessees, subtenants or vendees of the premises."
- 4.13.2 The covenants established in this Section shall, without regard to technical classification and designation, be binding for the benefit and in favor of the Lessor, its successors and assigns, the City of Industry and any successor in interest to the fee ownership of the Premises or any part thereof. The covenants, contained in this Section shall remain in effect for the term of this Agreement.

4.14 PUBLIC FINANCING

The parties hereby agree and acknowledge that, in order to assist with the financing of construction of Streets A, B, C, D and Grand Crossing Parkway and the storm drain system, sanitary sewer, domestic, fire and reclaimed water systems, and dry utilities located within the public rights of way of Streets A, B, C, D and Grand Crossing Parkway, the parties may initiate proceedings to form one or more community facilities or other assessment or public financing districts which may encompass all or certain portions of the Premises. The parties hereby agree to cooperate in good faith with each other to (i) initiate such formation proceedings, (ii) form one or more community facilities or other assessment or financing districts over all or any portion of the Premises, and (iii) cause such public financing district or districts to issues bonds to assist with the financing of construction of Streets A, B, C, D and Grand Crossing Parkway and the storm drain system, sanitary sewer, domestic, fire and reclaimed water systems, and dry utilities located within the public rights of way of Streets A, B, C, D and Grand Crossing Parkway. Company and Lessor hereby agree that the Premises or any portion thereof, as determined by the parties, shall be included within the boundaries of said public financing district or districts. In addition, the parties hereto hereby agree to cooperate with each other and, if necessary, the City of Industry to promptly execute all other documents and take all other actions reasonably requested by Lessor and/or Company in connection with any and all actions and approvals required to be taken in the ordinary course of formation of such community facilities or other assessment or public financing district over all or any portion of the Premises or any actions related thereto in accordance with the terms of the Mello-Roos Community Facilities Act of 1982, as amended, Rule 15c2-12 promulgated by the Securities and Exchange Commission, or any other California law governing formation of such public financing districts. The parties hereby agree and acknowledge that, in order to assist with the annual maintenance of the landscaping and irrigation systems installed upon the slopes manufactured during the grading of the Project, the parties may initiate proceedings to form one or more maintenance districts which may encompass all or certain portions of the Premises. The parties hereby agree to cooperate in good faith with each other to (i) initiate such formation proceedings, and (ii) form one or more

maintenance districts over all or any portion of the Premises. Company and Lessor hereby agree that the Premises or any portion thereof, as determined by the parties, shall be included within the boundaries of said district or districts. In addition, the parties hereto hereby agree to cooperate with each other and, if necessary, the City of Industry to promptly execute all other documents and take all other actions reasonably requested by Lessor and/or Company in connection with any and all actions and approvals required to be taken in the ordinary course of formation of such district over all or any portion of the Premises or any actions related thereto in accordance with the terms of any applicable Federal or California law governing the formation and operation of such districts.

IN WITNESS WHEREOF, Lessor and Company have executed these presents the day and year first above written.

ATTEST:	SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY
BY:	BY:
	INDUSTRY EAST BUSINESS CENTER, LLC, a Delaware Limited Liability Company
	BY: Majestic Realty Co., a California corporation, its Manager
APPROVED AS TO FORM: Casso & Sparks, LLP	By:
By:	Ву:
	Its:

EXHIBIT "A" LEGAL DESCRIPTION

PARCEL 8 OF PARCEL MAP NO. 353, IN THE CITY OF INDUSTRY, COUNTY OF LOS ANGELES, STATE OF CALIFORNIA, FILED IN BOOK ____ PAGES ___ THROUGH ____, INCLUSIVE OF PARCEL MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY.

EXHIBIT "B"

SCOPE OF SUCCESSOR AGENCY IMPROVEMENTS

SCOPE OF AGENCY IMPROVEMENTS

The Agency will, at its cost and expense (unless a public financing district is formed to assist with the financing of a portion of such costs in accordance with Section 4.14 of the Lease Agreement), design and construct the public improvements, located within the public rights of way, reasonably necessary to adequately serve the Project and to provide the mitigation required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center. Said public improvements shall include:

A. Roads

Design and construct streets and all utilities and infrastructure for the following project roadways. (See attached Site Development Plan.)

Street improvements will include curb, gutter, and road pavement design and grades and sidewalks. Street lighting, traffic signals, pavement striping, traffic control signage, landscape and irrigation for roadway medians are also included.

Street A – from Grand Ave west to Grand Crossing Parkway and from Grand Avenue east to Street "B", including special intersection design at Grand Avenue, all street improvements, utilities and infrastructure.

Street B – From Street "A" to Grand Avenue and intersection modifications at Grand Avenue all street improvements, utilities and infrastructure.

Street C – From Street "B" to Street "D", all street improvements, utilities and infrastructure.

Street D – From Street "C" to the end of the cul de sac, all street improvements, utilities, and infrastructure.

Grand Crossing Parkway – From Grand Avenue west to the project boundary, including special intersection design at Grand Avenue, all street improvements, utilities and infrastructure.

Grand Avenue – Modify to accommodate project traffic and project related turn movements. All street improvements, utilities and infrastructure.

B. Master Hydrology

Prepare a master hydrology report and construct the required improvements for the project site based on full buildout of Project as depicted in the EIR The hydrology report will require approval from the Los Angeles County Food Control District for storm drain connections to San Jose Creek.

C. Storm Drain System

Design and construct a storm drainage system to convey water from the planned public streets described above. The storm drainage system will be designed to carry storm water flows from the planned development areas and adjacent property. Lateral stubs will be extended to the street right of way for each development lot. (See attached Storm Drain System.)

D. Sanitary Sewer:

Design and construct sanitary sewer main lines within the project roadways and easements where necessary, to serve each lot within the project development. Plans will include main line connections to existing trunk sewers. (See attached Sewer System.)

E. Water

Design and construct potable water lines within Grand Crossing Parkway, Street A, Street B, Street C and Street D. Lines will connect to the existing systems in Grand Avenue, and Baker Parkway. Service laterals to each lot will be provided, however meter vaults and back flow preventors will not be set. The location of such meter vaults and back flow preventers shall be established by the Company. (See attached Water System).

Install Fire Hydrants as required.

F. Reclaimed Water

Design and construct reclaimed water lines within Grand Avenue, Grand Crossing Parkway, Street A, Street B, Street C and Street D and any other reclaimed lines required for the Project. Design and construct water reservoir described in EIR.

Grand Crossing Parkway, Street A, Street B, Street C and Street D water lines will connect to the existing systems in Grand Avenue and Baker Parkway. Services laterals to each lot will be provided, however meter vaults and back flow preventers will not be set. The location of such meter vaults and back flow preventers will be established by the Company. (See attached Reclaimed Water System.)

G. Dry Utilities

Design and construct ducts, vaults, junction boxes and related substructure elements for Electrical, Telephone (including fiber optics), and main line Gas systems. Line and ducts will stubbed to the right of way line to provide service to each lot. Such work shall be coordinated with utility companies. (See attached Gas Lines.)

H. Grading

Design and construct mass grading at elevations predetermined by Company and Agency, to remove or reduce existing hill areas and to remove and stabilize landslide areas within the Project. Export material will be transported and placed as compacted fill within the street rights of way for the construction of public improvements, or for slopes and creation of the development planning areas described in the EIR. Any excess material will be exported from the Project. The existing utility lines crossing such hill areas will be relocated and any required wetlands mitigation will be undertaken.

I. Former Valley Land Development Company Landfill

Provide the mitigation required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center relating to the Valley Land Development Company landfill.

J. <u>Slope Landscaping</u>

Design and construct the landscaping and irrigation systems required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center relating to the re-vegetation and landscaping of the slopes manufactured during the grading described in Item H above.

K. Additional Work

Any additional public improvements required to satisfy the CEQA mitigation requirements for the Project.

SUCCESSOR AGENCY ITEM NO. 5.5

Staff Report & Resolution will be provided

Prior to the Meeting

LEASE AGREEMENT

THIS LEASE AGREEMENT (hereinafter referred to as "Agreement") entered into this 23rd day of May, 2019 by and between the SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY, a body corporate and politic (hereinafter referred to as "Lessor") and INDUSTRY EAST BUSINESS CENTER, LLC, a Delaware limited liability company authorized to do business in the State of California (hereinafter referred to as "Company"):

WITNESSETH:

WHEREAS, Lessor is the owner of certain real property located in the State of California described on Exhibit "A" attached hereto (hereinafter referred to as "Premises"); and

WHEREAS, by an unrecorded Lease Agreement dated April 28, 2005 (the "Lease"), Lessor's predecessor in interest, the Industry Urban-Development Agency leased the Premises and certain other real property located in the State of California (the "Larger Property") to Company's predecessor in interest, Industry East Land, LLC. Industry East Land, LLC assigned its interest in the Lease to Company by an unrecorded Assignment of Lease dated September 21, 2005. It was the intent of the parties that the Larger Property would be subdivided, financed and developed, in phases, over a number of years; and

WHEREAS, in order to facilitate the subdivision, financing and development of the Larger Property in phases, Article 4.8 of the April 28, 2005 Lease granted Company the right to have any portion of the Larger Property removed from the legal description of the April 28, 2005 Lease and made the subject of a separate lease between Lessor and Company for the remainder of the term of the April 28, 2005 Lease and on the same terms and conditions as the April 28, 2005 Lease; and

WHEREAS, Lessor and Company are removing the Premises described on Exhibit "A" from the legal description of the April 28, 2005 Lease and entering into this separate lease of the Premises pursuant to Article 4.8 of the April 28, 2005 Lease; and

WHEREAS the term of the Lease has not commenced pursuant to Article 1.2 of the Lease prior to this Agreement. The term of the leasehold interest under this Agreement shall commence upon the date specified in Section 1.2.1 herein.

WHEREAS, Company proposes to lease the Premises from Lessor and to develop thereon and operate and manage certain buildings and other improvements as permitted by this Agreement; and

WHEREAS, Company is engaged in the business of planning, constructing, maintaining, leasing financing and operating such projects; and

WHEREAS, on and subject to the terms and conditions contained in this Agreement, Lessor is willing to lease the Premises to Company for such purposes;

NOW, THEREFORE, for and in consideration of the agreements, covenants and conditions herein, Lessor and Company agree as follows:

ARTICLE I

1.1 **DEFINITIONS**

In addition to the terms defined elsewhere in this Agreement, the following terms shall have the following meanings:

- 1.1.1 The term "Approved Budget," whenever used herein, means any written budget prepared by Company and approved by the Lessor's Designated Representative pursuant to the procedure set forth in Section 1.6 entitled BUDGET APPROVAL.
- 1.1.2 The term "Assignee," whenever used herein, means the purchaser or any heir, successor, or assign of a lender subsequent to a sale or assignment as defined in Section 2.18 entitled FINANCING.
- 1.1.3 The term "Capital Improvement Expenditures," whenever used herein, means any expenses associated with the Facilities (other than the costs and expenses of their initial construction) which are treated as capital expenditures under generally accepted accounting principles, consistently applied.
- 1.1.4 The term "Lessor's Designated Representative (hereinafter referred to as 'LDR')," whenever used herein, means the Executive Director of Lessor, or a successor designee acting on behalf of the Lessor.
- 1.1.5 The term "Commence Construction," whenever used herein, means commencing construction of the Facilities on the Premises by Company causing its construction contractor to obtain occupancy and control of the area and to begin actual site development and construction thereon.
- 1.1.6 The term "Facilities," whenever used herein, means all the buildings, utilities, rail improvements, site development work, parking lots, roadways, landscaping and other improvements constructed by Company on the Premises in accordance with the terms and conditions of this Agreement.
- 1.1.7 The term "Company," whenever used herein, means Industry East Business Center, LLC, a limited liability company authorized to do business in the State of California, which is entering into this Agreement as the developer and operator of the Facilities on the Premises as described herein.
- 1.1.8 The term "Lessor," whenever used herein, means the Successor Agency to the Industry Urban-Development Agency, a body corporate and politic, and where this Agreement speaks of "Approval by Lessor," or "Approved by Lessor," such phrases mean approving action by the Successor Agency to the Industry Urban-Development Agency.
- 1.1.9 The term "CC&Rs," whenever used herein, means any Covenants, Conditions and Restrictions developed by the Company and approved by the LDR for Sublessees and Tenants which will include, but not be limited to, specific guidelines for uses of the Premises.
- 1.1.10 The term "Debt Service," whenever used herein, means the Company's payment of principal, interest and any other sum due and owing (monthly or otherwise) pursuant to the terms and conditions of any Leasehold Mortgage for construction, interim and/or permanent financing of the Facilities, but excluding any penalties, late charges, default interest and other amounts (except principal and interest) payable solely as a result of a

breach or default under such Leasehold Mortgage, unless such breach or default is caused in part by the acts or omissions of Lessor, including the failure of Lessor to make Capital Contributions as called for in this Agreement.

- 1.1.11 The term, "Commencement Date," whenever used herein, means the date that the term of this Lease commences under the terms of Section 1.2.1 below.
- 1.1.12 The term "Environmental Laws," whenever used herein, means any one or all of the laws and/or regulations of the Environmental Protection Agency or any other federal, state or local agencies regulating Hazardous Material, including, but not limited to the following as the same are amended from time to time:

COMPREHENSIVE ENVIRONMENTAL RESPONSE, COMPENSATION AND LIABILITY ACT (42 U.S.C. Section 9601 et seq.)

RESOURCE CONSERVATION AND RECOVERY ACT (42 U.S.C. Section 6901 et seq.)

TOXIC SUBSTANCES CONTROL ACT, as amended (15 U.S.C. Section 2601 et seq.)

SAFE DRINKING WATER ACT (42 U.S.C. Section 300h et seq.)

CLEAN WATER ACT (33 U.S.C. Section 1251 et seq.)

CLEAN AIR ACT (42 U.S.C. Section 7401 et seq.)

and the regulations promulgated thereunder and any other laws, regulations and ordinances (whether enacted by the Federal, State or local government) now in effect or hereinafter enacted that deal with the regulation or protection of the environment (including, but not limited to, the ambient air procedures and records detailing chlorofluorocarbons [CFC]), ambient air, ground water, surface water and land use, including sub-strata land.

1.1.13 The term "Hazardous Material," whenever used herein, means the definitions of hazardous substance, hazardous material, toxic substance, regulated substance or solid waste as defined within the following:

COMPREHENSIVE ENVIRONMENTAL RESPONSE, COMPENSATION AND LIABILITY ACT (42 U.S.C. Section 9601 et seq.)

RESOURCE CONSERVATION AND RECOVERY ACT (42 U.S.C. Section 6901 et seq.)

HAZARDOUS MATERIALS TRANSPORTATION ACT (49 U.S.C. Section 1801 et seq.) and all present or future regulations promulgated thereto.

DEPARTMENT OF TRANSPORTATION TABLE (49 C.F.R. Section 172.101) and amendments thereto

ENVIRONMENTAL PROTECTION AGENCY (40 C.F.R. Part 302 and amendments thereto)

All substances, materials and wastes that are, or that become, regulated under, or that are classified as hazardous or toxic under any environmental law, whether such laws are Federal, State or local.

- 1.1.14 The term "Leasehold Mortgagee", wherever used herein, means any lender of Company holding the beneficial interest in any Leasehold Mortgage and which Leasehold Mortgagee is a bank, savings and loan, insurance company, pension trust, real estate investment trust, mortgage company, or other institutional lender unaffiliated with the Company.
- 1.1.15 The term "Leasehold Mortgage", wherever used herein means any mortgage or deed of trust which is in favor of a Leasehold Mortgagee encumbering the Company's interest in this Agreement, the leasehold estate created hereunder and/or the Facilities from time to time located on the Premises and which has been incurred for, and the proceeds of which are used for, the purpose of financing the construction of the Facilities and the payment of other Project Costs or refinancing an existing Leasehold Mortgage. The proceeds of any refinancing, to the extent not used to pay off the Leasehold Mortgage being refinanced, shall be used for the purpose of: (i) repairing, constructing or reconstructing the Facilities; (ii) held in reserve for such purposes; (iii) the payment of other Project Costs; (iv) used for the repayment of Equity Contributions and interest thereon or (v) used for such other purposes as Lessor and Company may, in writing, agree. No Leasehold Mortgage shall encumber any interest in any real property other than Company's leasehold interest under this Agreement and the Facilities.
- 1.1.16 The term "Management Fee," whenever used herein, means the payment, as a Project Cost, to Company or any property manager it selects for the administration and management of the Premises and the Facilities. Such fee shall be three percent (3%) of Total Revenue received by Company from Tenants and Sublessees each month during the term of this Agreement. The Management Fee shall be paid to Company or its property manager on a monthly or other basis selected by Company, as funds are available from Total Revenue to pay the Management Fee, before any payments of Rent or repayment of Equity Contributions.
- 1.1.17 The term "Maintenance and Operations expense," whenever used herein, means the expense for maintenance, operation, administration and repair of the Premises and the Facilities, including, but not limited to, repair, maintenance or replacement of the plumbing, heating, ventilating and air conditioning systems, electrical systems, lighting facilities, fire protection systems, utility installations, fixtures, walls, foundations, roof, ceilings, floors, structural systems, doors, glass, skylights, landscaping and irrigating systems, driveways, parking lots, fences, retaining walls, signs, sidewalks, and the cost of all janitorial service, trash disposal, water, gas, electricity, and other utilities, together with any taxes thereon.
- 1.1.18 The term "Net Revenue," whenever used herein, means the amount of cash available after deducting from Total Revenue in the following order of priority: (a) Debt Service; (b) all Project Costs incurred during such calendar year of the term of this Agreement; (c) reasonable reserves for future Maintenance and Operations expense, future Capital

- Improvement expense or any reserve required by any Leasehold Mortgagee and (d) the repayment of Equity Contributions and interest thereon.
- 1.1.19 The term "Participating Parties" or "Parties," whenever used herein, means Company as lessee and the Successor Agency to the Industry Urban-Development Agency as Lessor (hereinafter jointly referred to as "Parties"), their successors and assigns.
- 1.1.20 The term "Premises," whenever used herein, means that area described on Exhibit "A". Final legal descriptions of the Premises will be attached to the Memorandum of Lease described in Section 1.2.4.
- 1.1.21 Except as otherwise provided in this Agreement, the term "Project Cost," whenever used herein, means all necessary costs incurred by Company (excluding Rent) prior to or after the date hereof in connection with the subdividing, designing, constructing, owning, leasing, financing or managing the Premises or the Facilities and other improvements constructed on the Premises by Company, or its successors and assigns. Except as otherwise provided in this Agreement, Project Cost shall include (without duplication), but not be limited to, the following: (a) all architectural and engineering expense, (b) all plan check fees, building permit fees, school fees, drainage fees and all other governmental, railroad and utility fees, licenses and permit cost, (c) all costs of environmental impact reports, traffic studies, biological studies and all other reports and studies required by governmental agencies, railroads or utility companies, (d) all costs to subdivide and re-subdivide the Premises, (e) all costs incurred by Company in constructing the Facilities or other permitted improvements on the Premises, (f) all Capital Improvement Expense, (g) all Maintenance and Operations expense, (h) all Management Fees, (i) except as provided in Section 2.12.2.10, all costs incurred for uninsured losses, earthquake, flood or other casualty, or repairs or replacements to the Premises and the Facilities or other improvements and the unreimbursed portion of any insured losses, (i) except as provided in Article III, all costs of complying with Environmental Laws, (k) all insurance premiums for insurance required hereunder and property taxes, (1) all leasing commissions and all other marketing and advertising expense, (m) all legal and accounting fees, (n) all loan fees, points, appraisal fees and other costs associated with the obtaining of Leasehold Mortgages, (o) all interest on Leasehold Mortgages, (p) all costs incurred by Company to create a community facilities district or districts or other assessment or other public financing district or districts to finance the construction of street improvements, utilities or other public improvements upon the Premises and (q) all other expenses related to the use, maintenance, leasing, financing and operation of the Premises and the Facilities. Project Costs shall only include costs, expenses and other amounts to the extent the same are reasonable and competitive in amount. Without limiting the preceding sentence, any costs, expenses or other amounts specified in an Approved Budget or otherwise approved in writing by LDR shall be deemed a reasonable and competitive amount. Further, Project Cost shall not include any cost, expense or other amount which is included within any other cost, expense or amount which is permitted by this Agreement to be deducted from Total Revenue in calculating Net Revenue, the intention of the Parties being that no cost, expense or other amount which is so permitted to be deducted from Total Revenue be deducted therefrom more than once.
- 1.1.22 The term "Sublease," whenever used herein, means the documents signed by a Sublessee or Tenant of Company for the leasing of a portion of the Premises or space in the Facilities.

- 1.1.23 The term "Sublessee" or "Tenant," whenever used herein, means any individual, corporation, limited liability company, trust, business, firm or other entity that leases or otherwise occupies or uses all or any portion of the Premises or the Facilities under a lease, rental agreement or other arrangement with Company.
- 1.1.24 The term "Release," whenever used herein, means any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, disposing or dumping of any Hazardous Material in violation of Environmental Laws.
- 1.1.25 The term "Total Revenue," whenever used herein, means the total amount of all rents, charges, fees and/or other income derived or received, directly or indirectly, by Company from the use, operation and/or leasing of all or part of the Premises or the Facilities, including, without limitation, all rents, charges, fees and/or other income and amounts received from Sublessees or Tenants of all or any part of the Premises or the Facilities during each full or partial calendar year of the term of this Agreement and the net proceeds received by Company as a result of a Total Taking, Partial Taking or Temporary Taking from an eminent domain proceeding referred to in Section 2.19. Insurance proceeds are excluded from Total Revenue as they are covered by other provisions of this Agreement.
- 1.1.26 The term "Equity Contribution" whenever used herein, means any funds provided by Company, Lessor or any Leasehold Mortgagee in accordance with the provisions of Section 1.7.1.1 or Section 1.7.1.2 hereof.
- 1.1.26 The term "Rent" whenever used herein, means fifty percent (50%) of Net Revenue during each calendar year of the term of this Agreement.

1.2 TERM AND CONDITIONS PRECEDENT

- 1.2.1 The term of this Agreement shall commence upon the expiration of 30 days following approval by the First District Consolidated Oversight Board and the California Department of Finance, if, and/or, as required by law.
- 1.2.2 The term of this Agreement will expire upon the earlier of: (a) sixty five (65) years from the recording of a Notice of Completion on the first building constructed by Company on the Premises or (b) sixty eight (68) years from the Commencement Date.
- 1.2.3 Lessor and Company agree to execute a memorandum of this Agreement evidencing the existence of this Agreement, the ownership of the Facilities by Company, the rights of Company in the Premises, the fact that this Agreement contains a right of first refusal to purchase the Premises and setting forth the Commencement Date and Termination Date of this Agreement. When the first Notice of Completion is recorded, Company and Lessor shall execute and record an amended memorandum of this Agreement setting forth the exact Termination Date.
- 1.2.4 Company will be entitled to receive, as a Project Cost, an ALTA leasehold policy of title insurance issued by a title company selected by Company, with liability in an amount reasonably determined by Company and insuring Company's interests hereunder. Such leasehold policy will be subject only to exceptions permitted by Company. By September 1, 2019, Lessor shall deliver a preliminary title reports covering the Premises,

as well as copies of the underlying document listed as exceptions to title is such report, to Company. If Company objects to any exceptions, it will give written notice of such objections (the "Notice of Objections") to Lessor within such one hundred twenty (120) days after receipt of the preliminary report and the underlying documents. Company's failure to give a Notice of Objections within such One Hundred Twenty (120) day period shall be conclusively deemed to be Company's acceptance of all exceptions. Within thirty (30) days following receipt of a Notice of Objections, Lessor, at its option, may cure, or agree to cure prior to the commencement of the term, some or all of the exceptions that are the subject of the Notice of Objections. If at the end of such thirty (30) day period, Lessor has not cured, or agreed to cure prior to the commencement of the term, any or all of such exceptions, Company may elect by written notice given within fifteen (15) days thereafter to terminate this Lease (the "Termination Notice"). Company's failure to give the Termination Notice within such fifteen (15) day period shall be conclusively deemed to be Company's election to accept such leasehold policy subject to all then uncured exceptions. Upon any termination of this Agreement pursuant to the provisions of this Section, the parties shall be released from all further liabilities and obligations under this Agreement, except for such as have accrued prior to the date of such termination. Notwithstanding the foregoing, Lessor agrees to remove, prior to the commencement of the term of this Lease, all loans secured by deeds of trust or mortgages encumbering the Premises as well as any leases or rental agreements encumbering all or any portion of the Premises. Lessor agrees not to create or permit the creation of any additional exceptions to title, after the date of the preliminary title report, without the written consent of Company.

As Facilities are constructed on the Premises, Company shall be entitled to obtain, as a Project Cost, additional ALTA leasehold title insurance with liability in an amount reasonably determined by Company and insuring Company's interests and the interest of any Leasehold Mortgagee hereunder.

1.2.5 Any amendments to this Agreement will not be effective as against any Leasehold Mortgagee unless Company has obtained the written consent of such Leasehold Mortgagee.

1.3 PREMISES

1.3.1 Lessor does hereby demise and let unto Company and Company does hereby take from Lessor that certain real property (hereinafter referred to as "Premises") as follows:

The land described on Exhibit "A" which is attached hereto and made a part hereof, together with all improvements located thereon (which improvements are owned by Company) together with all appurtenances, rights, privileges and easements appertaining thereto.

1.3.2 Lessor shall cause the work described on Exhibit "B" to be performed in accordance with the provisions of Exhibit "B" which is attached hereto and made a part hereof. Lessor shall commence such work within thirty (30) days after the date of this Agreement and shall complete such work as soon as reasonably possible, subject to extensions due to force majeure events as provided in Section 4.1. Lessor shall use commercially reasonable efforts to coordinate the timing of such work with Company's development of the Premises. All such work shall be at Lessor's sole cost and expense and shall not be considered an Equity Contribution by Lessor.

1.4 USE OF PREMISES

- 1.4.1 Upon performance of the agreements, provisions and conditions contained in this Agreement, Company will have the use of the Premises for the redevelopment of the Premises and construction and operation of Facilities and for any other business activities related thereto and for retail, warehouse, industrial, office and any other use permitted by applicable zoning regulations and for no other purposes, unless approved in writing by the LDR. So long as no default has occurred and is continuing hereunder, Lessor covenants peaceful and quiet enjoyment of the Premises by Company. Lessor covenants and warrants that the Premises are not now affected by any covenants, restrictions, easements or agreements which prohibit the construction, operation and maintenance upon the Premises of commercial structures with appurtenant parking, nor do the same prevent access to and from the Premises and the roadways adjoining the Premises.
- 1.4.2 Company acknowledges that the Rent payable to Lessor under this Agreement is directly dependent upon the generation by Company of Net Revenue, and that the failure by Company to use commercially reasonable efforts to maximize Net Revenue will have a direct and adverse effect upon Lessor's economic return under this Agreement. Therefore, as a material inducement to Lessor to enter into this Agreement, Company agrees that it will at all times construct, manage, lease and operate all Facilities in a manner and through the use of commercially reasonable efforts so as to maximize Net Revenue throughout the term of this Agreement.

1.5 DEVELOPMENT AND SUBLEASING

- 1.5.1 Company will develop the Premises and construct the Facilities substantially in accordance with plans and specifications prepared by Company in accordance with applicable governmental statutes, ordinances, rules and regulations, including those of the City of Industry and the Successor Agency to the Industry Urban-Development Agency.
- 1.5.2 Company may enter into subleases with Sublessees or Tenants on terms and conditions that are satisfactory to Company, provided such subleases are on sublease forms that have been approved by the LDR. LDR must approve or disapprove any sublease forms within 15 days of submittal by Company. Failure of the LDR to give written notice of disapproval, stating the specific objections to any such form, within the time specified shall be deemed approval.
 - 1.5.2.1 Company may negotiate changes to such approved forms which are typically given to Sublessees or Tenants in similar transactions without the consent of Lessor. In the event there are substantive changes or exceptions to the standard forms of sublease arrangements which are not typically given to Sublessees or Tenants in similar transactions, the Company must obtain the written approval of LDR, which approval will not be unreasonably withheld. LDR must approve or disapprove any such requested change within 2 business days of submittal by Company. Failure of the LDR to give written notice of disapproval, stating the specific reason for disapproving the requested change within the time specified shall be deemed approval.

- 1.5.2.2 All Subleases must be for purposes as permitted in Section 1.4, entitled USE OF PREMISES, of this Agreement and shall expressly provide that they are subject and subordinate to all provisions of this Agreement.
- 1.5.2.3 Except pursuant to Subleases as provided in this Section 1.5, Company shall not suffer or permit the use or occupancy of all or any portion of the Premises by any person or entity.
- 1.5.2.4 Rental interruption insurance or rental abatement insurance (as the case may be) shall be maintained (if available) by Company or Sublessee's or Tenants' providing for the payment of at least six (6) months' rent, taxes, insurance and maintenance expenses payable by Sublessees or Tenants under its Sublease.
- 1.5.3 If requested by Lessor, the Company will provide Lessor with a copy of any rules, regulations or other standards of operation developed by Company and distributed to Sublessees and Tenants.
- 1.5.4 In the event that the sublease of a portion of the Premises constitutes a change of ownership for property tax purposes, it is the desire of the Parties that such change in ownership and reappraisal of the subleased property be limited to the particular portion of the Premises which are the subject of such sublease and not be deemed a change of ownership for assessment purposes for any remaining portion of the Premises.

1.6 BUDGET APPROVAL

- 1.6.1 Prior to the construction of any new building on the Premises, Company shall submit to Lessor copies of a written construction budget for such new building. Each such budget will show all projected Project Costs related to the design, construction, financing and leasing of such new building.
 - 1.6.1.1 Within fifteen (15) days of receipt of the proposed budget, the LDR will review and approve or disapprove the proposed budget submitted by the Company. If the LDR does not respond in writing within fifteen (15) days, the budget shall be deemed to be the Approved Budget.
 - 1.6.1.1.1 If disapproved, the LDR will inform Company in writing of its disapproval describing the disapproved provisions of the proposed budget or requesting further clarification of the budget elements. Company will respond within fourteen (14) days with verification of the budget elements or with a modified written budget that is reasonably satisfactory to the LDR. The Participating Parties agree to negotiate in good faith to resolve any conflicting issues that may arise. If the LDR fails to timely respond, the proposed budget will be deemed approved and will become an Approved Budget.
 - 1.6.1.1.1 If, however, the Participating Parties cannot agree upon the provisions of the proposed budget and/or the elements contained in the proposed budget, or, if following approval of the Approved Budget, disputes arise concerning the

Approved Budget which the Parties are unable to resolve through good faith negotiation, a neutral third party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises will be selected by the LDR to arbitrate the disputed terms.

- 1.6.1.1.2 If, however, the Company does not accept the neutral third party selected by the LDR, Company will be allowed to select a second neutral party with similar experience. The two selected parties will then select a third neutral party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises who will arbitrate the disputed terms.
- 1.6.1.1.3 The LDR and Company agree to be bound by the decisions reached by the selected arbitrator(s). The Participating Parties will cause the arbitrator(s) to make a determination within fourteen (14) days following submittal.
- 1.6.1.1.4 The Participating Parties agree that each will bear its own costs and expenses incurred for attorney's fees and all other preparation and presentation costs for the arbitration process. If only one arbitrator is selected, the Participating Parties will equally share the cost of such arbitrator. If three arbitrators are used, each Party will bear the cost of the arbitrator selected by it and the Participating Parties will equally share the cost of any third arbitrator.
- 1.6.1.2 The agreed upon budget will be deemed the Approved Budget for the proposed building.
- 1.6.2 Company will be entitled to expend funds in accordance with the Approved Budget for the construction of the proposed building. The cost of constructing any such building shall be a Project Cost. In the event Company is over-budget on a particular line item, Company may reallocate excess funds from one line item to another line item to cover any short falls in any line item. Any expenses not covered by the Approved Budget which increase the total cost of constructing such building by more than ten percent (10%) are subject to the reasonable written approval of LDR. Any Equity Contribution required to cover any such additional Project Cost shall be the responsibility of Company.

1.7 RENTALS AND EQUITY CONTRIBUTIONS

Rentals due under this Agreement and Equity Contributions for the construction and operation of the Facilities will be as follows:

1.7.1 Equity Contributions.

- 1.7.1.1 The Parties anticipate that from time to time during the term of this Agreement, the Company will obtain financing for the development of the Premises and the Facilities and other Project Costs in accordance with the terms and conditions of Section 2.18 entitled FINANCING. Any funds required for the completion of the initial development of the Premises and construction of the Facilities, in excess of the financing obtained by Company, shall be provided by Company as an Equity Contribution.
- 1.7.1.2 Any additional capital required for the operation or maintenance of the Premises and the Facilities or other Project Costs following completion of construction of the Facilities on any separate legal parcel of the Premises and any capital required to repay any Leasehold Mortgages thereon, shall be contributed fifty percent (50%) by Company and fifty percent (50%) by Lessor within thirty (30) days of the receipt of demand by Company for such capital, which demand shall be accompanied by information and evidence, in reasonable detail, substantiating the amount and reasons for such demand. If any party fails to make such Equity Contribution, the other party may make such Equity Contribution for the delinquent party.
- 1.7.1.3 In the event of default by Company and the subsequent foreclosure and sale of the leasehold interest to another party, or in the event of a deed in lieu of foreclosure, the total unpaid balance of the Leasehold Mortgage at the date of the foreclosure sale or recording of the deed in lieu of foreclosure, including all costs of foreclosure, shall be considered an Equity Contribution of the Leasehold Mortgagee or purchaser at foreclosure, as the case may be, as of the date of such foreclosure sale or the date of recording the deed in lieu of foreclosure.
- 1.7.1.4 Any portion of Total Revenue remaining after payment by Company of Debt Service and all costs (except Rent) associated with the ownership, operation, financing, maintenance and leasing of the Premises and the Facilities, during each calendar year of the term of this Agreement will be applied to the Parties Equity Contributions until such time as all Equity Contributions are repaid in full together with interest thereon. All Equity Contributions made in accordance with the provisions of Section 1.7.1.1 and all Equity Contributions made in accordance with Section 1.7.1.3 shall be repaid in full prior to the repayment of any Equity Contributions made in accordance with Section 1.7.1.2. All Equity Contributions shall bear interest from the date of contribution until repaid at the lower of the annual rate of interest publicly announced from time to time by Bank of America, N.A. as its "prime" rate of interest plus three (3%) percent or the maximum rate allowed by law. The interest rate payable hereunder shall fluctuate with any change in the "prime" rate, and such fluctuations in the interest rate shall be effective on the effective date of each and every change in the "prime" rate as, from time to time, announced by Bank of America, N.A. as its "prime" rate of interest. Should Bank of America, N.A. no longer exist or fail to announce a "prime" rate, the "prime" rate, for purposes of this Lease, shall be the "prime" rate published in The Wall Street Journal, or if The Wall street Journal fails to publish a "prime" rate, the rate that most closely approximates the

- "prime" rate published in The Wall Street Journal or other similar financial publication.
- 1.7.1.5 The Participating Parties will acknowledge the date any equity contribution is paid in full by written notice from the Company and acknowledgment by the LDR.

1.7.2 Rent.

- 1.7.2.1 Fifty percent (50%) of all Net Revenue, if any, during each calendar year or partial calendar year during the term of this Agreement shall be paid to Lessor as Rent for the Premises.
- 1.7.3 Rent shall be paid to Lessor in monthly installments on or before the twenty-fifth (25th) day following the end of each calendar month. Such payments shall be accompanied by a written statement setting forth in detail the sources and amount of Total Revenue received for the preceding month and allowable deductions for the Net Revenue calculation. A check for the Lessor's fifty percent (50%) share of Net Revenue will be submitted with such statement. Except as otherwise provided in this Agreement, all Rent shall be paid without deduction, offset or abatement.
- 1.7.4 Company will make all payments to the Lessor at 15625 East Stafford Street, Suite 200, City of Industry, California 91744 or to such other place as Lessor may direct Company in writing.
- 1.7.5 In addition to, and without limiting, any other right or remedy which Lessor may have under this Agreement or at law or in equity, in the event any required payment is not made by Company to Lessor as required and remains unpaid for a period of thirty (30) days or more, the Lessor will be entitled to, and Company will pay to the Lessor, interest at the lower of ten percent (10%) per annum or the maximum amount allowed by law on all amounts unpaid and which remain unpaid thirty (30) days past the due date.
- 1.7.6 On or prior to April 30, annually during the term of this Agreement and within one hundred twenty (120) days after the expiration of the term of this Agreement, Company will provide Lessor with a statement showing in detail the sources and amount of Total Revenue received for the entire preceding calendar year. Such statement shall be prepared by the Company's chief financial officer in accordance with sound accounting principles and practices and shall contain a written opinion of such officer as to whether the Total Revenue calculations and distribution of Net Revenue and Rent has been made in accordance with the provisions of this Agreement. Should such statements show that the amount of Rent paid during the period of review was less than that which was due, such statement shall be accompanied by Company's payment of the additional amount to Lessor. Should such statement show that Company paid Lessor more than was due, after review and verification by LDR, a credit will be issued to be applied against future monthly installments of Rent, except that if such should be the case at the end of the last month of this Agreement, Lessor will refund the overpayment to Company.

1.8. RECORDS AND AUDIT

1.8.1 Throughout the term of this Agreement, Company shall keep and maintain, in accordance with sound accounting principles and practices, accurate and complete books, records and

accounts of Total Revenue and all items constituting deductions for purposes of calculating Net Revenue. Within ten days of request by Lessor, Company agrees to provide for inspection and copying by the Lessor or its designated representatives, at a location in the metropolitan area of Los Angeles County, California, such books, records, and accounts for the then current and three (3) immediately preceding calendar years. Such inspection shall take place Monday through Friday, 9:00 AM to 5:00 PM, holidays excluded.

1.8.2 Lessor will, at any time, have the right to cause an audit of the Company's operation of the Premises and of Total Revenue and all items constituting deductions for purposes of calculating Net Revenue, for the then current or three (3) immediately preceding calendar years. Such audit shall be made by a Certified Public Accountant of Lessor's selection. If such audit discloses that the calculation of Net Revenue previously provided to Lessor by Company is understated (either intentionally or unintentionally) by a greater margin than one percent (1%) of Company's Total Revenue for the period of review, then Company will immediately pay to Lessor the cost of such audit; otherwise the cost of the audit will be paid by Lessor. In all events, Company shall pay to Lessor the additional payments shown to be payable to Lessor by Company.

1.9 IMPROVEMENTS, MAINTENANCE AND REPAIR BY LESSOR

1.9.1 Except as provided in Exhibit "B" and Section 3.1.2 dealing with contamination of the Premises, Lessor has no direct responsibility or obligation for any maintenance, repair or replacement of the Premises or improvements.

1.10 CONSTRUCTION OF IMPROVEMENTS BY COMPANY

- 1.10.1 Company will, construct and install the Facilities including, grading, fencing, paving, lighting, driveways, roads, railroad facilities, utilities, parking lots, drainage, buildings and other improvements.
- 1.10.2 In the event Company has not commenced the construction of Facilities or submitted a proposal for the construction of Facilities on a parcel of the Premises within 10 years after the Commencement Date, Lessor shall have the right to terminate this Agreement as to any such undeveloped parcel of the Premises. Lessor shall give Company and any Leasehold Mortgagee encumbering any such undeveloped parcel of the Premises ninety days prior written notice and the right to submit a proposal for the development of any such parcel or commence construction thereon before exercising its right to terminate. In the event a proposal for the development of such undeveloped parcel is submitted within said ninety days, Lessor shall not have said right to terminate. Lessor agrees not to exercise its right to terminate as to any such undeveloped parcel until any Leasehold Mortgagee encumbering such parcel has been given its rights to cure or foreclose as set forth in Section 2.18.
- 1.10.3 All improvements or alterations by Company will be in accordance with all applicable governmental rules and regulations. Upon completion of any initial or additional improvements during the term hereof, Company will provide as-built drawings of same to the Lessor along with a certification of construction costs for all permanent improvements.

1.10.4 At any time during the term or any extension of this Agreement, Company may, as a Project Cost, add to or alter initially constructed Facilities, subject to all conditions set forth in 1.10.3 above. Any such addition or alteration will be performed in a workmanlike manner in accordance with all applicable governmental regulations and requirements and will not weaken or impair the structural strength or reduce the value of the Premises or improvements thereon.

1.11 MAINTENANCE AND REPAIR

- 1.11.1 Throughout the term of this Agreement, Company shall maintain, or cause to be maintained, as a Maintenance and Operations expense, the Premises and the Facilities in good condition and repair (including the making of any necessary replacements, both structural and nonstructural), ordinary wear and tear excepted, and in accordance with all applicable laws, rules, regulations and orders of all (i) federal, state, county, municipal, and other governmental agencies and bodies having jurisdiction, (ii) all insurance companies insuring all or any part of the Premises or the Facilities or both. Company's obligation shall include, but not be limited to, all structural and nonstructural portions of the Premises and the Facilities, all heating, ventilating and air conditioning systems and escalators, parking lots, driveways, driving lanes, sidewalks, walkways, common areas and landscaped areas.
- 1.11.2 If Company shall fail to perform any of its obligations under this Section 1.11, Lessor may, but shall not be obligated to, give Company written notice specifying such failure in reasonable detail. If Company does not remedy such failure, or does not commence to remedy such failure and thereafter diligently pursue such remedy to completion, within fifteen (15) days following receipt of such notice, Lessor may, but shall not be obligated to, remedy such failure. If Lessor remedies any such failure, all reasonable costs and expenses incurred by Lessor in connection therewith shall be paid to Lessor by Company within ten (10) days following written demand, which shall be accompanied by copies of receipts for all such expenses incurred. Following such payment, the amount paid may be included as a Maintenance and Operations expense.

1.12 APPROVALS TO BE REASONABLY GIVEN

It is understood and agreed that all provisions of this Agreement that require approval by the Lessor or the LDR will receive timely response and such approvals will not be unreasonably withheld. Unless another time limit is otherwise specified in this Agreement, any approval or disapproval must be given within twenty (20) days. If the Lessor or LDR does not respond in writing within twenty (20) days, the item for which Company requests approval shall be deemed approved. If disapproved, the LDR will inform Company in writing of its disapproval and state the reason for such disapproval. The Participating Parties agree to negotiate in good faith to resolve any conflicting issues that may arise. If, however, the Participating Parties cannot agree upon the item that was disapproved, a neutral third party with at least ten 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises will be selected by the LDR to arbitrate the disputed issue. If, however, the Company does not accept the neutral third party selected by the LDR, Company will be allowed to select a second neutral party with similar experience. The two selected parties will then select a third neutral party with at least 10 years experience in real estate developments similar to those constructed or proposed to be constructed on the Premises who will arbitrate the disputed issue. The LDR and Company agree to be bound by the decisions reached by the selected arbitrator. The Participating Parties will cause the arbitrator to make a determination within fourteen (14)

days following submittal. The Participating Parties agree that each party will bear its own costs and expenses incurred for attorney's fees, preparation and presentation costs for the arbitration process. The Participating Parties will share the cost of any third arbitrator.

Notwithstanding the foregoing, in the event of an emergency, the Company may take immediate, commercially reasonable action to prevent loss, injury or damage to persons or property or to preserve the Premises and the Facilities without the prior approval of Lessor and any costs reasonably incurred in connection therewith shall be a Project Cost.

ARTICLE II

2.1 ASSIGNMENT

- 2.1.1 Company will not assign this Agreement or its rights or duties hereunder or any estate created hereunder, in whole or in part, except with the prior written approval of Lessor, which approval will not be withheld unreasonably or delayed; provided, that it shall be reasonable for Lessor to withhold its approval if the assignee presented is not a proper and fit person or entity with financial resources and demonstrated business experience sufficient in Lessor's reasonable business judgment to be capable of performing the obligations hereunder which are the subject of the assignment. Further, any such assignment will be specifically subject to all provisions of this Agreement. Any assignment without the prior written approval of Lessor is Void.
 - 2.1.1.1 Except as provided in Section 2.1.1.3, any voluntary or involuntary transfer of fifty percent (50%) or more of Company's or any permitted successor's or assign's voting common stock, or the transfer of fifty percent (50%) or more of the partnership or membership interest, or the acquisition or transfer of fifty percent (50%) or more of Company's ownership, or the transfer of substantially all of the assets of the Company or any such successor or assign will be deemed an assignment requiring the prior written approval of Lessor.
 - 2.1.1.2 As a condition precedent to any assignment, Company shall provide Lessor reasonable prior written notice of the proposed assignment with such appropriate documentation and other information as Lessor shall reasonably request in order for Lessor to make its decision to grant or withhold approval. If the prior written approval of Lessor is given to any such assignment, the proposed assignee shall, in recordable form, expressly assume all of the terms, covenants and conditions of this Agreement which are the subject of the assignment. Any assignment will not release Company from its obligations under this Agreement arising from events occurring prior to the date of assignment, but shall release Company from all obligations arising after the effective date of the assignment.
 - 2.1.1.3 Any transfers by partners or members of Company or shareholders of partners or members of Company to each other or for estate purposes or upon death will not be considered an assignment hereunder.
- 2.1.2 The foregoing provisions relating to assignment shall not be applicable to a Leasehold

Mortgage hypothecating or encumbering Company's interest in this Agreement and the leasehold estate created thereby and the Facilities located on the Premises and further shall not be applicable to any sale by judicial foreclosure or pursuant to a power of sale by any Leasehold Mortgagee or to any transfer in lieu of such sale, or to any sales or transfers by any Leasehold Mortgagee (or any affiliate, parent or subsidiary of any Leasehold Mortgagee that acquires title to the Company's interest in this Agreement and the leasehold estate created thereby and the Facilities as a result of a sale by judicial foreclosure or pursuant to a power of sale or any transfer in lieu of such sale), subsequent to a sale by judicial foreclosure or pursuant to a power of sale or subsequent to any transfer in lieu of such sale. Any further assignments shall be subject to the provisions of Section 2.1, 2.1.1.1, 2.1.1.2 and 2.1.1.3.

2.2 SUBLEASING

Company will not sublease, sublet or rent to, or permit any persons, firms or corporations to occupy any part of the Premises or the Facilities without having complied with the following:

2.2.1 Company may enter into any Subleases which comply with Section 1.5 entitled DEVELOPMENT AND SUBLEASING. A copy of any such executed Sublease will be provided to the LDR. Any arrangements for the leasing of space which are not in conformance with Section 1.5 entitled DEVELOPMENT AND SUBLEASING shall not be entered into without the prior written approval of the LDR.

2.3 ATTORNMENT

- 2.3.1 All Subleases entered into by the Company will be subject to all terms and conditions of this Agreement. If Company defaults under this Agreement, all Sublessees will recognize Lessor as the successor to the Company under their respective Subleases, and will render performance thereunder to Lessor as if the Sublease were executed directly between Lessor and the Sublessees; provided, however, Lessor agrees that so long as a Sublessee is not in default under its Sublease, Lessor will provide quiet enjoyment to such Sublessee and Lessor agrees to be bound by all of the terms and conditions of each such Sublease.
- 2.3.2 All Subleases entered into by the Company will contain the following provision:
 - If, by reason of a default on the part of Company in the performance of the terms and provisions of the underlying Agreement, the underlying Agreement and the leasehold estate of Company thereunder is terminated by summary proceedings or otherwise in accordance with the terms of the underlying Agreement, Sublessee will attorn to Lessor and recognize Lessor as lessor under this Sublease; provided, however, Lessor agrees that so long as Sublessee is not in default, Lessor agrees to provide quiet enjoyment to Sublessee and to be bound by all the terms and conditions of this Sublease.
- 2.3.3 In the event this Agreement is terminated for any reason, all Sublessees will be liable to Lessor for their payment of rents and fees and will be subject to all the provisions and terms contained in their Subleases.

2.4 SUCCESSORS AND ASSIGNS

Subject to the provisions of this Article II, this Agreement shall inure to the benefit of and be binding upon the legal representatives, successors and assigns of the Parties.

2.5 CONTROL OF PERSONNEL

Company will, in and about the Premises, exercise reasonable control over the conduct, demeanor and appearance of its employees, agents and representatives and the conduct of its contractors and suppliers. The provisions of this Section 2.5 shall in no way limit the liability of the Company with respect to the negligence, acts or omissions of its employees, agents, representatives, contractors and suppliers, nor shall such provisions limit any obligations of indemnity on the part of the Company under this Agreement.

2.6 SIGNS

Company and its Sublessees and Tenants shall have the right to install, erect and operate signs upon the Premises so long as such signs are installed, erected, operated and maintained in compliance with all applicable laws and ordinances.

2.7 ENTRY AND INSPECTION OF PREMISES

- 2.7.1 Lessor, its authorized officers, employees, agents, contractors, subcontractors or other representatives will have the right to enter upon all portions of the Premises and the Facilities constituting public areas when the same are open to the public, and shall have the right to enter upon all other portions of the Premises and the Facilities during normal business hours, for the purpose of inspecting the Premises and the Facilities, determining whether Company is in compliance with the terms and provisions of this Agreement, and for fulfilling Lessor's obligations and exercising Lessor's rights hereunder; provided however, that such entry will be in such manner as to not unreasonably interfere with the operations or reasonable security requirements of Company or its Sublessees. Except in the event of an emergency, Lessor shall provide Company at least two (2) business days prior written notice of its intention to enter upon the non-public areas of the Premises or the Facilities for such purposes, and, if Company so elects, Lessor shall be accompanied by a representative of Company during such entry.
- 2.7.2 No such entry upon the Premises by or on behalf of Lessor will cause or constitute a termination of this Agreement nor be deemed to constitute an interference with the possession thereof nor constitute a revocation of or interference with any of Company's rights in respect thereof for exclusive use of the Premises.
- 2.7.3 The rights of entry for inspection and other purposes as contemplated by the parties to this Agreement, pursuant to this Section, are for the sole benefit of the parties. No benefit to any third party is contemplated or intended.

2.8 INTENTION OF PARTIES

This Agreement is intended solely for the benefit of Lessor and Company and is not intended to benefit, either directly or indirectly, any third party or member(s) of the public at large. Any work done or inspection of the Premises by Lessor is solely for the benefit of Lessor and Company.

2.9 LIENS

Subject to the rights to contest set forth in Section 2.10 entitled Taxes and Assessments, Company will cause to be removed any and all tax liens and liens arising out of or because of any construction or installation performed by or on behalf of Company or any of its contractors or subcontractors upon the Premises or arising out of or because of the performance of any work or labor to it or them at the Premises or the furnishing of any materials to it or them for use at the Premises. Should any such lien be made or filed, Company will bond against or discharge the same within thirty (30) days after written request by LDR. The cost of bonding against or discharging any liens relating to construction or installation of the Facilities shall be a Project Cost.

2.10 TAXES AND ASSESSMENTS

Throughout the term of this Agreement, Company shall pay before delinquency, as a Project Cost, all real and personal property taxes and assessments, including, without limitation, all city, county, school district and other taxes and general and special assessments, levied upon or assessed against the Premises, any Facilities now or hereafter located thereon, the leasehold estate, any personal property of Company located on or in the Premises or any Facilities, or arising in respect of the occupancy, use or possession of the Premises or the Facilities located thereon and which are assessed or become due during the term of this Agreement. All such taxes for the first and last years of the term hereof shall be equitably prorated between the parties. Company shall have the right, in Company's or Lessor's name, to contest the validity of any tax or assessment by appropriate proceedings timely instituted and diligently prosecuted. Lessor, if requested by Company, shall cooperate with Company in any such proceedings; provided, that all costs and expenses (including attorneys' fees and costs) incurred by Lessor in connection therewith shall be paid by Company as a Project Cost. Company shall not be responsible for any of Lessor's franchise, inheritance, income, succession, transfer or other tax levied on the Lessor's rights in and to the Premises or Lessor's right to receive income from the Premises or under this Agreement. Company will keep current municipal, state or local licenses or permits required for the conduct of its business.

2.11 INDEMNITY

During the term of this Agreement, Company agrees to indemnify and hold Lessor forever harmless from and against all claim, liability, loss, demand, judgments or other expense (including, but not limited to, defense costs, expenses and reasonable attorney fees) imposed upon Lessor by reason of injuries or death of persons (including wrongful death) and damages to property caused during and because of Company's use or occupancy of the Premises or any actions or non-actions of Company, its officers, employees, agents, or other representatives, including movement of vehicles, provided, however, that (without limiting Lessor's right to coverage under any applicable insurance policies) such indemnity will not apply as to any negligent or intentional act or omission of Lessor, its employees, agents or representatives.

During the term of this Agreement, Lessor agrees to indemnify and hold Company harmless from and against all claim, liability, loss, demand, judgments or other expense (including, but not limited to, defense costs, expenses and reasonable attorneys fees) imposed upon Company by reason of injuries or death of persons (including wrongful death) and damages to property caused by Lessors use or occupancy of the Premises or any actions or non-actions of Lessor, its officers, employees, agents, contractors or other representatives, including movement of vehicles, provided, however, that (without limiting Company's right to coverage under applicable insurance policies) such indemnity will not apply to any negligent or intentional act or omission of Company, its employees, agents or representatives.

2.12 INSURANCE AND BONDS

2.12.1 Bonds

- 2.12.1.1 No bonds will be required for any Facilities constructed by Commerce Construction Co. Otherwise, Company, as a Project Cost, will require its general contractor on any construction contract in excess of two million dollars to furnish bonds covering the faithful performance of the construction of the improvements or installation of equipment, payment of all obligations arising thereunder to take effect upon completion of the project. Bonds may be secured through the contractor's usual sources provided the surety is authorized and licensed to do business in the State of California. Company will be allowed to name any Leasehold Mortgagee as an additional obligee under any such bond.
- 2.12.1.2 Prior to execution of a construction contract with a contractor required to furnish bonds, and not later than ten (10) calendar days after notification of award, Company will require such contractor to furnish contract bonds to LDR as follows:
 - a. Labor and material payment bond in the amount of one hundred percent (100%) of the contract price.
 - b. Payment and performance bond in the amount of one hundred percent (100%) of the contract price.

LDR may, if considered appropriate in its sole judgment, waive the requirements of this Section 2.12.1.2 upon written request by Company.

- 2.12.1.3 The bonds referred to in Section 2.12.1.1 and 2.12.1.2 above will be written on payment and performance bond and labor and material payment bond forms generally used for similar projects in Southern California and approved by LDR.
- 2.12.1.4 Company will require its contractor to require the attorney-in-fact who executes the required bonds on behalf of the surety to affix thereto a certified and current copy of his power of attorney.

2.12.2 Insurance

2.12.2.1 Before any construction contractor (including Commerce Construction Co.) Commences Construction of any Facilities or any equipment installation on or about the Premises, the construction contractor shall procure and maintain insurance for such construction and installation protecting both Company and Lessor, as well as the construction contractor. Such insurance will provide coverage and limits as are commercially reasonable in the industry. Such insurance will include, but is not limited to:

General Liability on an "occurrence" basis only

Automobile Liability
Builder's Risk equal to the maximum contract cost of the project.

- 2.12.2.2 Company's (or its contractor's) insurance will be primary as respects
 Lessor and Company, their officers, employees and authorized agents.
 Any other coverage carried by or available to Lessor, its officers,
 employees and authorized agents will be excess over the insurance
 required of the Company or its contractor and shall not contribute with it.
- 2.12.2.3 Company and its contractors will maintain worker's compensation insurance in the amounts and form as required by California law.

 Certificates evidencing the valid, effective insurance policies will be provided to LDR.
- 2.12.2.4 Company will keep insured with responsible insurance underwriters the Facilities and any improvements from time to time constructed by it upon and within the Premises to the extent of not less than the full replacement value of such improvements from time to time (without deduction for depreciation) using an industry standard all risk form of protection. Company will use commercially reasonable efforts to cause its Sublessees or Tenants to keep insured with responsible insurance underwriters any improvements constructed by such Sublessees or Tenants to the same extent.
- Company will obtain and keep in full force and effect a policy(s) of 2.12.2.5 general liability insurance on an "occurrence" basis only and not "claims made." The coverage must be provided either on an ISO Commercial General Liability form, or an Accord Comprehensive General Liability form, or equivalent, approved by the LDR and Company. Any exceptions to coverages must be fully disclosed on the required Certificate. If other than these forms are submitted as evidence of compliance, complete copies of such policy forms will be submitted to LDR within ten (10) days after notice to Company. Policies must include, but need not be limited to, coverages for bodily injury, property damage, personal injury, Broad Form property damage, premises and operations, severability of interest, products and completed operations, contractual and independent contractors with no exclusions of coverage for liability resulting from the hazards of explosion, collapse, and underground property damage.

Company will maintain limits of no less than five million dollars (\$5,000,000) combined single limit per occurrence for bodily injury (including death), personal injury and property damage.

2.12.2.6 Company will furnish Automobile Liability coverage for claims for damage because of bodily injury or death of any person, or property damage arising out of the ownership, maintenance or use of any motor vehicles whether owned, hired or non-owned. Company will maintain limits of no less than five million dollars (\$5,000,000) combined single limit "per accident" for bodily injury and property damage.

- 2.12.2.7 All required insurance coverage as stated in Section 2.12.2 will be a Project Cost and evidenced by a current Certificate(s) of Insurance. Such Certificates will include, but will not be limited to, the following:
 - 2.12.2.7.1 All Certificates for each insurance policy are to be signed by a person authorized by that insurer and licensed by the State of California.
 - Each insurance company's rating as shown in the latest Best's Key Rating Guide will be fully disclosed and entered on the required Certificates of Insurance. If the insurance company providing the coverage has a Best rating of less than A+, the adequacy of the insurance supplied by Company (or its contractor), including the rating and financial health of each insurance company providing coverage, is subject to the approval by the LDR. Such approval will not be unreasonably withheld.
 - 2.12.2.7.3 Company (or its contractor) will furnish renewal Certificates for the required insurance during the period of coverage required by this Agreement. Company (or its contractor) will furnish renewal Certificates for the same minimum coverages as required in this Agreement.
 - 2.12.2.7.4 Lessor, its officers and employees shall be covered as additional insureds with respect to liability arising out of the activities by or on behalf of the named insured in connection with this Agreement. All property insurance policies will contain a waiver of subrogation clause in favor of Lessor.
 - 2.12.2.7.5 Each insurance policy supplied by the Company (or its contractor) must be endorsed to provide that the amount and type of coverage afforded to the Lessor by the terms of this Agreement will not be suspended, voided, canceled or reduced in coverage or in limits except after thirty (30) days' prior written notice by mail to Company, Lessor and any Leasehold Mortgagee.
 - 2.12.2.7.6 Any deductible, as it relates to coverage provided under this Agreement, will be fully disclosed on the Certificates of Insurance. Any deductible provided will be reasonable and customary for the type of risk.
 - 2.12.2.7.7 If aggregate limits are imposed on the insurance coverage, then the amounts of such limits must be not less than five million dollars (\$5,000,000) per occurrence or per accident. All aggregates must be fully disclosed and the amount entered on the required certificate of insurance. Company's insurer must notify the LDR of any erosion of the aggregate limits. The "per occurrence" limits of insurance required herein must be

maintained in full, irrespective of any erosion of aggregate. A modification of the aggregation limitation may be permitted if it is deemed necessary and approved by the LDR and Company.

- 2.12.2.8 If the Company fails to maintain any of the insurance coverages required herein, then the Lessor, after 10 days prior notice to Company, may purchase replacement insurance or pay the premiums that are due on existing policies in order that the required coverages may be maintained. The Company is responsible for any expenses paid by the Lessor to maintain such insurance and Lessor may collect the same from the Company.
- 2.12.2.9 The insurance requirements specified herein do not relieve the Company (or its contractor) of its responsibility or limit the amount of its liability to the Lessor or other persons and the Company is encouraged to purchase such additional insurance as it deems necessary.
- 2.12.2.10 Company (or its contractor) is responsible for and must remedy all damage or loss to any property, including property of Lessor, caused in whole or in part by the Company or its contractor, any subcontractor or anyone employed, directed or supervised by the Company; provided, however neither Company nor any Leasehold Mortgagee shall be liable for any uninsured or underinsured loss relating to the Premises or the Facilities, unless such loss is a result of Company's failure to obtain the insurance it is required to maintain under this Agreement, in which event Company shall be liable for such loss and such loss shall not be a Project Cost. Company is responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with this Agreement.
 - 2.12.2.11 Company and Lessor agree and acknowledge that any Leasehold Mortgagee shall be named as an additional insured under any insurance policy relating to the Premises, the Facilities or otherwise required under this Agreement. Company agrees to provide any Leasehold Mortgagee with appropriate certificates of insurance relating to all policies.
- 2.12.2.12 Company, at its option, may satisfy its obligations hereunder to insure within the coverage of any so-called blanket policy or policies of insurance which it now or hereafter may carry, by appropriate amendment, rider, endorsement or otherwise; provided, however, that the interests of Lessor shall thereupon be as fully protected by such blanket policy or policies as they would be if this option to so insure by blanket policy were not permitted.
- 2.12.2.13 Waiver of Subrogation. Without affecting any other rights or remedies, Company and Lessor each hereby release and relieve the other, and waive their entire right to recover damages against the other, for loss of or damage to its property arising out of or incident to the perils required to be insured against herein. The effect of such releases and waivers is

not limited by the amount of insurance carried or required, or by any deductibles applicable thereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such companies may have against Lessor or Company, as the case may be, so long as the insurance is not invalidated thereby.

2.13 DAMAGE AND DESTRUCTION

In the event that any Facilities located on the Premises are damaged or destroyed, and the net proceeds of insurance are sufficient to cover the Company's estimated cost of rebuilding (or if Company failed to maintain insurance it is required to maintain under this Agreement and the proceeds of such insurance, had it been maintained, would have been sufficient to cover the Company's estimated cost of rebuilding), Company shall repair and rebuild the said Facilities, and this Agreement shall continue in full force and effect, and Company shall commence and continue such repair or rebuilding with reasonable diligence and shall complete such repair and rebuilding within a reasonable time after the same is commenced; provided, however, that any delay in the completion of the said repairs resulting from fire or other casualty, strikes, shortages of material or labor, governmental laws, rules and regulations, the elements or matters beyond the reasonable control of Company, shall extend such reasonable time within which Company may complete said repairs or rebuilding by the period of such delay. The net proceeds of any insurance, to the extent that such proceeds are received by Lessor or any Leasehold Mortgagee, less any expenses of recovery thereof (including attorneys' fees), shall be made available to Company to be applied to the cost and expense of repair or rebuilding the damage or destruction insured against, subject to reasonable conditions and payable on the usual architect's certificates, but Lessor or any Leasehold Mortgagee holding said insurance proceeds may withhold an amount reasonably necessary to insure completion of such repairs or rebuilding (not to exceed ten percent (10%) of such proceeds) until completion and the expiration of the period within which mechanics' or materialmen's liens may be filed and until receipt of satisfactory evidence that no liens exist.

In the event the Company maintains the insurance it is required to maintain under this Agreement and the proceeds of insurance are not sufficient to cover the Company's estimated cost of rebuilding, or in the event the damage or destruction occurs during the last 10 years of the lease term, Company will have the option to terminate this Agreement, subject to the rights of any Leasehold Mortgagee, which option will be exercisable by written notice to Lessor within forth five (45) days after the occurrence of such event. Company will only have such option to terminate this Agreement as to the specific legal parcel or parcels upon which the Facilities are damaged or destroyed. Any such termination by Company shall require the prior written consent of any Leasehold Mortgagee. In the event the Company elects to terminate this Agreement based upon such damage, destruction, or substantial loss, the Company will be liable for and will pay for all cleanup or demolition of the Premises necessary to make the Premises ready for repair, replacement, restoration or rebuilding: (a) if the Company or its employees or agents caused such damage, destruction or substantial loss to occur, (b) to the extent of any insurance proceeds payable with respect to such damage, destruction or substantial loss, or (c) to the extent of any insurance proceeds which would have been payable with respect to such damage, destruction or substantial loss had Company maintained the insurance it is required to maintain under this Agreement. In the event Company does not exercise such option then Company will promptly repair, replace, restore or rebuild said improvements.

2.14 DEFAULT BY COMPANY

2.14.1 Default by Company

Subject to the provisions of Section 2.16 and Section 2.18, Company will be considered in default as Lessee under this Agreement in the event of any one or more of the following occurrences:

2.14.1.1 Company fails to pay the rent or any other money payments required by this Agreement when the same are due and the continuance of such failure for a period of thirty (30) days after written notice thereof from the LDR to Company or Company fails to maintain any insurance required to be maintained by it under this Agreement and such failure continues for a period of fifteen (15) days after written notice thereof from the LDR to Company.

2.14.1.2Company fails to perform any other covenant contained in this Agreement and such failure continues for a period of sixty (60) days after written notice thereof from the LDR to Company; provided, however, that if a nonmonetary default cannot be cured with reasonable diligence within such sixty (60) day period, then Lessor shall not have the right to terminate this Agreement or pursue any other remedy against Company, as long as Company commences the curing of such default within said sixty (60) day period and thereafter proceeds with the curing of such default to completion with reasonable diligence, with allowance for delays due to the action or failure to act of governmental authorities, strikes, acts of God or other matters beyond the reasonable control of Company.

2.14.1.3 Company voluntarily abandons the Premises for a period of sixty (60) days following written notice from the LDR to Company.

2.14.2 Cure

Subject to the mortgagee protection provisions set forth in this Agreement, Company will be considered in default of this Agreement if Company fails to fulfill any of the terms, covenants, or conditions set forth in this Agreement following the expiration of the cure period and written notice set forth in Sections 2.14.1 and 2.14.3.

2.14.3 Termination For Default By Company

Subject to the lender protection provisions of Section 2.18, if default is made by Company as described in Section 2.14.1 hereinabove, and such default is not cured as provided in Sections 2.14.1 and 2.14.2, Lessor shall provide Company with an additional written notice thereof and if such failure to cure continues for an additional period of sixth (60) days. Lessor shall be entitled to pursue any and all rights and remedies which it may have under this Agreement or at law or in equity, including, without limitation, the right to terminate this Agreement. All of such rights and remedies shall be cumulative and not alternative.

2.14.3.1 If Lessor elects to terminate this Agreement, it will in no way prejudice the right of action for rental arrearages owed by Company.

2.14.3.2 Redelivery and disposal of improvements will be as described in Section 2.17, entitled SURRENDER AT END OF TERM.

2.15 DEFAULT BY LESSOR

2.15.1 Default By Lessor

Lessor will be considered in default of this Agreement if Lessor fails to fulfill any of the terms, covenants or conditions set forth in this Agreement if such failure shall continue for a period of more than sixty (60) days after delivery by Company of a written notice of such default.

2.15.2 Cure

If the default cannot be cured with reasonable diligence within such Sixty (60) day period, then Company shall not have the right to terminate this Agreement or pursue any other remedy against Lessor, as long as Lessor commences the curing of such default within said sixty (60) day period and thereafter proceeds with the curing of such default to completion with reasonable diligence, with allowance for delays due to the action or failure to act of governmental authorities, strikes, acts of God or other matters beyond the reasonable control of Lessor.

2.15.3 Remedies

Subject to the mortgagee protection provisions set forth in this Agreement, if default is made by Lessor as described in Section 2.15.1 hereinabove, and is not cured as provided in Section 2.15.2 hereinabove, Company shall provide Lessor with an additional written notice thereof and if such failure to cure continues for an additional period of sixty (60) days, Company shall be entitled to pursue any and all rights and remedies which it may have under this Agreement or at law or in equity, including, without limitation, the right to terminate this Agreement. All of such rights and remedies shall be cumulative and not alternative. Redelivery and disposal of improvements will be as described in Section 2.17, entitled SURRENDER AT END OF TERM.

2.16 WAIVERS AND ACCEPTANCE OF FEES

- 2.16.1 No waiver of default by either party hereto of any of the terms, covenants or conditions hereof to be performed, kept or observed will be construed to be or act as a waiver of any subsequent default of any of the terms, covenants, conditions herein contained to be performed, kept and observed.
- 2.16.2 No acceptance of fees or other money payments in whole or in part for any period or periods during or after default of any of the terms, conditions or covenants to be performed, kept or observed by the Company will be deemed a waiver on the part of the Lessor of any right or remedy which it may have under this Agreement or at law or in equity.
- 2.16.3 No acceptance of fees or other money payments in whole or in part for any period or periods during or after default of any of the terms, conditions or covenants to be performed, kept or observed by the Lessor will be deemed a waiver on the part of the Company of any right or remedy which it may have under this Agreement or at law or in equity.

2.17 SURRENDER AT END OF TERM

- 2.17.1 Subject to the mortgagee protection provisions set forth in Section 2.18 and subject to the rights of Sublessee's, Company covenants that at the termination of this Agreement, howsoever caused, it will quit and surrender the Company's right, title and interest in the Premises in good repair and condition, excepting reasonable wear and tear, acts of God, the public enemy or the action of the elements.
- 2.17.2 All Facilities located on the Premises at the expiration or sooner termination of the term of this Agreement, howsoever caused, shall, without payment or compensation of any kind to Company, then become Lessor's property, in fee simple, free and clear of all claims, liens and encumbrances to or against them by Company, any Leasehold Mortgagee, any Sublessee or Tenant, or any other third party claiming by or through Company. Subject to the mortgagee protection provisions set forth in Section 2.18 and subject to the rights of Sublessee's, upon termination of this Agreement howsoever caused, Company shall have the right to remove from the Premises, within sixty (60) days following the Termination Date, all equipment, trade fixtures and personal property belonging to Company or Sublessee's, provided Company or its Sublessees repair any damage caused by such removal.

For purposes of this Subsection 2.17.2, the words "equipment, trade fixtures and personal property" will include, but not be limited to, (i) signs (electrical or otherwise), (ii) all equipment used in connection with the conduct of Company's or its Sublessees business whether or not such equipment is attached to the Premises or the Facilities, (iii) any other mechanical device, and (iv) all other miscellaneous equipment, furnishings and fixtures installed on or placed on or about the Premises and used in connection with Company's or its Sublessees business thereon; provided, however, that such words shall not include elevators, escalators, plumbing systems, electrical systems, life safety systems, boilers, heating, ventilating and air conditioning systems, floor and wall coverings, ceiling lights, built-in shelving and cabinets, doors, windows, outside walls and fencing, and landscaping. All equipment, trade fixtures and personal property that is not removed within sixty (60) days following the Termination Date, shall become the property of Lessor.

2.17.3 Without limiting the provisions of this Section 2.17, Company agrees that upon the expiration or sooner termination of this Agreement, howsoever caused, it will execute, acknowledge and deliver to Lessor within thirty (30) days after demand from Lessor, all such documents and instruments as shall be reasonably necessary to evidence and confirm the transfer of ownership of the Facilities from Company to Lessor, "as-is" in their existing condition without any further consideration required from Lessor.

2.18 FINANCING

2.18.1 Notwithstanding anything to the contrary contained in this Agreement, Company will have the right at such time or times as Company desires and without Lessor's consent, to hypothecate Company's interest in all or part of this Agreement, the Premises and the Facilities with one or more Leasehold Mortgages. Such Leasehold Mortgages may contain such terms and conditions as are acceptable to Company and Company will have the right, without the consent of Lessor, at any time during the term hereof to execute and deliver to any or all of its Leasehold Mortgagees any documents which will operate as

collateral security for any loan or loans made, even if such document or documents result in a form or type of conveyance or assignment of the leasehold interest demised hereunder. It is hereby agreed that Company or any such Leasehold Mortgagees will have the right to immediately record such document or document(s) with an appropriate public official or officials. Company agrees that copies of all such documents of conveyance or assignment as contained in this Section 2.18 will be provided to the LDR forthwith. Lessor agrees to cooperate in executing any documents reasonably requested of Lessor by Company or the Leasehold Mortgagee in connection with any such Leasehold Mortgage; PROVIDED, HOWEVER, UNDER NO CIRCUMSTANCES SHALL LESSOR BE OBLIGATED TO SUBORDINATE ITS FEE INTEREST IN THE PREMISES TO ANY LEASEHOLD MORTGAGE, AND, NOTWITHSTANDING ANY TERM OR PROVISION OF ANY SUCH LEASEHOLD MORTGAGE OR THIS AGREEMENT TO THE CONTRARY, UNDER NO CIRCUMSTANCES SHALL ANY SUCH LEASEHOLD MORTGAGE CONSTITUTE AN INDEBTEDNESS OR OBLIGATION OF LESSOR NOR SHALL LESSOR BE LIABLE IN ANY WAY FOR THE PAYMENT OF ANY PORTION OF THE INDEBTEDNESS EVIDENCED BY SUCH LEASEHOLD MORTGAGE OR FOR THE PAYMENT OR PERFORMANCE OF ANY OTHER OBLIGATION THEREUNDER OR SECURED THEREBY.

- 2.18.2 Lessor will deliver to any such Leasehold Mortgagee written notice of any default of Company under the terms of this Agreement and said notice will specify the nature of the default. Before terminating this Agreement, Lessor will allow such Leasehold Mortgagee to cure or commence to cure any default of Company in accordance with the provisions of this Agreement. The time period to cure any default of Company will commence when said notice is delivered to Leasehold Mortgagee and Leasehold Mortgagee shall have the same lengths of time to cure the specified default as are permitted Company in Section 2.14. In the event Company fails to timely cure a default after receipt of written notice and expiration of any applicable cure period, Lessor agrees to provide any Leasehold Mortgagee with a second written notice and provide such Leasehold Mortgagee with an additional thirty (30) day cure period. Lessor will not have the right to exercise any remedies under this Agreement so long as a Leasehold Mortgagee is diligently prosecuting to complete a cure of any default. If such default is of a nature which is incapable of being cured by Leasehold Mortgagee, Lessor agrees not to exercise its remedies arising from such default if (i) Leasehold Mortgagee notifies Lessor in writing within such thirty (30) day cure period that Leasehold Mortgagee intends to foreclose its mortgage and Leasehold Mortgagee commences and diligently pursues such foreclosure; and (ii) Leasehold Mortgagee makes all payments due by Company under this Agreement through the date of foreclosure.
- 2.18.3 Any default by the Company in the payment of money as required under the terms of this Agreement may be cured by the Leasehold Mortgagee in accordance with the terms of Sections 2.14 and 2.18.2 of this Agreement, and Lessor will accept any such payment or cure from such Leasehold Mortgagee during the term of the Leasehold Mortgage.
 - 2.18.3.1 Should the Company default under the terms of this Agreement and should the default be such that it cannot be cured by the payment of money, Lessor will accept payments of rent from such Leasehold Mortgagee and this Agreement will not terminate, but will remain in full force and effect, pending Leasehold Mortgagee's cure of such default within the time periods described herein or resort to foreclosure or sale proceedings under its deed of trust or other security instruments.

- 2.18.4 If any default has been cured by a Leasehold Mortgagee or Assignee, Lessor agrees that upon completion of any foreclosure proceedings or sale under the deed of trust or other security instrument securing the loan, or upon delivery of a deed in lieu of foreclosure, the Leasehold Mortgagee or purchaser at such sale or any heir, successor, or assign of Leasehold Mortgagee (Assignee) subsequent to such sale will be recognized by Lessor as the lessee under the terms of this Agreement for all purposes for the remaining term hereof. The leasehold interest of the Leasehold Mortgagee or such Assignee will not be adversely affected or terminated by reason of any nonmonetary default occurring prior to the completion of such proceedings or sale, provided such default has been promptly remedied (to the extent it is capable of being remedied by Leasehold Mortgagee), or if such default requires possession to cure, provided such Leasehold Mortgagee, Assignee or purchaser promptly commences to cure upon taking possession of the Premises.
- 2.18.5 Such Leasehold Mortgagee will not become personally liable under the terms and obligations of this Agreement unless and until it assumes the obligations and is recognized by Lessor as lessee under this Agreement and will be liable only so long as such Leasehold Mortgagee or Assignee maintains ownership of the leasehold interest or estate and recourse to such Leasehold Mortgagee or Assignee shall be limited solely to Leasehold Mortgagee's or Assignee's interest in the Premises.
- 2.18.6 At any time during the term of this Lease, within ten (10) days after a written request by the Company or any Leasehold Mortgagee, Lessor, through Lessor's Designated Representative, will execute, acknowledge and deliver to the Company or such person or entity as the Company designates, a certificate stating:
 - (a) that this Agreement is the only Agreement between Lessor and Company concerning the Premises and is unmodified and in full force and effect in accordance with its terms (or if there have been modifications, that this Agreement is in force and effect as modified, and identifying the modification agreements, or if this Agreement is not in full force and effect, that it is not);
 - (b) the commencement and expiration dates of this Agreement and the date to which rental has been paid to Lessor under this Agreement;
 - (c) whether or not, to the knowledge of the Lessor, there is an existing default by Company in the payment of Rent or any other sum of money under this Agreement, and whether or not there is any other existing default by either party under this Agreement with respect to which a notice of default has been served, and if there is such a default specifying its nature and extent;
 - (d) whether or not, to the knowledge of Lessor, there are any set-offs, defenses or counter-claims against enforcement of the obligations to be performed by Lessor or Company under this Agreement; and,
 - (e) such other reasonable information relating to this Agreement that a Leasehold Mortgagee or assignee may request.
- 2.18.7 All notices of default required to be delivered to a Leasehold Mortgagee under this Agreement will be sent to any Leasehold Mortgagee by overnight courier service or certified mail, return receipt requested. No such notice shall be valid or effective as

- against the Leasehold Mortgagee until and unless actually received by the Leasehold Mortgagee as evidenced by the courier service's delivery records or the return receipt.
- 2.18.8 In the event this Agreement is terminated for any reason prior to the end of the term (it being the intent that the Agreement will remain in full force and effect if Leasehold Mortgagee performs), Lessor shall enter into a new Agreement with Leasehold Mortgagee holding a first priority interest covering the Premises, provided that the Leasehold Mortgagee (i) requests such new Agreement by written notice to Lessor within sixty (60) days after termination, and (ii) cures all prior defaults of Company that are capable of being cured by Leasehold Mortgagee. The new Agreement shall be for the remainder of the term, effective at the date of such termination, at the same rent and on the same covenants, agreements, conditions, provisions, restrictions and limitations contained in this Agreement. The new Agreement shall have the same title priority as this Agreement and shall be subject only to the exceptions to title having priority over this Agreement or such additional exceptions to which such Leasehold Mortgagee has consented in writing. In the event the Lessor and such Leasehold Mortgagee enter into any such new Agreement, title to all the Facilities located upon the Premises as of the date of such new Agreement shall automatically vest in such Leasehold Mortgagee. If requested by such Leasehold Mortgagee, Lessor agrees to execute and deliver to Leasehold Mortgagee within 10 days after request therefore, a quitclaim deed in recordable form conveying title to such Facilities to such Leasehold Mortgagee. Lessor also agrees to assign to such Leasehold Mortgagee all Subleases with Tenants whose Subleases attorn to Lessor upon the termination of this Agreement.
- 2.18.9 Lessor shall cooperate in amending this Agreement from time to time to add any provision which may reasonably be requested by any Leasehold Mortgagee or proposed lender for the purpose of implementing the mortgagee protection provisions contained in this Agreement and allowing such Leasehold Mortgagee reasonable means to protect or preserve the lien of its loan and the value of its security. Lessor agrees to execute and deliver (and to acknowledge if necessary for recording purposes) any agreement necessary to effect any such amendment; provided, however, that any such amendment shall not in any way affect or change the term of this Agreement or the Rent or other amounts payable to Lessor under this Agreement, subordinate the fee interest of Lessor in the real property underlying the Premises, nor otherwise in any material respect adversely affect any rights of Lessor under this Agreement.
- 2.18.10 The bankruptcy or insolvency of Company will not operate or permit the Lessor to terminate this Agreement as long as all Rent or other monetary payments required to be paid by Company continue to be paid and other required obligations are performed in accordance with the terms of this Agreement.
- 2.18.11 To the extent any of the terms of this Agreement are inconsistent with the terms of this Section regarding mortgagee protection provisions, the mortgagee protection provisions will control.
- 2.18.12 Every Leasehold Mortgage shall contain a provision that copies of all notices of default by Company thereunder must be sent to Lessor. In the event of any default by Company under any Leasehold Mortgage, the Lessor reserves the right to make any payments due to the Leasehold Mortgagee before the Leasehold Mortgagee resorts to any foreclosure or sale proceedings under its deed of trust or other security instrument.

- 2.18.12.1 Following any foreclosure or deed in lieu of foreclosure the leasehold estate created by this Agreement and the Facilities may be transferred or assigned as provided in Section 2.1.2.
- 2.18.13 Any mortgage placed by the Lessor on the fee title to the Premises shall be subordinate to this Agreement (and any replacement to this Agreement), any separate lease entered into in accordance with Section 4.8 and all Subleases.
- 2.18.14 Prior to any termination of this Agreement by Lessor, any Leasehold Mortgagee shall be allowed sufficient time to complete any foreclosure action, including time for delays due to official restraint (including by law, process or injunction issued by a court), so long as such Leasehold Mortgagee is making payments required by this Agreement which can be reasonably determined prior to acquiring the Company's interest under this Agreement. Leasehold Mortgagee shall have the right to terminate foreclosure proceedings at any time if Company has cured all defaults under any loan from Leasehold Mortgagee. However, no such termination of foreclosure proceedings shall prevent Lessor from pursuing, or continuing to pursue, its rights and remedies under this Agreement against Company for any uncured defaults under this Agreement.
- 2.18.15 So long as any Leasehold Mortgage is in effect, there shall be no merger of the leasehold estate created by this Agreement into the fee simple estate in the Premises without the prior written consent of the Leasehold Mortgagee.
- 2.18.16 Any Leasehold Mortgagee shall have the right to participate in any settlement or adjustment of losses under insurance policies maintained by Company under this Agreement. Such Leasehold Mortgagee shall be named as a loss payee or additional insured, as applicable, in accordance with any loan documents executed by Company, under the insurance policies required under this Agreement. Company shall use commercially reasonable efforts to obtain provisions in Leasehold Mortgages providing that any proceeds of insurance shall first be used for the purposes provided for in this Agreement before any portion thereof is applied to repay any indebtedness under such Leasehold Mortgage.

2.19 EMINENT DOMAIN

- 2.19.1 In the event of any acquisition of or damage to all or any part of the Premises or any interest therein by eminent domain, whether by condemnation proceeding or transfer in avoidance of an exercise of the power of eminent domain or otherwise during the term of this Agreement, the rights and obligations of the parties with respect to such appropriation shall be as provided in this Section. As used herein:
- a. "Taking" shall mean any taking or damage, including severance damage, by eminent domain or by inverse condemnation or for any public or quasi-public use under any statute. The transfer of title may be either a transfer resulting from the recording of a final order in condemnation or a voluntary transfer or conveyance in lieu of an exercise of eminent domain or while condemnation proceedings are pending. The taking shall be deemed to take place on the earlier of (i) the date actual physical possession is taken by the condemnor or (ii) the date on which the title passes to the condemnor.
- b. "Total Taking" shall mean the taking of fee title to all of the Premises or so much

of the Premises that the portions of the Premises not so taken are, in the reasonable judgment of Company, not reasonably suited for the uses of the Premises that have been previously made by Company. A Temporary Taking, defined below, may be treated as a Total Taking if it meets the test set forth above.

- c. "Partial Taking" shall mean any taking of fee title that is not a Total Taking.
- d. "Temporary Taking" shall mean the taking for temporary use of all or any portion of the Premises for a period ending on or before the expiration of the term.
- e. "Award" shall mean all compensation paid for the taking whether pursuant to a judgment or by agreement or otherwise.
- f. "Notice of Intended Taking" shall mean any notice or notification on which a reasonably prudent man would rely and which he would interpret as expressing an existing intention of taking as distinguished from a mere preliminary inquiry or proposal and shall include without limitation the service of a condemnation summons and complaint on either Lessor or Company or the receipt by either Lessor or Company from a condemning agency or entity of a written notice of intent to take containing a description or map of the taking reasonably defining the extent thereof.
- 2.19.2 Upon receipt of any of the following by either Party hereto, such Party shall promptly deliver a copy thereof, endorsed with the date received, to the other and any Leasehold Mortgagee which has given its name and address to such Party for such purpose.
 - a. Notice of Intended Taking;
 - b. Service of any legal process relating to the condemnation of all or any part of the Premises;
 - c. Notice in connection with any proceedings or negotiations with respect to any such condemnation; and
 - d. Notice of intent or willingness to make or negotiate a private purchase, sale or other transfer in lieu of condemnation.
- 2.19.3 Lessor, Company and any Leasehold Mortgagee that would be affected by such taking, shall each have the right to represent its respective interest in each proceeding or negotiation with respect to any taking or intended taking and to make full proof of its claims. No agreements, settlement, sale or transfer to or with the condemning authority shall be made without the prior written consent of Lessor, Company and Leasehold Mortgagee, which consent shall not be unreasonably withheld. Each of the parties hereto agrees to execute and deliver to the other any instruments that may be required to effectuate or facilitate any of the provisions of this Section where such execution or delivery will not adversely affect the right of such party to receive just compensation for any loss sustained in such negotiation or proceeding.
- 2.19.4 In the event of a Total Taking, this Agreement and all interests, rights, liabilities and obligations of the Parties hereunder shall terminate as of the date of Taking, except for liabilities and obligations arising out of events occurring prior to the date of termination.

In the event of a Taking of fee title to less than all of the Premises which Company reasonably believes to be a Total Taking, Company may, by written notice to Lessor approved by all Leasehold Mortgagees, within one hundred twenty (120) days after Company receives Notice of Intended Taking thereof, elect to treat such taking as a Total Taking. If Company fails to make such timely election, such Taking shall be deemed to be a Partial Taking for all intents and purposes.

Any taking determined to be a Total Taking in accordance with the foregoing paragraph shall be treated as a Total Taking if (i) Company delivers possession of the Premises to Lessor within one hundred twenty (120) days after Company shall have delivered written notice to Lessor electing to treat such taking as a Total Taking and (ii) Company has complied with all of the provisions hereof relating to the apportionment of the awards.

In a Total Taking, all sums, including damages and interest, awarded for the fee or the leasehold or both shall be distributed in the following order of priority:

- a. First, to Leasehold Mortgagees to the extent of the then balance due on all Leasehold Mortgages;
- b. Second, to Lessor an amount equal to the then present value of the reversionary interest of Lessor in the real property underlying the Premises at the Expiration of this Agreement;
- c. Third, subject to Section 1.1.25, the balance of the Award to Company.

All sums awarded for the leasehold or the fee shall (i) be delivered to Lessor and Company (or to Leasehold Mortgagee), respectively, if such award has been apportioned between Lessor and Company by such condemning authority, or (ii) deposited promptly with Leasehold Mortgagee (or in the event that there is no lender of record, with an escrow agent selected by Company in the reasonable exercise of its discretion), if only a single award is made, and distributed and disbursed as set forth above. Sums being held by the escrow agent pending disbursement shall be deposited in a federally insured interest bearing account and, upon distribution, each party having a right to any of the sums being disbursed shall be entitled to receive the interest attributable to its share of said sums.

2.19.5 In the event of a Partial Taking, this Agreement shall remain in full force and effect as to the portion of the Premises remaining.

On a Partial Taking, all sums, including damages and interest, awarded for the fee or the leasehold or both shall be distributed in the following order of priority:

- a. First, to Company an amount equal to the cost of making all repairs and restorations to any Facilities on the Premises affected by such taking to the extent necessary to restore the same to a complete architectural and economically viable or functioning units (to the extent permitted, however, taking into consideration the amount of land remaining after such taking or purchase);
- b. Second, to any Leasehold Mortgagee to the extend that the security if its Leasehold Mortgage has been impaired as a result of the Partial Taking or as

required by its loan documents, whichever is greater;

- c. Third, to Lessor an amount equal to the then present value of the reversionary interest of Lessor in the real property underlying the Premises at the Expiration of this Agreement in that portion of the real property underlying the Premises that is taken in such Partial taking;
- d. Fourth, subject to Section 1.1.25, the balance of the Award to Company.

All sums awarded for the leasehold or the fee shall (i) be delivered to Lessor and Company (or to Leasehold Mortgagee), respectively, if such award has been apportioned between Lessor and Company by such condemning authority, or (ii) deposited promptly with Leasehold Mortgagee (or in the event that there is no lender of record, with an escrow agent selected by Company in the reasonable exercise of its discretion), if only a single award is made, and distributed and disbursed as set forth above. Sums being held by the escrow agent pending disbursement shall be deposited in a federally insured interest bearing account and, upon disbursement, each party having a right to any of the sums being disbursed shall be entitled to receive the interest attributable to its share of said sums.

- 2.19.6 In the event of a Temporary Taking, this Agreement shall remain in full force and effect, neither the rents reserved hereunder nor the term hereof shall be reduced or affected in any way, and, subject to Section 1.1.25, Company shall be entitled to any award for the use or estate taken, subject to the requirements of any Leasehold Mortgagee.
- 2.19.7 Unless the respective values are determined by the court in the eminent domain proceeding, the values of the interests for which Lessor and Company are entitled to compensation in the event of a Total or Partial Taking shall be determined by the mutual written agreement of Lessor and Company. If Lessor and Company are unable to agree on the value of said interests within thirty (30) days after the deposit of the sums awarded with the escrow agent, then within thirty (30) days after the expiration of that period, each such party shall submit its good faith estimate of the value of said interests as of the date of the taking. If the higher of said estimates is not more than 105% of the lower of such estimates, the value shall be the average of the submitted estimates. If otherwise, then within ten (10) days the question shall be submitted to arbitration pursuant to the rules of the American Arbitration Association.

ARTICLE III

3.1 ENVIRONMENTAL POLICY

3.1.1 Violation Of Environmental Laws

During the term of this Agreement, Company will not cause or permit any Hazardous Material to be used, generated, manufactured, produced, stored, brought upon, or released on, under or about the Premises, or transported to and from the Premises, by Company, its Sublessees and Tenants, or their respective agents, employees, contractors, invitees or a third party in violation of the Environmental Laws.

3.1.1.1 LDR will have access to the Premises to inspect same to insure that

Company is using the Premises in accordance with Environmental Laws.

- Company, at the LDR's reasonable request, as a Project Cost, will conduct such testing and analysis as necessary to ascertain whether Company is using the Premises in compliance with environmental requirements. Any such tests will be conducted by qualified independent experts chosen by Company and subject to LDR's reasonable written approval. Copies of such reports from any such testing will be provided to LDR.
- 3.1.1.3 Company will promptly provide copies of all notices, reports, claims, demands or actions concerning any environmental concern or release or threatened release of Hazardous Materials to Lessor.

3.1.2 Contamination Of Premises

- 3.1.2.1 If the illegal use of any Hazardous Material on, under or about the Premises Company, its agents, employees or contractors results in any contamination of the Premises, in violation of an Environmental Law, Company will promptly take all actions as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Company will take all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual harm and to bring the Premises into compliance with all environmental requirements. If any such contamination is a result of the gross negligence of Company or its employees or if Company or its employees intentionally contaminate the Premises, all costs incurred to bring the Premises into compliance with environmental laws shall be at Company's cost and expense and not as a Project Cost. If any such contamination is a result of the actions of agents, contractors, third parties or the actions of Company that were not intended to contaminate the Premises, all costs incurred to bring the Premises into compliance with environmental laws shall be a Project Cost.
- 3.1.2.2 Lessor will be solely responsible for any environmental condition existing on or about the Premises prior to the commencement of the term of this Agreement or any environmental conditions caused by Lessor, its agents, employees, contractors or invitees during the term. Lessor will promptly take all actions, at its cost and expense, as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Lessor will take all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual harm and to bring the Premises into compliance with all environmental requirements.
- 3.1.2.3 If the illegal use of any Hazardous Material on, under or about the Premises by Sublessees and Tenants, their respective agents, employees, contractors, servants or invitees results in any contamination of the Premises, in violation of an Environmental Law, Company will promptly take all actions, as a Project Cost, as are necessary to return the Premises to the condition existing prior to the introduction of any such Hazardous Material to the Premises. Company will take

all steps necessary to remedy and remove any such Hazardous Materials as are necessary to protect the public health and safety and the environment from actual harm and to bring the Premises into compliance with all environmental requirements. All amounts recovered (whether by way of settlement of a claim, judgment, indemnity or otherwise) by Company from any such Sublessee, Tenant, agent, employee, contractor, servant or invitee in connection with the foregoing matters shall be included in Total Revenue for the full or partial calendar year of the term of this Agreement in which recovered (Company's reasonable costs of recovery shall be deducted as a Project Cost), or if recovered following the expiration or termination of the term of this Agreement, fifty percent (50%) of such recovery (after deducting from such recovery the Company's reasonable costs of recovery not theretofore included in Project Costs) shall be promptly paid over to Lessor.

3.1.3 Compliance With All Governmental Authorities

During the term of this Agreement, Company will promptly make all submissions required of Company and comply with all requirements of the appropriate governmental authority under all Environmental Laws. During the term of this Agreement, Lessor will promptly make all submissions required of Lessor to, and comply with, all requirements of the appropriate governmental authority under all Environmental Laws.

- 3.1.3.1 Should any governmental agency having authority over such matters determine that a site characterization, site assessment, and/or cleanup plan be prepared or that a cleanup should be undertaken because of any spills or discharges of hazardous materials at the Premises by Company, its agents, employees or contractors which occur during the term of this Agreement, then Company shall prepare and submit the required plans and financial assurances, and carry out the approved plans. If any such spills or discharges are a result of the gross negligence of Company or its employees or if Company or its employees intentionally spilled or discharged hazardous materials at the Premises, all costs incurred to prepare and submit the required plans and carry out the approved plans shall be at Company's cost and expense and not as a Project Cost. If any such spills or discharges are a result of the actions of agents, contractors, third parties or the actions of Company that were not intended to spill or discharge hazardous materials at the Premises, all costs incurred to prepare and submit the required plans and carry out the approved plans shall be a Project Cost.
- 3.1.3.2 Company hereby agrees to indemnify Lessor from all costs incurred in connection with any investigation of site conditions or any cleanup, remediation, removal or restoration work required by any Federal, State or local governmental agency or political subdivision because of an environmental condition on, under or about the Premises that is caused by Company, its Sublessees, their agents, employees, contractors or invitees. If any such environmental condition is a result of the gross negligence of Company or its employees or if Company or its employees intentionally caused such environmental condition, all costs incurred to bring the Premises into compliance with environmental laws shall be at Company's cost and expense and not as a Project Cost. If any such environmental condition is a result of the actions of Sublessees, agents, contractors, third parties or the actions of Company that were not intended to contaminate the Premises, all costs incurred to bring the Premises into

compliance with environmental laws shall be a Project Cost. Lessor hereby agrees to indemnify Company from all costs incurred in connection with any investigation of site conditions or any cleanup, remediation, removal or restoration work required by any Federal, State or local governmental agency or political subdivision because of an environmental condition on, under or about the Premises that existed prior to the Commencement Date of this Agreement or was caused by Lessor, its agents, employees, contractors or invitees during the term.

3.1.3.3 Should any governmental agency having authority over such matters determine that a site characterization, site assessment, and/or cleanup plan be prepared or that a cleanup should be undertaken because of any spills or discharges of hazardous materials at the Premises by Sublessees and Tenants, and their respective agents, employees, contractors or invitees which occur during the term of this Agreement, then Company shall (as a Project Cost) prepare and submit the required plans and financial assurances, and carry out the approved plans. Company will promptly provide all information requested by Lessor to determine the applicability of the Environmental Laws to the Premises, or to respond to any governmental investigation or to respond to any claim of liability by third parties which is related to such environmental contamination. All amounts recovered (whether by way of settlement of a claim, judgment, indemnity or otherwise) by Company from any such Sublessee, Tenant, agent, employee, contractor, servant or invitee in connection with the foregoing matters shall be included in Total Revenue for the full or partial calendar year of the term of this Agreement in which recovered (Company's reasonable costs of recovery shall be deducted as a Project Cost), or if recovered following the expiration or termination of the term of this Agreement, fifty percent (50%) of such recovery (after deducting from such recovery the Company's reasonable costs of recovery not theretofore included in Project Costs) shall be promptly paid over to Lessor.

ARTICLE IV

4.1 FORCE MAJEURE

Neither Lessor, Company or any Leasehold Mortgagee will be deemed to be in breach of this Agreement by reason of failure to perform any of its obligations hereunder if, while and to the extent that such failure is due to strikes, boycotts, labor disputes, embargoes, shortages of materials, acts of God, acts of the public enemy, acts of governmental authority, unusual weather conditions, floods, riots, rebellion or sabotage. However, the provisions of this Section will not apply to, nor excuse any failure by Company to pay Rents, fees or any other money payments required under the provisions, covenants or agreements contained in this Agreement.

4.2 QUIET ENJOYMENT

Lessor agrees that, on payment of the Rents and fees and performance of the covenants, conditions and agreements on the part of Company to be performed hereunder, Company will have the right to peaceably occupy and enjoy the Premises.

4.3 NOTICES

All notices, requests, consents and approvals under this Agreement will be given only by personal delivery, certified mail return receipt requested, overnight courier service or facsimile. The same shall be deemed to have been given: (i) if personally delivered, upon receipt, (ii) if by certified mail, upon the date indicated on the return receipt, (iii) if by overnight courier service, upon the date delivered as shown by the records of the courier, and (iv) if by facsimile, at the time of electronic confirmation of successful transmission.

Notices intended for the Lessor will be addressed to:

Successor Agency to The Industry Urban-Development Agency 15625 East Stafford Street, Suite 200 City of Industry, California 91744 Attn: Executive Director

Facsimile: (626) 961-6795

or to such other address as may be designated by the Lessor by written notice to Company.

Notices intended for the Company will be addressed to:

Industry East Land, LLC c/o Majestic Realty Co. 13191 Crossroads Parkway North City of Industry, California 91746 Attn: Edward P. Roski, Jr.

Facsimile: (562) 692-1553

or to such other address as may be designated by the Company by written notice to Lessor.

4.4 HEADINGS, TITLES OR CAPTIONS

Article, section or paragraph headings, titles or captions are inserted only as a matter of convenience and for reference, and in no way define, limit or describe the scope or extent of any provision of this Agreement.

4.5 INVALID PROVISIONS

It is expressly understood and agreed by and between the parties hereto that in the event any covenant, condition or provision herein contained is held to be invalid by any court of competent jurisdiction, the invalidity of such covenant, condition or provision will in no way affect any other covenant, condition or provision herein contained; provided, however, that the invalidity of any such covenant, condition or provision does not materially prejudice either Lessor or Company in their respective rights and obligations contained in the valid covenants, conditions or provisions of this Agreement.

4.6 STATE OF CALIFORNIA LAW

This Agreement will be interpreted under and governed by the internal laws of the State of California, without regard to principles of conflicts of law.

4.7 ENTIRE AGREEMENT

This document represents the entire agreement between the parties hereto and it cannot be effectively modified or canceled by mutual agreement or in any manner, nor may any term or provisions of this Agreement be waived, except by an instrument in writing approved by Lessor and executed by duly authorized officials, officers or agents of the Parties.

4.8 SUBDIVISION OF THE PREMISES AND RIGHT TO SEPARATE LEASES

- The parties hereto acknowledge and agree that the Premises will be developed by Company with 4.8.1 Facilities, in phases, over a number of years. During the term of this Agreement, Company may, from time to time, as a Project Cost, to subdivide the Premises to create such separate legal parcels as Company reasonably deems desirable for the development of the Premises and to cause subdivision maps covering the Premises, or any portion thereof, to be prepared and processed through the appropriate governmental agencies, executed by Company and Lessor, and filed for record in accordance with all laws and ordinances applicable thereto. Subject to the foregoing, at any time, and from time to time during the term of this Agreement, Lessor agrees to, within 15 days after the written request of Company, execute and deliver such instruments as may be appropriate, necessary, or required for: (i) the grant or dedication of any easement, right of way or other property right to any public entity or service corporation for ingress, egress, road or utility purposes and reasonably required for the subdivision or development of the Premises, or (ii) the obtaining of governmental approvals, consents, zoning changes, conditional uses, variances, subdivision maps or the like, or for the purpose of providing adequate utility services to the Premises, or permitting Company to construct the Facilities or other improvements on the Premises or make any alteration or addition to the Facilities.
- 4.8.2 The parties further acknowledge and agree that as a result of such phased development, Company will be obtaining construction, interim and permanent financing for the development of various portions of the Premises and the Facilities at different times during the Term of this Agreement. In order to facilitate such phased development and to facilitate separate financing for various phases thereof, the parties hereby agree that the Company shall have the right, at any time, and from time to time, during the term of this Agreement, or any extension thereof, to have any portion of the Premises removed from the legal description of the Premises and to have the Lessor and Company enter into a new separate lease covering the portion of the Premises so removed from this Agreement.
 - 4.8.2.1 All such new leases shall be for the remainder of the Term of this Agreement and shall be on the same terms and conditions as this Agreement, including, without limitation, the right of first refusal as set forth herein. If requested by Company, Lessor agrees to grant any easements reasonably required for access, road or utility purposes between the Premises remaining subject to this Agreement and the portion of the Premises which becomes subject to any such new lease.
 - 4.8.2.2 At any time Company desires a separate lease, Company shall prepare an amendment to this Agreement deleting the portion of the Premises to be covered by the separate lease and shall also prepare such separate lease and submit both documents to Lessor for execution. Lessor agrees to execute such amendment and separate lease within twenty (20) days after Company submits such documents. Concurrently with the execution of each such amendment and separate lease, the Parties shall execute a memorandum in recordable form evidencing the existence of such separate lease, the portion of the Premises subject to such separate lease, the effective date and termination date of such separate lease and the fact that there is a right of first refusal to purchase the Premises. Concurrently with the execution of such amendment and separate lease, the parties shall

also execute a memorandum in recordable form evidencing the existence of such amendment to this Agreement. Any such separate lease shall have the same title priority as this Agreement and shall be subject only to the exceptions to title having priority over this Agreement or such additional exceptions to which Company has consented in writing. In the event the parties enter into any such separate lease, title to all Facilities located upon the Premises subject to such separate lease shall automatically vest in Company as of the effective date of such separate lease. If requested by Company, Lessor agrees to execute and deliver to Company within 10 days after request therefore, a quitclaim deed in recordable form conveying title to such Facilities on the premises subject to the separate lease to Company. Lessor also agrees to assign to Company all Subleases with Tenants whose Subleases attorn to Lessor upon the termination of this Agreement and creation of such separate lease.

4.8.2.3 The creation of a separate lease of a particular parcel that was included in the initial conveyance, pursuant to this Article 4.8 is intended to merely continue the initial grant. The amendment of this Lease to delete separately leased premises and the creation of any separate lease is not intended to: (1) expand the rights or interests created by the initial conveyance; (2) constitute a "creation" of a taxable possessory interest; (3) constitute a "renewal" or "extension" of a taxable possessory interest; and/or (4) grant any additional property or authorize uses not already permitted by this Lease.

4.9 RIGHT OF FIRST REFUSAL TO PURCHASE

At all times during the term of this Agreement and any extensions or renewals thereof, should Lessor receive a bona fide offer from any third party to purchase all or any portion of the Premises, which offer Lessor desires to accept, Lessor shall, before accepting such offer, provide Company with a full and complete copy of such offer and shall first offer in writing to sell the portion of the Premises which is the subject of such offer to Company on the same terms and conditions as set forth in said offer. Upon receipt of any such notice and copy of such offer from Lessor, Company shall have thirty (30) days thereafter within which to accept the same. Should Company fail to accept any such offer within said thirty (30) day period, Lessor shall be free to sell said portion of the Premises to the original offeror upon the same terms and conditions offered to Company without further notice to Company. Should Lessor after having made such offer to Company as above described, fail to sell said portion of the Premises upon the same terms and conditions offered to Company, Lessor shall give Company notice and the first right to purchase in the manner set forth above of any further or different offers received by Lessor for the purchase of said portion of the Premises and shall first offer to sell the same to Company upon the same terms and conditions before accepting any such further or different offer. The right of first refusal set forth herein is a continuing right of first refusal to purchase and shall apply to all subsequent bona fide offers from third parties after a sale by Lessor, or its successors, to a party other than Company.

4.10 UTILITIES

To the extent not paid by Sublessees, Company shall, as a Project Cost, pay for all water, gas, heat, light, power, telephone service, trash disposal and other utilities and services supplied to the Premises and the Facilities located thereon, together with any taxes thereon.

4.11 OWNERSHIP OF IMPROVEMENTS

All improvements constructed on the Premises by Company (including, without limitation, the Facilities) at any time and from time to time, will be owned by Company during the term of this Agreement. If requested by Company, any Leasehold Mortgagee or title insurance company, Lessor agrees to execute and deliver to Company within ten 10 days after request therefor, a quitclaim deed in recordable form conveying or confirming title to all such improvements in Company. All such buildings and improvements constructed upon the Premises by Company are and shall remain real property and may not be severed from this Agreement or the leasehold estate created hereby.

4.12 LEASING COMMISSIONS

In the event Company retains the services of Majestic Realty Co. or any affiliate of Company to Sublease the Premises and the Facilities, the leasing commissions paid to Majestic Realty Co. or any such affiliate, as a Project Cost, shall not exceed the scheduled commissions for similar services charged by the major brokerage companies operating in the County of Los Angeles, California.

4.13 COVENANTS FOR NON-DISCRIMINATION

The Company covenants by and for itself and any successors in interest that there shall be no discrimination against or segregation of any person or group of persons on account of race, color, creed, religion, sex, marital status, age, handicap, national origin or ancestry in the use and occupancy of the Premises, nor shall the Company itself or any person claiming under or through it establish or permit any such practice or practices of discrimination or segregation with reference to the use or occupancy of the Premises.

- 4.13.1 The Company shall refrain from restricting the rental or lease of the Premises on the basis of race, color, creed, religion, sex, marital status, handicap, national origin or ancestry of any person. All such leases or contracts shall contain or be subject to substantially the following nondiscrimination or nonsegregation clauses:
 - 1. In leases: "The lessee herein covenants by and for himself or herself, his or her heirs, executors, administrators and assigns, and all persons claiming under or through him or her, and this lease is made and accepted upon and subject to the following conditions:
 - "There shall be no discrimination against or segregation of any person or group of persons on account of race, color, creed, religion, sex, marital status, handicap, ancestry or national origin in the leasing, subleasing, transferring, use, occupancy, tenure or enjoyment of the premises herein leased nor shall the lessee himself or herself, or any person claiming under or through him or her, establish or permit any such practice or practices of discrimination or segregation with reference to the selection, location, number, use or occupancy of tenants, lessees, sublessees, subtenants or vendees in the premises herein leased."
 - 2. In contracts: "There shall be no discrimination against or segregation of, any person, or group of persons on account of race, color, creed, religion, sex, marital status, handicap, ancestry or national origin, in the sale, lease, sublease, transfer, use, occupancy, tenure or enjoyment of the premises, nor shall the transferee himself or herself or any person claiming under or through him or her, establish or permit any such practice or practices of discrimination or segregation with

reference to the selection, location, number, use or occupancy of tenants, lessees, subtenants or vendees of the premises."

4.13.2 The covenants established in this Section shall, without regard to technical classification and designation, be binding for the benefit and in favor of the Lessor, its successors and assigns, the City of Industry and any successor in interest to the fee ownership of the Premises or any part thereof. The covenants, contained in this Section shall remain in effect for the term of this Agreement.

4.14 PUBLIC FINANCING

The parties hereby agree and acknowledge that, in order to assist with the financing of construction of Streets A, B, C, D and Grand Crossing Parkway and the storm drain system, sanitary sewer, domestic, fire and reclaimed water systems, and dry utilities located within the public rights of way of Streets A, B, C, D and Grand Crossing Parkway, the parties may initiate proceedings to form one or more community facilities or other assessment or public financing districts which may encompass all or certain portions of the Premises. The parties hereby agree to cooperate in good faith with each other to (i) initiate such formation proceedings, (ii) form one or more community facilities or other assessment or financing districts over all or any portion of the Premises, and (iii) cause such public financing district or districts to issues bonds to assist with the financing of construction of Streets A, B, C, D and Grand Crossing Parkway and the storm drain system, sanitary sewer, domestic, fire and reclaimed water systems, and dry utilities located within the public rights of way of Streets A, B, C, D and Grand Crossing Parkway. Company and Lessor hereby agree that the Premises or any portion thereof, as determined by the parties, shall be included within the boundaries of said public financing district or districts. In addition, the parties hereto hereby agree to cooperate with each other and, if necessary, the City of Industry to promptly execute all other documents and take all other actions reasonably requested by Lessor and/or Company in connection with any and all actions and approvals required to be taken in the ordinary course of formation of such community facilities or other assessment or public financing district over all or any portion of the Premises or any actions related thereto in accordance with the terms of the Mello-Roos Community Facilities Act of 1982, as amended, Rule 15c2-12 promulgated by the Securities and Exchange Commission, or any other California law governing formation of such public financing districts. The parties hereby agree and acknowledge that, in order to assist with the annual maintenance of the landscaping and irrigation systems installed upon the slopes manufactured during the grading of the Project, the parties may initiate proceedings to form one or more maintenance districts which may encompass all or certain portions of the Premises. The parties hereby agree to cooperate in good faith with each other to (i) initiate such formation proceedings, and (ii) form one or more maintenance districts over all or any portion of the Premises. Company and Lessor hereby agree that the Premises or any portion thereof, as determined by the parties, shall be included within the boundaries of said district or districts. In addition, the parties hereto hereby agree to cooperate with each other and, if necessary, the City of Industry to promptly execute all other documents and take all other actions reasonably requested by Lessor and/or Company in connection with any and all actions and approvals required to be taken in the ordinary course of formation of such district over all or any portion of the Premises or any actions related thereto in accordance with the terms of any applicable Federal or California law governing the formation and operation of such districts.

IN WITNESS WHEREOF, Lessor and Company have executed these presents the day and year first above written.

ATTEST:	SUCCESSOR AGENCY TO THE INDUSTRY URBAN-DEVELOPMENT AGENCY
BY:	BY:
	INDUSTRY EAST BUSINESS CENTER, LLC, a Delaware Limited Liability Company
	BY: Majestic Realty Co., a California corporation, its Manager
APPROVED AS TO FORM: Casso & Sparks, LLP	By:
Ву:	By:
	Its:

EXHIBIT "A" LEGAL DESCRIPTION

PARCEL 3 OF PARCEL MAP NO. 352, IN THE CITY OF INDUSTRY, COUNTY OF LOS ANGELES, STATE OF CALIFORNIA, FILED IN BOOK 401 PAGES 29 THROUGH 48, INCLUSIVE OF PARCEL MAPS, IN THE OFFICE OF THE COUNTY RECORDER OF SAID COUNTY.

EXHIBIT "B"

SCOPE OF SUCCESSOR AGENCY IMPROVEMENTS

SCOPE OF AGENCY IMPROVEMENTS

The Agency will, at its cost and expense (unless a public financing district is formed to assist with the financing of a portion of such costs in accordance with Section 4.14 of the Lease Agreement), design and construct the public improvements, located within the public rights of way, reasonably necessary to adequately serve the Project and to provide the mitigation required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center. Said public improvements shall include:

A. Roads

Design and construct streets and all utilities and infrastructure for the following project roadways. (See attached Site Development Plan.)

Street improvements will include curb, gutter, and road pavement design and grades and sidewalks. Street lighting, traffic signals, pavement striping, traffic control signage, landscape and irrigation for roadway medians are also included.

Street A – from Grand Ave west to Grand Crossing Parkway and from Grand Avenue east to Street "B", including special intersection design at Grand Avenue, all street improvements, utilities and infrastructure.

Street B – From Street "A" to Grand Avenue and intersection modifications at Grand Avenue all street improvements, utilities and infrastructure.

Street C – From Street "B" to Street "D", all street improvements, utilities and infrastructure.

Street D – From Street "C" to the end of the cul de sac, all street improvements, utilities, and infrastructure.

Grand Crossing Parkway – From Grand Avenue west to the project boundary, including special intersection design at Grand Avenue, all street improvements, utilities and infrastructure.

Grand Avenue – Modify to accommodate project traffic and project related turn movements. All street improvements, utilities and infrastructure.

B. Master Hydrology

Prepare a master hydrology report and construct the required improvements for the project site based on full buildout of Project as depicted in the EIR The hydrology report will require approval from the Los Angeles County Food Control District for storm drain connections to San Jose Creek.

C. Storm Drain System

Design and construct a storm drainage system to convey water from the planned public streets described above. The storm drainage system will be designed to carry storm water flows from the planned development areas and adjacent property. Lateral stubs will be extended to the street right of way for each development lot. (See attached Storm Drain System.)

D. <u>Sanitary Sewer:</u>

Design and construct sanitary sewer main lines within the project roadways and easements where necessary, to serve each lot within the project development. Plans will include main line connections to existing trunk sewers. (See attached Sewer System.)

E. Water

Design and construct potable water lines within Grand Crossing Parkway, Street A, Street B, Street C and Street D. Lines will connect to the existing systems in Grand Avenue, and Baker Parkway. Service laterals to each lot will be provided, however meter vaults and back flow preventors will not be set. The location of such meter vaults and back flow preventers shall be established by the Company. (See attached Water System).

Install Fire Hydrants as required.

F. Reclaimed Water

Design and construct reclaimed water lines within Grand Avenue, Grand Crossing Parkway, Street A, Street B, Street C and Street D and any other reclaimed lines required for the Project. Design and construct water reservoir described in EIR.

Grand Crossing Parkway, Street A, Street B, Street C and Street D water lines will connect to the existing systems in Grand Avenue and Baker Parkway. Services laterals to each lot will be provided, however meter vaults and back flow preventers will not be set. The location of such meter vaults and back flow preventers will be established by the Company. (See attached Reclaimed Water System.)

G. Dry Utilities

Design and construct ducts, vaults, junction boxes and related substructure elements for Electrical, Telephone (including fiber optics), and main line Gas systems. Line and ducts will stubbed to the right of way line to provide service to each lot. Such work shall be coordinated with utility companies. (See attached Gas Lines.)

H. Grading

Design and construct mass grading at elevations predetermined by Company and Agency, to remove or reduce existing hill areas and to remove and stabilize landslide areas within the Project. Export material will be transported and placed as compacted fill within the street rights of way for the construction of public improvements, or for slopes and creation of the development planning areas described in the EIR. Any excess material will be exported from the Project. The existing utility lines crossing such hill areas will be relocated and any required wetlands mitigation will be undertaken.

I. Former Valley Land Development Company Landfill

Provide the mitigation required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center relating to the Valley Land Development Company landfill.

J. Slope Landscaping

Design and construct the landscaping and irrigation systems required for the Project to satisfy the requirements of the California Environmental Quality Act ("CEQA") and the approved Environmental Impact Report ("EIR") for the Industry Business Center relating to the re-vegetation and landscaping of the slopes manufactured during the grading described in Item H above.

K. Additional Work

Any additional public improvements required to satisfy the CEQA mitigation requirements for the Project.

SUCCESSOR AGENCY
ITEM NO. 5.6



SUCCESSOR AGENCY TO THE

INDUSTRY URBAN - DEVELOPMENT AGENCY

MEMORANDUM

TO:

Honorable Chair and Members of the Successor Agency to the

Industry Urban-Development Agency Board

FROM:

Troy Helling, Executive Director 774

STAFF:

Joshua Nelson, Contract Agency Engineer, CNC Engineering

DATE:

June 13, 2019

SUBJECT:

Consideration of a Betterment Agreement for the Enhancement of Landscaping between the City of Industry, Successor Agency to the Industry Urban-Development Agency and the City of Diamond Bar for the SR 57/60 Confluence Grand Avenue at Golden Springs

Drive Project (MP 99-31 #22)

Background:

In June 1999, the former Industry Urban-Development Agency ("IUDA") entered into a lease to develop a 400-acre site into industrial and commercial buildings known as the Grand Crossing project. This lease required the IUDA to pay for all infrastructure including all project related traffic mitigation measures required for the development of the property. The Successor Agency is now charged with the responsibility of carrying out the existing obligations. An Environmental Impact Report ("EIR") was prepared and certified for the Grand Crossing project in April 2000. This EIR identified all project related traffic mitigation measures including necessary improvements to the intersection of Grand Avenue at Golden Springs Drive. The Project is located within the City of Diamond Bar and will partially satisfy the traffic mitigation measures identified in the EIR.

On April 23, 2015, the City and Successor Agency to the Industry Urban-Development Agency entered a Memorandum of Understanding (MOU) which outlines the arrangement for processing of payments relative to the funding agreement with Los Angeles County Metropolitan Transportation Authority ("Metro") and the use of Successor Agency funds for project costs.

The City secured \$6.7 million in local Proposition C grant funds from Metro's 2011 Call for Projects. This provides a 40% share in project costs related to the right of way acquisition, construction, and construction support services. The remaining 60% share is to be funded with bond proceeds from the Successor Agency. The Project payments made by the City will be reimbursed by both the Metro grant funds and Diamond Bar for the requested enhancements. The Successor Agency is responsible for bidding the project while the City of Industry is the grant recipient and therefore created the need for this to be a three way party agreement.

The Betterment Agreement has been reviewed by Staff and legal counsels of both cities and is found to be in order.

Discussion:

In connection with the SR 57/60 Confluence Project at Grand Avenue at the Golden Springs Drive intersection, the City of Diamond Bar requested enhanced landscaping and decorative traffic signal poles to be added to our project, referred to as a "betterment".

The proposed Golden Springs Drive and Grand Avenue intersection alteration is an approximately \$15.9 million project to add one northbound lane, and one southbound approach lane along Grand Avenue with sidewalk widening to complement the existing freeway mainline (the "Project").

The Project includes improvements to the Grand Avenue and Golden Springs Intersection widening. A new golf cart tunnel will be constructed under Grand Avenue as part of the widening. The project will also widen Golden Springs Drive between Copley Drive and Racquet Club Drive to add a dedicated westbound right-turn lane, widen sidewalks and add pedestrian countdown signals. This project prepares the way for the future 57/60 Chokepoint Relief Project.

The enhanced landscape treatment that is part of the betterment is located along both sides of Grand Avenue and Golden Springs Drive and extends approximately 700' from the intersection. This work includes plant materials, irrigation, specialty fencing, signage, hardscape treatment and the use of stone pavers in lieu of painted crosswalks at all four quadrants of the intersection. Diamond Bar has engaged a landscape consultant to prepare landscape construction drawings and specifications. These documents were included in the drawings for the bid documents, and identified in a separate schedule in the bid proposal. The estimated cost for this schedule of work is \$1,038,000.

Diamond Bar also requests that decorative traffic signal poles be installed at all four corners of the intersection of Grand Avenue and Golden Springs Drive. The cost for the decorative pole upgrade is estimated at \$193,000. Such upgrades were incorporated into the Project's traffic signal plans.

Diamond Bar requested that the City of Industry split the betterment work 50-50. Previously, Council gave direction to Staff to offer to contribute to 40% of the betterment construction cost to match what Los Angeles County Metropolitan Transportation Authority was reimbursing the City for the non-betterment work. At this time Diamond Bar is still insisting on 50-50, but per Council direction the attached agreement reflects 40%.

Enhanced Landscaping	\$1,037,965
Decorative Traffic Signal Poles	\$193,000
Construction Management Costs	\$50,000
Total Cost	\$1,280,965
Diamond Bar's Share (60%)	\$768,579
Industry's Share (40%)	\$512,386

In order to facilitate Diamond Bar's requests, Staff recommends that the City and the Successor Agency to the Industry Urban-Development Agency, enter into a Betterment Agreement with Diamond Bar. The Agreement outlines the proposed enhancements, the cost responsibility, and arrangement for processing of payments associated with the betterments Diamond Bar requested.

Recommendation:

It is hereby recommended that the Successor Agency to the Industry Urban-Development Agency approve and execute the Betterment Agreement. Upon approval this agreement will be forwarded to the City of Diamond Bar for signature.

Exhibit:

A. Betterment Agreement for the Enhancement of Landscaping in Conjunction with the SR 57/60 Confluence Project at the Grand Avenue at Golden Springs Drive Intersection

TH/JN:jv

EXHIBIT A

Betterment Agreement for the Enhancement of Landscaping in Conjunction with the SR 57/60 Confluence Project at the Grand Avenue at Golden Springs Drive Intersection

[Attached]

BETTERMENT AGREEMENT

FOR THE ENHANCEMENT OF LANDSCAPING IN CONJUNCTION WITH THE SR 57/60 CONFLUENCE PROJECT AT THE GRAND AVENUE AT GOLDEN SPRINGS DRIVE INTERSECTION

This Betterment Agreement ("Agreement") is made and entered into this day of
, 2019 ("Effective Date") by and between the City of Industry, a California
municipal corporation, ("Industry"), the Successor Agency to the Industry Urban-
Development Agency, a public body ("Successor Agency"), and the City of Diamond Bar, a
California municipal corporation ("Diamond Bar"). Industry, the Successor Agency and
Diamond Bar are hereinafter collectively referred to as "Parties", and individually as "Party".

RECITALS

WHEREAS, a portion of the SR 57/60 Confluence Project ("Project") is located in the City of Diamond Bar, at the intersection of Grand Avenue and Golden Springs Drive. The Project provides for certain improvements, including an additional northbound through lane, an additional southbound approach lane, a westbound right-turn lane, increased traffic lane widths along Grand Avenue, sidewalk widening, new decorative traffic signals, new street lights on the north side of Golden Springs Drive and the replacement of the golf cart tunnel underneath Grand Avenue; and

WHEREAS, the total estimated Project construction cost is \$15,888,000, inclusive of the costs of construction, utility relocations, and the estimated cost of the Betterment work requested by Diamond Bar under this Agreement;

WHEREAS, Industry has been awarded up to \$6,727,566 in local Proposition C grant funds from the Los Angeles County Metropolitan Transportation Authority's ("Metro") 2011 Call for Projects Program, which provides 40% of eligible project costs for right-of-way acquisition, construction, and construction support services; and

WHEREAS, Industry entered into a funding agreement with Metro, dated November 18, 2014, which identifies the remaining 60% share of the total Project costs to be funded with bond proceeds from the Successor Agency; and

WHEREAS, Industry entered into a Memorandum of Understanding ("MOU") on April 23, 2015, with the Successor Agency to provide payment for all Project related costs, with full reimbursement provided by the Successor Agency and Metro as described hereinabove; and

WHEREAS, under the terms of the MOU, the Successor Agency is required to enter into a construction contract for the Project; and

WHEREAS, the Successor Agency will be advertising the Project for bid and overseeing construction of the Project; and

WHEREAS, the Project will remove and replace certain landscaped medians and traffic signal poles located in Diamond Bar; and

WHEREAS, Diamond Bar has requested that the Project provide enhanced landscaping to replace the landscaped medians that are being removed. Such improvements are more specifically described in Exhibit "A", attached hereto and incorporated herein by reference (hereafter, the "Betterment"); and

WHEREAS, Diamond Bar proposes to reimburse Industry for Sixty Percent (60%) of the actual costs Industry incurs for the construction and construction management of the Betterment. Attached as Exhibit "B" and incorporated herein by reference, is the Parties' estimated cost of the Betterment (hereafter, "Cost Schedule"); and

WHEREAS, Industry agrees to pay for one hundred percent (100%) of Diamond Bar's Project Management services for all project construction exclusive of the Betterment work; and

WHEREAS, Diamond Bar agrees to assume one hundred percent (100%) of the cost for Diamond Bar's Project Management services associated with the Betterment.

NOW, THEREFORE, for and in consideration of the mutual covenants and conditions herein contained, the Parties hereby agree as follows:

SECTION I

Successor Agency Agrees:

- 1. To cause the construction of the Betterment.
- 2. To provide all required construction management and inspection services for the Betterment.
- 3. To track all associated construction costs for the Betterment and to use separate bid schedules for the Betterment so that the actual cost of the Betterment can be determined. The only exception to that will be the traffic signal which will be bid as a lump sum item and includes some Betterment work. Based on the engineer's estimate thirty percent (30%) of the traffic signal work is due to the Betterment and thus thirty percent (30%) will be used to calculate Diamond Bar's share of that item. Construction management and inspection costs will be calculated based on the ratio of the actual construction cost of the Betterment to the total construction cost of the Project.
- 4. Within 60 days of the recordation of the notice of completion for the Project (which includes the Betterment), to furnish Diamond Bar with a complete set of full-size asbuilt plans and any equipment manuals for the Betterment in both hard and electronic format.
- 5. To require its contractor to include Diamond Bar, its officers, agents and employees, as additional insureds to the same extent it requires its contractor to name Industry, its officers, agents and employees as additional insured. Successor Agency shall provide Diamond Bar with a copy of the endorsement it receives evidencing same prior to issuance of the notice to proceed for the Project.

SECTION II

INDUSTRY AGREES:

- 1. To invoice Diamond Bar for sixty percent (60%) of the bid cost of the Betterment from the selected contractor. Diamond Bar shall pay such cost within 30 days of receipt of such invoice and if not paid within that time, the Successor Agency may elect not to construct the Betterment.
- 2. To notify Diamond Bar if there is any increase in the cost of the Betterment above the bid cost, prior to approval of same by the Successor Agency. Diamond Bar may elect to terminate this Agreement as provided in Section IV below if it desires not to incur such increased costs.
- 3. To furnish an accounting of final actual cost of the Betterment and provide Diamond Bar an invoice of the same within one hundred twenty (120) days after acceptance of the Betterment by the Successor Agency.
- 4. To pay one hundred percent (100%) of Diamond Bar's Project Management services for the project exclusive of the Betterment. These will be calculated based on the ratio of the project construction costs exclusive of the Betterment to the total Project construction costs including the Betterment.

SECTION III

DIAMOND BAR AGREES:

- 1. To provide the Successor Agency with all construction plans and specifications for the Betterment as necessary to include the Betterment in the Successor Agency's request for bids and the construction contract at no cost to Industry or the Successor Agency.
- 2. To inform the Successor Agency in writing within fifteen (15) days after receipt of each set of plans, studies, specifications, and/or cost estimates from Industry for the Betterment, if any of the materials are incomplete or if additional information is necessary to facilitate Diamond Bar's review of the Betterment materials.
- 3. After acceptance of the Betterment by the Successor Agency as set forth in Section I.4 of this Agreement, to accept all rights-of-way and permanent easements acquired specifically and solely for the Project by Industry.
- 4. To enforce, at its cost, available rights under existing franchise agreements if existing public and/or private utilities conflict with the construction of the Betterment.
- 5. To issue the Successor Agency or its contractor, a no-fee permit for construction of the Project within Diamond Bar's rights-of-way and perform any required inspections on behalf of Diamond Bar at no cost to Industry or the Successor Agency other than the aforementioned Project Management Services.
- 6. To pay sixty percent (60%) of the bid cost of the Betterment from the selected contractor within fifteen (30) days of receipt of an invoice from Industry for same.
- 7. If the actual cost exceeds the bid cost and Diamond Bar has not terminated this

- Agreement, to pay sixty percent (60%) of the actual cost within 30 days of receipt of Industry's accounting thereof as set forth in Section II.3 above.
- 8. To assume all Project Management costs related to the Betterment incurred by Diamond Bar. These will be calculated based on the ratio of the Betterment construction costs to the total Project construction costs including the Betterment.
- 9. Diamond Bar shall indemnify, defend and hold harmless Industry, the Successor Agency and any and all of their employees, officials and agents from and against any liability (including liability for claims, suits, actions, arbitration proceedings, administrative proceedings, regulatory proceedings, losses, expenses or costs of any kind, whether actual, alleged or threatened, including legal counsel fees and costs, court costs, interest, defense costs, and expert witness fees), to the extent the same arise out of, are a consequence of, or are in any way attributable to, in whole or in part, the performance of this Agreement by Diamond Bar or by any individual or agency for which Diamond Bar is legally liable, including but not limited to officers, agents, employees or subcontractors of Diamond Bar.

SECTION IV

IT IS MUTUALLY AGREED AS FOLLOWS:

- 1. Should any portion of the Betterment be financed with funds with specific expenditure requirements or limitations, all applicable laws, regulations and policies relating to the use of such funds shall apply notwithstanding other provisions of this Agreement.
- 2. If the location of existing facilities of public and/or private utilities conflicts with the construction of Project, the Successor Agency will identify such facilities located within Diamond Bar's right-of-way and request that Diamond Bar cooperate with the Successor Agency in enforcing any available prior rights held by Diamond Bar for such facilities' protection, relocation, or removal; provided that to the extent a utility is not required to bear the cost of such protection, relocation, or removal of its facilities, that will be a Project cost to be borne by Industry and/or the Successor Agency; provided that if such costs arise solely due to the Betterment, Diamond Bar shall be responsible for such costs. Diamond Bar may authorize the Successor Agency to coordinate and inspect such protection, relocation, or removal work, at Diamond Bar's discretion. Nothing in this Agreement shall restrict or affect Diamond Bar's or the Successor Agency's ability to enter into separate agreements with utilities for any purpose, including for reimbursements of utility costs for protection, relocation, maintenance, or removal of their facilities.
- 3. The Successor Agency will furnish a resident engineer to oversee Betterment construction, and Diamond Bar may furnish its own representative. Diamond Bar's representative and Industry's resident engineer will cooperate and consult with each other, but the decisions of Industry's resident engineer shall remain the sole and primary directive for all Betterment work; provided that such directive does not materially change the approved plans and specifications for the Betterment or its estimated cost. The Successor Agency's resident engineer shall obtain approval from Diamond Bar's representative before authorizing any material changes.
- 4. Prior to the Successor Agency's acceptance of the Betterment as completed, Industry will confer with Diamond Bar in good faith to obtain Diamond Bar's written concurrence

that the Betterment has been completed in substantial conformance with the approved plans and specifications. Diamond Bar shall not unreasonably withhold such written concurrence which once provided, shall constitute Diamond Bar's acceptance of the Betterment.

- 5. Prior to the issuance of the Notice to Proceed for the Project, any Party may terminate this Agreement by sending a written termination request to the other Parties. In such an event, this Agreement shall terminate and the Betterment shall not be constructed as part of the Project and any funds provided to Industry by Diamond Bar shall be returned within 30 days thereof, deducting any costs to which Industry may be entitled.
- 6. After the date that construction is to commence as set forth in the Notice to Proceed for the Project, Diamond Bar may terminate this Agreement by sending a written termination to Industry and the Successor Agency and proposing a date of termination that is no earlier than ninety (90) days after the date of the written termination request. Upon receipt or issuance of a notice of termination, the Successor Agency shall cease or cause its contractor to cease, all work on the Betterment except such work which is reasonably necessary to close out the work. Industry shall then tabulate all outstanding costs incurred, and any pending costs associated with the Betterment through the date of cessation, and submit a refund notification to Diamond Bar for any unspent Betterment funds. Industry shall remit payment to Diamond Bar within 30 days of the refund notification.
- 7. No Party shall be liable to any other for any damages, delay costs, or termination costs of any type or a failure to perform any part of this Agreement due to causes beyond the control of the Parties. Such causes include, but are not limited to, acts of God, acts of the public enemy, acts or inactions of federal, state or local governments including funding reductions or eliminations, fires, floods, and severe weather.
- 8. If any Party breaches this Agreement, the non-breaching Party(ies) shall, prior to seeking any remedy for such breach, notify the breaching Party in writing, setting forth the details of the alleged breach and the requested cure. The breaching Party shall promptly commence curing such breach and pursue to completion within thirty (30) days of such written notification or within such other reasonable time that the Parties may agree to, if the breach is not curable within a 30-day period,. If the breaching Party fails to promptly pursue such cure to completion, the breaching Party shall be in default under the terms of this Agreement. In the event of an uncured default, the non-breaching Party(ies) may pursue any legal or equitable remedies available to it including specific performance and the non-breaching Party(ies) shall have no obligation to make any payments to or undertake any work for the breaching Party until and unless the default is cured.
- 9. Every notice, demand, request, or other document or instrument delivered pursuant to this Agreement shall be in writing, and shall be either personally delivered, by Federal Express or other reputable overnight courier, or sent by certified United States Postal Service (USPS) mail, postage prepaid return receipt requested, to the addresses set forth below, or to such other address as a Party may designate from time to time.

To INDUSTRY: City of Industry

15625 E. Stafford Street, #100 City of Industry, CA 91744

Attention: Troy Helling, City Manager

Phone: (626)333-2211

To SUCCESSOR AGENCY: Successor Agency to

the Industry Urban-Development Agency

15625 E. Stafford Street, #100 City of Industry, CA 91744

Attention: Troy Helling, Executive Director

Phone: (626)333-2211

To DIAMOND BAR: City of Diamond Bar

21810 Copley Drive Diamond Bar, CA 91765

Attention: David Liu, Public Works

Director/City Engineer Phone: (909) 839-7040

- 10. This Agreement contains the entire understanding between the Parties relating to the obligations of the Parties described in this Agreement. All prior or contemporaneous agreements, understandings, representations, and statements, oral or written and pertaining to the subject of this Agreement or with respect to the terms and conditions of this Agreement, are merged into this Agreement and shall be of no further force or effect. Each Party is entering into this Agreement based solely upon the representations set forth herein and upon each Party's own independent investigation of any and all facts such Party deems material. This Agreement may be amended in writing at any time by the mutual consent of the Parties. No amendment shall have any force or effect unless executed in writing by the Parties.
- 11. If any term or provision of this Agreement or the application thereof to any person or circumstance shall, to any extent, be invalid or unenforceable, then such term or provision shall be amended to, and solely to, the extent necessary to cure such invalidity or unenforceability, and in its amended form shall be enforceable. In such event, the remainder of this Agreement, or the application of such term or provision to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby, and each term and provision of this Agreement shall be valid and be enforced to the fullest extent permitted by law. This Agreement shall be construed and enforced in accordance with the laws of the State of California. Venue for any cause of action shall be Los Angeles County, California.
- 12. No assignment of this Agreement shall relieve the assigning Party of its obligations under this Agreement until such obligations have been assumed in writing by the assignee. When duly assigned in accordance with the forgoing, this Agreement shall be binding upon and inure to the benefit of the assignee.
- 13. The waiver by Industry, the Successor Agency, or Diamond Bar of any breach of any term, covenant or condition herein contained shall not be deemed to be a waiver of

such term, covenant or condition or of any subsequent breach of the same or any other term, covenant or condition herein contained. No term, covenant or condition of this Agreement shall be deemed to have been waived by Industry, the Successor Agency, or Diamond Bar unless in writing.

14. Neither Industry, the Successor Agency, nor Diamond Bar intend that there be a third-party beneficiary to this Agreement.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be executed by their respective officers as of the Effective Date.

CITY OF INDUSTRY	CITY OF DIAMOND BAR
, Mayor	Carol Herrera, Mayor
ATTEST:	ATTEST:
Julie Gutierrez-Robles, Deputy City Clerk	Tommye A. Cribbins, City Clerk
APPROVED AS TO FORM:	APPROVED AS TO FORM:
James M. Casso, City Attorney	David A. DeBerry, City Attorney
SUCCESSOR AGENCY TO THE INDUSTRY URBAN- DEVELOPMENT AGENCY	
, Chairperson	
ATTEST:	
Julie Gutierrez-Robles, Deputy Agency Secretary	
APPROVED AS TO FORM:	
James M. Casso, Agency General Counsel	

Exhibit A

The Betterment shall consist of the following: enhanced landscape treatment located along both sides of Grand Avenue and Golden Springs Drive as indicated on landscape drawings provided by Diamond Bar, extending approximately 700' from the intersection. Diamond Bar shall provide the Successor Agency with the final signed plans for the Betterment within 30 days of the Effective Date of this Agreement. The Betterment shall be constructed pursuant to the final signed plans.

The enhanced landscape treatment includes plant materials, irrigation, accent lighting, decorative fencing, hardscape treatment and the use of stone pavers in lieu of painted crosswalks at all four quadrants of the intersection.

All work to be performed in accordance with the enhanced landscaping design approved and provided by Diamond Bar.

Exhibit B COST SCHEDULE

Estimated Cost of Requested Betterment

1) Enhanced Landscaping (Schedule H in bid specs) \$1,037,965 2) Traffic Signal Upgrades (30% of Schedule C in bid specs) \$ 193,000

Betterment Subtotal = \$1,230,965

Construction Cost and Utility Relocations = \$14,657,035

Estimated Total Project Construction Cost inclusive of Betterment \$15,888,000

Final payment to be based upon the bid amounts for Schedule H and 30% of Schedule C as contained in the awarded construction contract.

Note:

Industry payment for Diamond Bar's project management services is estimated to be 92.3% of the Management Service fees incurred by Diamond Bar based upon the following formula:

92.3% = \$14,657,035 Estimated Total Project Construction Cost exclusive of Betterment Estimated Total Project Construction Cost inclusive of Betterment

Diamond Bar payment of Diamond Bar's project management services is estimated to be 7.7% of actual Management Service fees incurred by Diamond Bar

The above percentages to be adjusted according to the Total Project Construction Cost and Betterment Cost as determined per the awarded construction contract.